

Contagious Employee Wellbeing. Examining Organisational and Team Influences on Employee Wellbeing Through a Multilevel Lens

by Aglae Hernandez Grande

Thesis submitted in fulfilment of the requirements for the degree of

Doctor of Philosophy

under the supervision of:	Professor Simon Darcy
	Dr Mihajla Gavin
	Associate Professor Ben Farr-Wharton

University of Technology Sydney Faculty of Business

June 2023

Declaration of Originality

CERTIFICATE OF ORIGINAL AUTHORSHIP

I, Aglae Hernandez Grande, declare that this thesis is submitted in fulfilment of the requirements for the award of Doctor of Philosophy, in the Management Discipline Group, UTS Business School at the University of Technology Sydney.

This thesis is wholly my own work unless otherwise referenced or acknowledged. In addition, I certify that all information sources and literature used are indicated in the thesis.

This document has not been submitted for qualifications at any other academic institution.

This research is supported by the Australian Government Research Training Program.

Signature:

Production Note: Signature removed prior to publication.

Aglae Hernandez Grande

Date: 2nd of June 2023

Acknowledgements

A PhD is a long and sometimes lonely journey. In my case, it has felt long but never lonely. This thesis would not have been possible without each of the people who have supported me and ensured that I never felt alone. To all of you, thank you.

To my best friend, my partner, my dive buddy, my biggest ally and supporter, Maria: thank you for your support, encouragement and motivation. You have selflessly been there to support and encourage me every single second of my journey. This thesis would not have been possible without your unconditional trust. *Gracias desde el fondo de mi corazón*.

To my wonderful parents, Aglae and Quim: thank you for all your sweets, cooked meals, words of motivation, long phone conversations and consistent willingness to read whatever I write. Thank you for always believing in me, for always pushing me a step further, for always offering support and comfort and for always reminding me that the goal was just around the corner. *Gràcies per creure sempre en mi*.

To my supervisors, Simon, Mihajla and Ben: from the bottom of my heart, I thank you. You have supported me and my PhD journey through all the ups and downs. You have been there when I needed support, but you have also given me space. In particular, thank you, Ben. I am extremely lucky to have such an inspiring mentor, supervisor and manager. I am forever grateful for everything you taught me and all the knowledge you shared.

To my ECU friends – my academic family – Fleur, Esme and Hossein: I could not have been luckier to have such a supporting and loving team. Thank you for being the best academic friends someone can have. I owe you so many coffees, hot chocolates, cookies, 1pm chips and toasties.

To all my friends and family, I thank you.

To all the statisticians, psychometricians and professors of quantitative methods who have offered their time to support my multilevel analysis learning curve: thank you, Professor Michael Zyphur, for your courses and for your consultation hours; thank you, Professors Bengt and Linda Muthén, for developing Mplus and for always being available to answer my questions; thank you, Dr Chris Stride and Mr Philip Holmes-Smith for your multilevel courses. Also, thank you, Professor Tim Bentley for being my stage three assessor and offering such constructive and insightful feedback. Last but definitely not least, thank you, Associate Professor Margaret Franken, for spending time proofreading my work and from whom I have learnt a lot.

Thank you.

Aglae

List of published works by the author

Farr-Wharton, B., Brunetto, Y., Hernandez Grande, A., Brown, K., & Teo, S. (2022). Emergency service workers: The role of policy and management in (re)shaping wellbeing for emergency service workers. *Review of Public Personnel Administration*. Advance online publication. https://doi.org/10.1177/0734371X22113097

List of refereed conference proceedings

- Hernandez Grande, A., Sharafizad, F., & Farr-Wharton, B. (2022, December 6-7). Team antecedents of employee wellbeing for first responders in Australia. 2022 ANZAM Conference, Gold Coast, QLD, AUS. https://anzamconference.org/
- Hernandez Grande, A., Farr-Wharton, B., Simpson, A.V., Prichard, R., & Reddy, P. (2019, April 16-18). Recommendations to measure wellbeing in the workplace. A meta-analysis of the wellbeing measures in the public and private sector [Paper presentation]. XXIII Annual Conference of the International Research Society for Public Management, Wellington, NZ. https://www.irspm.org/conferences/conference-2019
- Hernandez Grande, A., Farr-Wharton, B., Simpson, A.V., Prichard, R., & Reddy, P. (2019, July 18-21). Male privilege and employee wellbeing: A call for critical wellbeing [Poster]. 6th World Congress on Positive Psychology, Melbourne, VIC, AUS. https://www.pesa.edu.au/stec_event/ippa-6thpositive-psychology-world-congress-melbourne-2019/
- Farr-Wharton, B., Simpson, A.V., Prichard, R., Hernandez Grande, A., & Reddy, P. (2019, July 18-21). *The correlates of organizational compassion* [Paper presentation]. 6th World Congress on Positive Psychology, Melbourne, VIC, AUS. https://www.pesa.edu.au/stec_event/ippa-6th-positivepsychology-world-congress-melbourne-2019/
- Farr-Wharton, B., Simpson, A.V., Hernandez Grande, A., Prichard, R., & Reddy, P. (2019, April 16-18). Social justice and equity: A call for critical wellbeing [Paper presentation]. XXIII Annual Conference of the International Research Society for Public Management, Wellington, NZ. https://www.irspm.org/conferences/conference-2019
- Simpson, A.V., Farr-Wharton, B., Prichard, R., Hernandez Grande, A., & Reddy, P. (2019, June 5-6). Organizational compassion dampens workplace bullying amongst healthcare staff [Paper presentation]. 2019 POS Research Conference, Michigan, MI, USA. https://positiveorgs.bus.umich.edu/wp-content/uploads/POS-RC-2019-Program-FINAL.pdf

Table of contents

Acknowledgements	iii
List of tables	vii
List of figures	ix
List of equations	ix
List of published works by the author	iv
List of refereed conference proceedings	iv
Glossary (Notes on terminology)	x
List of abbreviations	X
Abstract	xii
Chapter 1. Introduction	xii
1.1. Context and research gap	1
1.2. Theoretical framework	5
1.2.1. Social exchange theory	6
1.2.2. Emotional contagion theory	7
1.2.3. Conservation of resources theory	7
1.3. Organisational resources	8
1.3.1. Managerial priority: Psychosocial safety climate	8
1.3.2. Managerial support: Leader-member exchange	8
1.4. Research rationale	8
1.5. Research aims and research questions	9
1.6. Significance and contribution to practice	10
1.7. Structure of the thesis	11
1.8. Conclusion of the chapter	12
Chapter 2. Theoretical and Conceptual Frameworks	14
2.1. Understanding workplace relationships and behaviours	14
2.2. The collective influence on employee wellbeing	20
2.3. Identifying organisational resources that influence employee wellbeing and job stres	s . 24
2.4. Theoretical blending as theoretical framework	31
2.5. Employee wellbeing	33
2.5.1. From happiness to wellbeing	33
2.5.2. From wellbeing to flourishing	37
2.5.3. Employee wellbeing and work stress	39
2.5.4. Wellbeing measures	42
2.6. Antecedents of employee wellbeing	44
2.6.1. Managerial priority: Psychosocial safety climate	45
2.6.2. Managerial support: Leader-member exchange	47
	v

2.7. Multilevel organisational theory. The need to study work teams as a cluster	50
2.8. Professional services industry in Australia	54
2.9. Conclusion of the chapter	55
Chapter 3. Methodology	56
3.1. Approaches to employee wellbeing research	56
3.2. Research paradigm	59
3.2.1. Positivism and post-positivism	63
3.3. Quantitative methods	63
3.4. Research design	64
3.4.1. Population sampling	66
3.4.2. Data collection	69
3.4.3. Survey considerations	
3.4.4. Questionnaire scales	74
3.4.5. Data analysis	81
3.5. Ethics and risks	
3.6. Conclusion of the chapter	
Chapter 4. Results	100
4.1. Descriptive statistics	100
4.2. Sample demographics	101
4.3. Results of measurement model	102
4.3.1. Multivariate normality	102
4.3.2. Intraclass correlation	104
4.3.3. Multilevel exploratory factor analysis	105
4.3.4. Multilevel confirmatory factor analysis	105
4.4. Results of the multilevel structural model	115
4.4.1. Bayes estimator	115
4.4.2. Model fit	116
4.4.3. Results of hypothesis testing	117
4.5. Conclusion of the chapter	124
Chapter 5. General Discussion and Implications	125
5.1. Overview of the research questions and hypotheses	125
5.1.1. Do employees influence each other's wellbeing when working together?	126
5.1.2. To what extent do organisational resources impact the level of employee we	llbeing?
5.2. Contributions and implications for theory and research	
5.3. Contributions and implications for practice and policy	135
5.4. Conclusion of the chapter	137
*	-

Chapter 6. Conclusions	138
6.1. Important findings	138
6.2. Limitations and recommendations for future research	140
6.3. Conclusion of the chapter	141
References	143
Appendices	180
Appendix 1. Survey	180
Appendix 2. Recruitment flyer and video link	183
Appendix 3. Generic participant information sheet	184
Appendix 4. Multilevel exploratory factor analysis for psychosocial safety climate and emotional contagion	. 186
Appendix 5. Multilevel McDonald's omega code for the scale of employee wellbeing	186
Appendix 6. Multilevel structural equation code to test the hypothesised model	186

List of tables

Table 1. Research questions
Table 2. Summary of the theoretical framework of this study
Table 3. Workplace relationships under analysis within the social exchange theoretical framework
Table 4. Propositions of emotional contagion theory 21
Table 5. Types of emotional contagion
Table 6. Summary of empirical literature examining employee wellbeing from the emotional
contagion theory perspective
Table 7. Organisational and team resources by category and kind
Table 8. Principles of conservation of resources theory 27
Table 9. Corollaries of the conservation of resources theory
Table 10. Summary of empirical literature examining employee wellbeing from the perspective
of the conservation of resources theory
Table 11. Measures of wellbeing
Table 12. Principles of multilevel organisational theory 52
Table 13. Research studies that have analysed the antecedents of employee wellbeing using
single-level structural equation modelling analysis between January and October 2022
Table 14. Research studies that have analysed the antecedents of employee wellbeing using
multilevel structural equation modelling analysis between January and October 2022
Table 15. Summary of the five major research paradigms according to their logic, ontology,
epistemology, axiology, and methodology

Table 16. Number of individuals at the within level and the number of groups at the between level
of recent organisational multilevel studies
Table 17. Latent constructs included in the survey
Table 18. Journal articles in English that referenced and used the job stress scale developed by
Cullet et al. (1985) between January and October 202276
Table 19. Journal article in English that referenced and used the emotional contagion scale
developed by Siebert et al. (2007) between January and October 2022
Table 20. Journal articles in English that used the psychosocial safety climate scale developed by
Dollard and Bakker (2010) between January and October 2022
Table 21. Journal articles in English that used the Leader-member exchange scale developed by
Graen and Uhl-Bien (1995) with Australian data between January and October 202280
Table 22. Ethical considerations
Table 23. Normalised minimum, maximum, mean and standard deviation of the model variables
on a scale of 0 to 1101
Table 24. Frequency analysis of the demographic questions102
Table 25. Skewness and kurtosis of all survey items
Table 26. Intraclass correlation at the company between level for the model variables104
Table 27. Intraclass correlations of the employee wellbeing scale items
Table 28. Standardised within and between factor loadings of the employee wellbeing scale .106
Table 29. Parameters to measure the McDonald's omega for the employee wellbeing scale 107
Table 30. Intraclass correlations of the work stress scale items107
Table 31. Standardised within and between factor loadings of the work stress scale108
Table 32. Parameters to measure the McDonald's omega for the work stress scale109
Table 33. Intraclass correlations of the psychosocial safety climate scale items109
Table 34. Standardised within and between factor loadings of the psychosocial safety climate
scale
Table 35. Parameters to measure the McDonald's omega for the psychosocial safety climate scale
Table 36. Intraclass correlations of the employee wellbeing scale items
Table 37. Standardised within and between factor loadings of the Leader-member exchange scale
Table 38. Parameters to measure the McDonald's omega for the Leader-member exchange scale
Table 39. Intraclass correlations of the emotional contagion scale items 113
Table 40.Standardised within and between factor loadings of the purified emotional contagion
scale

Table 41. Parameters to measure the McDonald's omega for the emotional contagion scale114
Table 42. Between-level, and within-level effects between employee wellbeing, work stress,
Leader-member exchange, psychosocial safety climate, and emotional contagion117
Table 43. Between-level, and within-level standardised effects between employee wellbeing,
work stress, Leader-member exchange, psychosocial safety climate, and emotional contagion
Table 44. Research questions 126

List of figures

Figure 1. Theoretical and conceptual frameworks of this thesis	6
Figure 2. Theoretical framework of this thesis	15
Figure 3. Csikszentmihalyi's model of flow	35
Figure 4. Pie chart of the sustainable happiness model	36
Figure 5. Hypothesised model	53
Figure 6. What informs a research design?	60
Figure 7. Boxplot of the responses for employee wellbeing by work group	85
Figure 8. Within and between variance in multilevel structural equation modelling	using latent
variables	87
Figure 9. Path diagram of a latent construct with three items	92
Figure 10. Hypothesised model with standardised parameter estimates	119
Figure 11. Moderation effect	120

List of equations

/
7
5
7
9
1
2

Glossary (Notes on terminology)

- **Co-worker**: A person with whom one works within the same team in an organisation.
- Emotional contagion: A form of social contagion that involves "catching" others' emotions and moods.
- **Employee wellbeing**: Workers' mental, physical, financial and psychological wellness at work.
- Employee: A person working in an organisation in exchange for a salary.
- Leader-member exchange: Organisational resource defined as the employee-manager relationship from the perspective of the employee.
- **Managerial priority**: The employees' perception of how their organisation prioritises their wellbeing above productivity and outputs.
- **Managerial support**: The employees' perception of how managers support and care about their wellbeing and are attentive to employee needs.
- **Professional services industry**: Industry comprised of occupations in the service sector and that require special training and/or a qualification.
- **Psychosocial safety climate**: Organisational resource defined as the perceptions of organisational practices, policies, and procedures designed to maintain employees' psychological health and safety.
- Team (or work team): A group of employees with complementary skills who work together to achieve specific goals and under the same direct manager.

List of abbreviations

- BCE: Before Common Era
- CFA: Confirmatory Factor Analysis
- **CFI**: Comparative Fit Index
- CMV: Common Method Variance
- EFA: Exploratory Factor Analysis
- MAR: Missing at Random
- MCAR: Missing Completely at Random
- MCFA: Multilevel Confirmatory Factor Analysis
- MEFA: Multilevel Exploratory Factor Analysis
- MLA: Multilevel Analysis
- MNAR: Missing not at Random
- MSEM: Multilevel Structural Equations Modelling

- **PERMA**: Positive Emotion, Engagement, Positive Relationships, Meaning, Accomplishments/Achievements
- **PSC**: Psychosocial Safety Climate
- **RMSEA**: Root Mean Square Error of Approximation
- SEM: Structural Equations Modelling
- SPSS: Statistical Package for the Social Sciences
- SRMR: Standardized Root Mean Square Residual
- TLI: Tucker-Lewis Index

Abstract

Employee wellbeing is receiving increasing attention as organisations grapple with increasingly complex workplace issues, including employee mental health. Interest in employee wellbeing has further increased since the beginning of the COVID-19 pandemic, as significant restrictions have impacted people's mental health. To date, workplace wellbeing initiatives and interventions have primarily focussed on a single conception of wellbeing and stress. However, there is no research on the impact team's emotional contagion on employee wellbeing. To explore the multiple dynamics that contribute to collective wellbeing, this thesis formulates hypotheses that draw from a combination of three theories: social exchange theory, conservation of resources theory, and emotional contagion theory. First, social exchange theory explains workplace behaviours. Emotional contagion theory identifies dynamics of the contagion effect of wellbeing between coworkers, while the conservation of resources theory assists in understanding how employees value organisational resources to reduce their levels of stress and support their wellbeing. By applying a pragmatic research philosophy under a multilevel statistical approach, this thesis looks at the cross-level moderating role of emotional contagion in the relationship between work stress and employee wellbeing. The impacts of the organisational resources of managerial priority and support are also explored through a chain mediation model. In this way, the thesis aims to build a more comprehensive model of employee wellbeing. Empirical testing utilises two-level, crosssectional data drawn from 237 employees nested in 41 work teams from professional services across Australia. Although this thesis is relevant for multiple industries, the team dynamics of the professional services industry and its workers' abilities to work from home during COVID-19 made it a fruitful area to study for this thesis. We identified evidence supporting the hypotheses of the study, including a significant cross-level moderation effect. Managerial support and work stress serve as mediators in the positive relationship between managerial priority and employee wellbeing at both levels of analysis. The significance of this research lies in its study of the contagion effect of wellbeing on other members of the work teams. This study highlights the importance of organisational resources along with the important role of co-workers in maintaining employee wellbeing. The main implication is that organisational interventions to improve the level of employee wellbeing benefit from factoring in team dynamics to mitigate stress levels.

Chapter 1. Introduction

This thesis aims to understand organisational and team influences on employee wellbeing focusing on the Australian context. This chapter offers an overview of the thesis, beginning with a presentation of the context of the study and the research gap it addresses. Over the past decades, employee wellbeing has received increasing attention from researchers, public administrators, and organisations. In this thesis, employee wellbeing focuses on workers' physical, social, and psychological wellness at work. The interest in supporting employee wellbeing increased during the COVID-19 pandemic as we were facing lockdowns, curfews and restrictions that severely impacted our mental health. Most research has not considered the fact that employees work in teams and instead, have studied employees as isolated units. Little or no research currently appears to exist on the collectivist conceptualisation of employee wellbeing. Without studying the team influence on employee wellbeing, current interventions, policies, and practices may not be supporting the wellbeing of employees to the full extent possible. The chapter goes on to introduce the theoretical framework by outlining the three theories that guide this research study: social exchange theory, emotional contagion theory, and conservation of resources theory. To understand if employees influence each other's wellbeing when working together, this study also studies how managerial resources support employee wellbeing. Both managerial support and managerial priority have been considered the most important organisational resources to support employees' wellbeing. Chapter 1 also includes the research rationale, research aims, and research questions that guide this study. The chapter concludes with the significance of this project, its contribution to practice, and a general outline of the structure of the presented thesis.

1.1. Context and research gap

The labour market and the employment landscape have significantly changed over the past few years. The world is moving towards a more automated workflow, where the design, execution, and validation of many tasks previously done by humans can now be undertaken by machines or robots (Sestino & De Mauro, 2022). Alongside this development, employees are undertaking other tasks outside of these automated processes, where human interactions are critical. Moreover, the employment landscape is being altered by demographic forces and the working population is aging significantly (Roscigno et al., 2022). Notwithstanding impact that "the Great Resignation", the current skills shortage, and the so-called "War for Talent" have on the economy and the society overall (Serenko, 2022). These changes, along with the gig economy, globalisation, and the impact of COVID-19, are increasing the complexity of business operations (Näswall et al., 2021; Suter et al., 2020). This complexity brings with it other important demands that must also be included and embedded in organisational policies, practices, aims, and goals. One of these

demands is the need to pay extra attention to the wellbeing of employees and find ways to reduce job stress and the impact of psychosocial hazards.

In July 2022, Safe Work Australia published a code of practice on how to manage psychosocial hazards at work. Safe Work Australia is Australia's leading agency in developing policy on health and safety. The code of practice, published under the section 274 of the Work Health and Safety Act 2022 (Comm), implies that organisations must identify and eliminate or, if not possible, minimise all hazards that could pose a psychosocial risk (Safe Work Australia, 2022). As such, addressing the impact of workplaces on wellbeing is no longer an option but by law, organisations are now required to provide measures that eliminate or reduce psychosocial risks and support employee wellbeing.

Wellbeing is a dynamic concept that goes beyond the state of being happy or healthy. Wellbeing is challenging to define, and there is little or no agreement on how organisations and policy makers can support employees' wellbeing. According to Seligman (2012), wellbeing has five important pillars that are encapsulated in his PERMA model. PERMA includes five building blocks: (1) *Positive emotions* or feeling good, (2) *Engagement* or being absorbed in your daily activities, (3) positive *Relationships* and the ability to be authentically connected to those who surround you, (4) finding *Meaning* and having a purposeful existence, and (5) a sense of *Accomplishment* and success in life. Seligman's definition of wellbeing merges two different approaches that have been traditionally used to define wellbeing. On the one hand, *hedonism* defines wellbeing as satisfaction, happiness, pleasure, and positive affect. On the other hand, *eudaimonia* encapsulates the fulfilment of finding meaning and personal success.

Employee wellbeing is receiving increasing attention as organisations grapple with ever-more complex workforce issues. There is a general belief that employee wellbeing is a key consideration for managers, organisations, and workers themselves. This importance is because wellbeing has a perceived connection to a number of positive individual and workplace outcomes. Yet, a large gap remains between perceived importance of employee wellbeing and readiness to support it with organisational resources (e.g., policies and practices that support wellbeing). Higher levels of employee wellbeing have been linked with higher levels of employee performance (Brunetto et al., 2014), and with better organisational outputs (Krekel et al., 2019). In contrast, poor employee wellbeing has been associated with higher levels of psychological distress (de Jonge & Schaufeli, 1998), increased turnover, absenteeism, and presenteeism (Baptiste, 2008).

Christakis and Fowler (2009) explain the power of social networks and how wellbeing is contagious in social circles. They argue that when someone is happy and positive, individuals

who are connected to that person tend to become happy and positive. This contagious effect does not stay in this first-degree connection; instead, it can impact up to third-degree connections (Christakis & Fowler, 2009). First-degree connections are considered direct connections, while third-degree connections are contacts of your contact's contact. Negative feelings and low levels of wellbeing can also be contagious (Steinert, 2020) and can occur in the workplace between coworkers and those with whom employees interact on a daily basis. This theoretical development assumes an influence of the team on wellbeing. Yet, this team influence on employee wellbeing has not been empirically tested.

Not only is wellbeing a challenging concept to define but this is also the case with employee wellbeing. According to Ryan and Deci, wellbeing "is a complex construct that concerns optimal experience and functioning" (2001, p. 141). There is little agreement on a definition, which blurs its utility. In this thesis, employee wellbeing focuses on workers' physical, social, and psychological wellness at work. Some scholars suggest that employee wellbeing also includes the hedonic and eudaimonic approaches, as mentioned above, but focuses on these experiences at work (Näswall et al., 2021).

The lack of agreement on a definition of employee wellbeing has decreased the chances of having a proper measure to be able to develop further research (Salmee et al., 2020). Indeed, over the past 50 years, some 30 measures of wellbeing have been developed and tested in an organisational context. Nonetheless, it is important to consider that the validation of both individual and collective antecedents of employee wellbeing may be limited due to the application of consistently applied but undeveloped measures. While there have been valuable contributions from reviews and meta-analyses involving wellbeing indicators (see for example, Kuoppala, et al., 2008), further research is needed to comprehensively validate these antecedents. Most of the developed and employed scales do not include a full definition of employee wellbeing, and instead most scales are a composite of factors of wellbeing (Bayhan Karapinar et al., 2020; Butler & Kern, 2016).

The lack of an agreement on a definition of employee wellbeing and its measurement poses ongoing theoretical and practical problems. These problems impact the development of robust theory and practice concerning the role of organisations in supporting employee wellbeing and the relative cost-benefit of such actions for organisations, notwithstanding ethical considerations. For many companies, incorporating policies and practices to support employees' wellbeing has not been an easy journey. Some researchers, practitioners and employees feel that there is much more to be done and that in many instances, companies do not have the necessary tools to support their employees' wellbeing (Volini et al., 2020). The 2020 Deloitte report on Global Human Capital Trends, "Designing work for well-being: Living and performing at your best" found that 80% of American organisations affirm that employee wellbeing is important or very important for their success (Volini et al., 2020). Employee wellbeing is not only important for organisational success, but for the society as a whole (Jaiswala et al., 2022). However, only 12% of these companies were actively working on including employee wellbeing within their programs. Deloitte calls this phenomenon "the readiness gap" (Volini et al., 2020, p. 35).

Wellbeing had the largest gap between importance and readiness across this year's trends, with 80 percent of organizations saying worker well-being is important or very important for their success over the next 12-18 months, but only 12 percent saying they are very ready to address this issue. (Volini et al., 2020, p. 35)

The COVID-19 pandemic has further increased the interest in employee wellbeing. In November 2019, several occurrences of people infected with the coronavirus were discovered in Wuhan, China. The virus rapidly travelled around the world, and on the 11th March 2020, the World Health Organisation declared the outbreak a pandemic (World Health Organisation, 2020). This caused a sudden change for society, and individuals were forced to stay at home, with lockdowns, curfews, and other restrictions put in place. Amongst these changes, many employees were asked to work from home to reduce close contact with non-family members and prevent the spread of the virus. Before COVID-19, flexible work arrangements were seen as a luxury, but since the pandemic was declared, provision of these arrangements shifted from a discretionary policy to a mandatory requirement, also known as "forced flexibility" (Franken et al., 2021, p. 1131). Since many employees were forced to work remotely, their wellbeing was impacted due to an increasing demand of care obligations at home and having to balance work and personal life while working from home (Franken et al., 2021). Therefore, since the beginning of the pandemic, the interest in employee wellbeing, along with the gap between importance and readiness, has grown.

Undeniably, employee wellbeing has an important impact on mental and physical health (Platts et al., 2022). The Australian Department of Health has planned a budget for 2022 and 2023 to prioritise mental health, preventive health, and sport (Department of Health, 2022). The budget has five different pillars: (1) prevention and early intervention, (2) suicide prevention, (3) treatment, (4) supporting vulnerable people, and (5) workforce and governance. Focusing on pillar five, which is concerned with prioritising mental health and suicide prevention in the workforce, the Department of Health has invested AUD\$89.2 million to increase the workforce specialised in mental health, and along with other mental health programs, they have budgeted AUD\$3 billion for 2022-2023 (Department of Health, 2022). The Australian Productivity Commission estimates the cost of poor mental health to be between AUD\$200 and AUD\$220 billion a year (King, 2021).

As with general wellbeing, social relationships impact employees' wellbeing. More specifically, psychologists argue that workplace relationships and behaviours have an important impact on employees' wellbeing (Finkel et al., 2017). But still today, most organisational programs that try to offer interventions to support employees' wellbeing offer individual managerial resources. Some examples of wellbeing initiatives for workplaces include employee assistance programs, counselling services, gym memberships, or mindfulness sessions (Goldstone et al., 2021). The literature highlights that for the success of these wellbeing initiatives, there are two important contributors. First, it is important to have a workplace safety climate (Bailey et al., 2015), and second, a supportive manager (Brunetto et al., 2011). However, these programs cannot cover the entire workforce. Some employees may still feel distressed or suffer high levels of anxiety and/or depression regardless of the organisational resources available to support them (Bamberger et al., 2015). Any distressed employees are likely to interact with co-workers and could potentially have a negative impact on the wellbeing of colleagues (Xerri et al., 2022). Moreover, according to Christakis and Fowler (2009), these distraught employees could even have a broader negative influence on the contacts of their co-workers. Yet, to the researcher's best knowledge, the extent of the influence of these distraught employees has not been accounted for or studied.

The antecedents and outcomes of employee wellbeing have been widely acknowledged and explored within the scholarly literature. However, there appear to be very few studies analysing different aspects of employee wellbeing incorporating the idea that employees are nested within groups and that as such, the group must have an impact on the individuals (e.g., Bakker, 2015; Benitez et al., 2019; Farr-Wharton et al., 2021; Salmee et al., 2020). While several studies have examined the impact of team-level variables such as climate, conflict, and cohesion on various wellbeing indicators (González-Romá et al., 2002; Gamero et al., 2008), a gap remains in understanding the specific role of team emotional contagion and its effects on employee wellbeing. The work team refers to those employees that work together and that interact with each other frequently, whether it is online or face-to-face (Xerri et al., 2022). To be able to establish how organisations and managers can positively contribute to improve the levels of employee wellbeing, there is an urgent need to understand a collectivist conceptualisation of employee wellbeing. Employee wellbeing is the central construct under investigation in this thesis with the aim of contributing to the development of healthier workplaces.

1.2. Theoretical framework

To study how team members and organisational resources contribute to employee wellbeing, it is necessary to understand workplace social relationships. This thesis is guided by a theoretical framework that comprises multiple theoretical underpinnings (see Figure 1). The process of using multiple theoretical lenses is understood as theoretical blending (Cornelissen & Durand, 2012).

The overarching framework is social exchange theory, which can be drawn on to better understand workplace relationships and behaviours (Porter, 2018). However, social exchange theory has been referred to as a broad conceptual paradigm, which can explain any interaction, at least post hoc (Cropanzano et al., 2016). As such, social exchange theory poses some important limitations in explaining how team members influence each other when working together as part of a group. For this study, it is essential to understand if employees influence each other when working together, and therefore, this study also draws upon two other theories: emotional contagion theory and conservation of resources theory. Overall, the intersection between these theories represents the theoretical framework of this study to understand how the team and organisational resources influence employee wellbeing.





Note. The theoretical and conceptual frameworks of this study using the theoretical blending of social exchange theory, emotional contagion theory, and conservation of resources theory.

1.2.1. Social exchange theory

Social exchange theory explains human relationships and interactions. According to this theory, there is a reciprocal dynamic between two or more parties where one gives and the other receives and gives back (Blau, 1964; Gouldner, 1960). Moreover, when the action from the giver is positive, it is also expected that the receiver would respond in a more positive way (Eisenberger et al., 1987; Gouldner, 1960). Similarly, when the actions are negative or perceived as such, the receivers tend to react more negatively. In a workplace setting, these social interactions occur with managers, colleagues, clients, and other potential stakeholders who are involved with the organisation. However, social exchange theory does not explain how the dynamic of these workplace interactions contribute to employee wellbeing. To understand how workplace

relationships and interactions contribute to employee wellbeing, it is necessary to create a body of knowledge that supports the wellbeing of staff.

1.2.2. Emotional contagion theory

Emotional contagion theory was developed under the umbrella of social exchange theory. This theory explains how people tend to "catch" others' emotions and converge emotionally with them (Hatfield et al., 1993). This emotional conversion also occurs during workplace interactions. Employees are not isolated elements of a company or organisation; instead, they normally work in teams and interact with others on a daily basis (Petitta et al., 2019). A work team is a group of employees that work together to achieve specific goals and under the same direct manager (Costa, 2003). These interactions between team members can be face-to-face or online, and according to emotional contagion theory, they have an influence on employee wellbeing. Catching colleagues' emotions has important implications for workplaces (Xerri et al., 2022). Some argue that employees who have high levels of emotional contagion and work with colleagues who experience lower levels of wellbeing tend to have a decreased level of employee wellbeing (Xerri et al., 2022). In contrast, employees who have high levels of emotional contagion and belong in a group with high levels of wellbeing tend to experience even higher levels of wellbeing. To understand how the team influences employee wellbeing, it is necessary to understand the synergy between team members and to what extent this emotional contagion impacts the level of wellbeing. Chapter 2 further explores the dynamics and propositions of emotional contagion theory.

1.2.3. Conservation of resources theory

Conservation of resources theory is a resource-oriented theory that explains how people protect and build resources that reduce their levels of stress and, therefore, increase their levels of wellbeing. A stressful situation is understood as a threat of loss to someone's resources, and according to conservation of resources theory, people tend to protect their current resources. While in a positive and nurturing situation, people tend to try to acquire new resources that can reduce their stress levels and improve their wellbeing. In a workplace, the literature argues that the resources that have a higher impact on stress reduction and employee wellbeing are those provided by the organisation and their direct managers (Bookwala & Fekete, 2009). These resources include managerial priority (Hall et al., 2010), and managerial support (Shi & Gordon, 2020). By studying the resources that employees strive to retain, protect, and build, it should be possible to identify the antecedents of work stress and employee wellbeing. Researchers have focused on conservation of resources theory to explain the individual antecedents of employee wellbeing and further develop organisational interventions to support wellbeing. However, there is a clear gap regarding how emotional contagion between colleagues shapes the impact of these interventions. While this thesis tries to conceptualise the collective influence on wellbeing, it is also necessary to understand how these organisational resources impact stress and wellbeing levels.

1.3. Organisational resources

According to conservation of resources theory, two organisational resources are argued to be antecedents of employee wellbeing: first, the employee's perception of how management prioritises their wellbeing above organisational profitability and second, the employee's perception of their relationship with their direct manager and how supportive this relationship is. Organisational scholars argue that these two organisational resources are important contributors in reducing work stress and boosting employee wellbeing (Farr-Wharton et al., 2022a; Xerri et al., 2022).

1.3.1. Managerial priority: Psychosocial safety climate

Psychosocial safety climate is a concept developed by Dollard and Bakker (2010) that conceptualises to what extent employees perceive their organisation protects and prioritises their wellbeing, health, and safety. According to Bailey et al. (2015), psychosocial safety climate is an important resource to consider when implementing interventions to reduce work stress and support employee wellbeing. Moreover, psychosocial safety climate is understood as an important source of organisational resources for employees. Conservation of resources theory states that if employees perceive there to be a high level of psychosocial safety climate in their workplace, they are more likely to strive to protect those resources. However, some researchers argue that managers act as organisational agents and without them, the influence of psychosocial safety climate on employee wellbeing is not as successful (Shi & Gordon, 2020).

1.3.2. Managerial support: Leader-member exchange

Managerial support is essential in the relationship between psychosocial safety climate and employee wellbeing (Shi & Gordon, 2020). Leader-member exchange, as conceptualised by Graen et al. (1982), refers to the quality of the relationship between a manager and an employee, encompassing the employee's perception of the support, trust, communication, and mutual exchange of resources with their manager. Brunetto et al. (2011) argue that organisational support and resources are accessed by employees through their direct manager. Therefore, the relationship supervisor-employee is an important antecedent of wellbeing as it is the source of organisational resources for employees.

1.4. Research rationale

There is an increasing interest in employee wellbeing, its measurement, and its enablers. Additionally, since July 2022, organisations in Australia must mitigate the impact of work on wellbeing (Safe Work Australia, 2022). Yet, consensus has, to date, not been achieved on a definition or measurement of this construct (Deci & Ryan, 2008; Dodge et al., 2012). Moreover, most of the interventions that have been developed target the individual level of employee wellbeing without accounting for the collective influence on it. Probably because of this lack of understanding of how the team influences employees' wellbeing, the gap between knowledge of its importance and readiness for action is significant (Volini et al., 2020). Although many companies consider employee wellbeing essential for their success, very few organisations feel ready to work on it due to the lack of understanding of how to support wellbeing with meaningful programs in their strategy (Volini et al., 2020). The lack of a definition and measurement poses ongoing problems for the development of robust theory and practice concerning the role of organisations in supporting employee wellbeing. Moreover, there is little evidence of scholarly work associated with the collective conceptualisation of employee wellbeing (Brunetto et al., 2020; Farr-Wharton et al., 2021). Most scholarly papers analysing employee wellbeing use a single-level approach. Yet, according to multilevel organisational theory, employees who belong to the same work team are more alike than those from different teams (Kozlowski & Klein, 2000). This similarity between employees who belong to the same work group has important implications for the statistical analysis and its results. Yet, most of the research examining wellbeing at work has ignored this crucial cluster effect.

1.5. Research aims and research questions

The study examines the influence of organisational resources and co-workers on employee wellbeing. The overall objective is to develop a more nuanced understanding concerning the collective drivers of wellbeing, and in so doing, the thesis seeks to move the current discourse beyond its almost exclusive focus on individual mental disposition. The research questions (RQ) for this study are included in Table 1 Error! Not a valid bookmark self-reference..

RQ1	Does emotional contagion among team members influence the levels of individual wor	
	stress and employee wellbeing?	
RQ2	To what extent do organisational resources impact the level of employee wellbeing?	

Table 1. Research questions

The two research questions presented above are the drivers of this study, as each will inform the central research aim of identifying how the team impacts employee wellbeing: RQ1 attempts to analyse whether or not emotional contagion theory influences work teams. RQ2 studies the relationship between organisational resources and their impact on employee wellbeing. The resources under investigation in this thesis are the employees' perception of managerial priority and support.

1.6. Significance and contribution to practice

With this thesis, I aim to contribute to the body of scholarly knowledge that is associated with employee wellbeing. This study intends to contribute to the literature by using a multilevel postpositivist approach to conduct an analysis of how the team influences employee wellbeing. I plan to establish a better understanding of wellbeing and, more specifically, employee wellbeing, as well as a deeper awareness of its contagious effect. I anticipate making further contributions to help develop healthier and better workplaces, where employee wellbeing is a crucial aim of every organisation. Furthermore, I hope that my contribution can assist with the development of workplace interventions to assist with the maintenance and acquisition of better levels of collective wellbeing – in particular, around the specific design of interventions incorporating a more nuanced conceptualisation of wellbeing and its influences, and in the management and leadership of work-teams. Moreover, this thesis aims to contribute to the understanding of the key role of managers and, in particular, line managers, as "transporters" of organisational resources. This is particularly important since line managers do not tend to be human resources experts and are usually promoted on their technical skills and not their interpersonal skills.

An improved understanding of employee wellbeing and its antecedents is likely to contribute to a reduction in the current rise of work-related stress, anxiety, and depression. Potentially, the findings of this study could assist in the reduction of work-related suicide rates, which have increased during the pandemic (Podubinski & Glenister, 2021). For these reasons, the findings and conclusions of this thesis could be of great significance given the current world situation. The COVID-19 pandemic has caused many workplace changes, such as the increase in flexible work arrangements. Moreover, the current "War for Talent", with the increasing competition between recruiters and the prominent skill shortage that the Australian economy is currently facing, adds extra pressures for managers and organisations (Goldstone et al., 2021; Serenko, 2022). Among those pressures, some organisations try to create a competitive advantage and be at the forefront of the employee value proposition. This study aims to provide practical recommendations for supporting employee wellbeing and reducing work stress. Additionally, the research tries to offer a more in-depth theoretical insight as to how organisations can contribute to the wellbeing of their employees by analysing how team members influence each other and the organisational antecedents of employee wellbeing. As such, the research uses a theoretical blending of three different and complementary theories to integrate plausible reasoning and provide new insights while avoiding "silos of knowledge" (Cornelissen & Durand, 2012). Therefore, this thesis aims to contribute to the development of three theories while using multiple lenses to frame the research.

Therefore, the innovation and significance of this study is that it seeks to understand if one employee's perception of their wellbeing can shape their team members' perceptions of wellbeing. Thus far, the majority of studies that examine employee wellbeing have been almost solely at the individual level of analysis. This is problematic because organisational scholars using a qualitative lens assert the existence of a collective (team-based) effect that can occur when individuals in a team have, for example, high perceptions of wellbeing (Potter et al., 2019). Until the advent of multilevel statistical analysis, finding evidence for this assertion using quantitative methods was impossible. Now that statistical and computational tools have become more advanced, to the point where they can factor in individual and team-level data points (and beyond), pursuing this study is timely and essential for advancing the conceptualisation of theory under the broader banner of *positive organisational scholarship*.

1.7. Structure of the thesis

This thesis comprises six chapters. The present chapter has provided an introduction to this thesis. It commenced with an overview of the context and the research gap. This is followed by a presentation of the theoretical underpinnings that frame this thesis. The chapter also introduces the two organisational resources that are considered influencers of employee wellbeing. It continues with the research rationale, along with this study's research aims and research questions. The chapter concluded with the significance and contribution to practice.

Chapter 2 offers a review of the extant literature examining the theoretical and conceptual frameworks of this research project. It presents social exchange theory as the overarching framework for this study, as well as emotional contagion theory, which falls under the umbrella of social exchange theory. Moreover, it introduces conservation of resources theory to understand if employees protect and maintain their resources to reduce their levels of job stress and boost employee wellbeing. Additionally, multilevel organisational theory is referred to as it assists in the development of the methodology, which is presented in Chapter 3. Multilevel organisational theory introduces the need for a two-level study, where the work team influence is considered, and employees are not studied as if they were isolated units. The chapter also explores the literature on work stress to understand its relationship with employee wellbeing. To conclude, the chapter presents the two organisational antecedents that are included in this thesis. First, managerial priority as psychosocial safety climate, and second, managerial support as Leadermember exchange.

Chapter 3 provides a detailed presentation of the methodology that this study has adopted. It first presents the approaches to employee wellbeing research and how it has been studied as it has garnered increasing attention. The chapter continues with a search of the different research

paradigms or philosophies that may help to inform the study. This thesis has been structured under a post-positivist approach. To answer the research questions, this study used quantitative methods and collected survey data from 237 employees within 41 different work teams from Australian professional services organisations. Following the research design, Chapter 3 includes some preliminary analysis of results such as a normality test, multilevel structural equation modelling, multilevel exploratory factor analysis, multilevel confirmatory factor analysis, reliability and validity of the utilised scales, multicollinearity analysis, and model fit. It concludes with the ethics and risks that have been taken into consideration.

Chapter 4 details the results of the data analysis. It begins with the results of the measurement model with the multilevel exploratory factor analysis and the multilevel confirmatory factor analysis. To continue, the chapter presents the results of the multilevel structural model with the description of the used estimator (Bayes), the model fit and the results of the hypothesis testing.

Chapter 5 provides and extensive discussion on the results presented in Chapter 4. It presents an overview of the research questions and hypothesis as well as this thesis' contributions and implications for theory, research, practice and policy.

Chapter 6 presents an overall conclusion of this thesis. Lastly, the chapter presents the results obtained through the extensive data analysis process. The data was analysed using multilevel structural equation modelling. This chapter has a particular focus on the important findings of this thesis as well as an overview of its limitations and potential areas for further research.

1.8. Conclusion of the chapter

The introductory chapter begun with a presentation of the context of this thesis. This section explored the current research gap and the need for a collective conceptualisation of employee wellbeing. The section explored the concept of employee wellbeing and why it is receiving increasing attention from organisations, managers, employees and researchers. The theoretical framework of this study was then presented. Three theoretical underpinnings frame this thesis. First, social exchange theory serves as the overarching theory and below it are emotional contagion theory and conservation of resources theory. The intersection between these three theories serves as the theoretical framework of this research project (see Figure 1). The next section of this chapter presented the organisational resources under analysis. Managerial priority and support are considered key organisational resources that work as antecedents of employee wellbeing. However, it is not known to what extent co-workers' emotions can positively impact the influence of managerial priority and support. The chapter continued with the presentation of the research rationale and discussed the research aims and the two research questions. The chapter finished with the significance of this study and its contribution to knowledge and practice as well as an overview of the structure of this thesis.

Chapter 2. Theoretical and Conceptual Frameworks

This section explores the existing literature on the theoretical and conceptual frameworks that support this study. The main aim of this thesis is to investigate the team influence on employee wellbeing. The nuances of human relations impose a particular complexity when analysing the impact of colleagues and management on employees' wellbeing. As such, this study draws on the blending of three different theories: social exchange theory, emotional contagion theory, and conservation of resources theory. The goal of this chapter is to frame the knowledge gap that defines the research questions of this study. The chapter continues with an in-depth exploration of the current literature that builds the conceptual framework. The aim of this conceptual framework is to define and develop an understanding of the phenomenon to be studied and assist with the development of the theorised model and the hypothesis. This thesis primarily focuses on the contagion effect between work stress and employee wellbeing. Accordingly, this forms the foundation for the first hypothesis. The following hypotheses develop the antecedents of employee wellbeing. Another theory has been included in the theoretical framework because it drives the methodology, i.e. multilevel organisational theory. Team members from the same work team tend to be more alike than those from other teams. This last theory helped with the development of the methodology for this study as it uses a multilevel statistical analysis approach, within a post-positivist research philosophy. According to multilevel organisational theory, teams are multilevel systems that need to be studied as a whole rather than be analysed member-bymember as most organisational studies are. To conclude this chapter, there is a contextual overview of the professional services industry in Australia. All participants in this study belong to the professional services industry, and this has particular implications for the results that are presented and discussed in Chapters 4 and 5.

2.1. Understanding workplace relationships and behaviours

The relationship between managers and employees, along with the relationship between coworkers can be studied as any other interpersonal relationship. These relationships have important implications for the individual as well as the team performance, and organisational outputs. For employees, some argue that positive workplace relationships and a perceived safe environment can reduce work stress (Farr-Wharton et al., 2022a) and enhance employee wellbeing (Xerri et al., 2022). While for organisations, nurturing workplace relationships contribute to higher performance and outputs (Jankelova & Joniakova, 2021).

As such, understanding workplace relationships and behaviours is key for organisations' success. According to Otamori and Igodo (2022, p. 57), "Employees are motivated to develop positive relations at the workplace". Nonetheless, these positive relationships are not easy to build and can be impacted by unpredictable challenges (Mitchell et al., 2012). Social exchange theory is a

key theoretical framework which can be drawn on to better understand workplace relationships and behaviours (Porter, 2018).

This thesis is guided by social exchange theory (Blau, 1964) to examine the impact of managers and colleagues on employee wellbeing. Social exchange theory has been proven to be an important construct for understanding workplace behaviours (Cropanzano & Mitchell, 2005). However, social exchange theory is more akin to a broad conceptual paradigm than a theory itself (Porter, 2018) and is frequently used as an overarching framework for several other theories. Many researchers even refer to social exchange theory as a family of multiple conceptual models, as it is one of the most enduring and widely used frameworks (Cropanzano & Mitchell, 2005; Mitchell et al., 2012; Porter, 2018). This endurance similarly applies to the two other theories that frame this study, the emotional contagion theory and conservation of resources theory (see Figure 2). Combined, these three theories provide a bespoke theoretical framework to address the research aims of this thesis.





Note. The theoretical framework of this study is the intersection between three theories: Social exchange theory, emotional contagion theory, and conservation of resources theory.

A summary of the theoretical background of this research project has been provided in Table 2. Both emotional contagion theory (Herrando & Constantinides, 2021) and conservation of resources theory (Ng & Feldman, 2012) can be perceived as components of social exchange theory, as they suggest that social life involves a sequence of transactions between two or more agents. However, to develop a better sense of the overall theoretical framework underlying this thesis as well as the current knowledge gap, this section will explore social exchange theory in more depth. Moreover, according to Kozlowski and Klein (2000), since organisations are multilevel entities, organisational theories need to incorporate multilevel dynamics with data structured in several levels. The dynamic of this data must consider its multilevel structure, where employees are nested within work teams (Kozlowski & Klein, 2000). Emotional contagion theory draws on the multilevel dynamics between those agents that closely interact on a frequent basis (see section 2.2.). However, since the aim of this research project is to study the impact of co-workers and managers on employees' wellbeing, this section comprises the analysis and presentation of social exchange theory as the general framework. The main aim of this section is to understand workplace relationships and behaviours to be able to subsequently analyse how these interactions influence the wellbeing of employees. The following sections then develop the details of the two other theories, with section 2.2. exploring the collective influence on employee wellbeing and section 2.3. identifying the organisational resources that influence wellbeing.

Theory	Application	References
Social exchange theory	This theory helps to understand	(Cropanzano &
	workplace behaviours and relationships.	Mitchell, 2005; Mitchell
		et al., 2012)
Emotional contagion	This theory helps to understand how	(Hatfield et al., 1993;
theory	employees tend to synchronise emotions	Sy et al., 2005)
	and feelings with co-workers. In other	
	words, how co-workers influence	
	employee wellbeing.	
Conservation of	This theory helps to identify the social	(Fritz & Sonnentag,
resources theory	influencers of employee wellbeing	2006; Hakanen et al.,
	within an organisation and to understand	2018; Halbesleben et
	how employees value these resources.	al., 2014)

Table 2. Summary of the theoretical framework of this study

Although social exchange theory has many variants, the most contemporary and generic model involves one party providing a benefit or doing harm to another party and follows with the other party responding. This is a reciprocal dynamic in which one party gives and another receives and gives back (Blau, 1964; Gouldner, 1960). Social exchange theory predicts that in reaction to positive initiating actions, the receiver will tend to reply in a more positive way (Eisenberger et al., 1987; Gouldner, 1960). In contrast, in reaction to negative initiating actions, the receiver will tend to reply more negatively. In a workplace setting, these interactions generally occur between co-workers, with clients, or between supervisors/direct managers and employees (Mitchell et al., 2012; Vine, 2004). High-quality social exchange relationships that provide positive value include

resources such as managerial priority, (Xerri et al., 2022), managerial support (Riggle et al., 2009) or training and development programs (Khan & Iqbal, 2020; Ocen et al., 2017). On the other hand, low-quality social exchange relationships that provide negative value include aspects that negatively affect the organisation and the team as well as the individual. Examples include mobbing, which is a sociological term that refers to the bullying of an employee by a group (Cropanzano et al., 2016; Medina-Craven & Ostermeier, 2021) or unachievable targets set by the team or the organisation (McLeod et al., 2021; Worku, 2019).

The quality of any interaction can be influenced by the relationship between the parties (Blau, 1964). In most cases, relationships with greater trust tend to be considered a valuable resource or provide more of a benefit to the parties, while relationships marked by distrust or fear tend to have a lower value and may end sooner (Cropanzano et al., 2016; Meira et al., 2022). In this context, studies suggest that there are two important relationships that must be studied to recognise how to support and promote employees' wellbeing from the organisational side (see Table 3). First, the relationship between the employee and the supervisor or line manager (Graen & Uhl-Bien, 1995; Hu & Liden, 2013). Second, this theory also suggests that relationships amongst co-workers can have an influence on employee wellbeing (Benitez et al., 2019). However, there are few studies that have analysed how these relationships influence employee wellbeing, and there are even fewer that consider a multilevel perspective (Terpstra-Tong et al., 2020).

Relationship	Relationship dynamics	References
Employee-	A positive relationship with a supervisor can	(Graen & Uhl-Bien,
supervisory	contribute positively to the wellbeing of the	1995; Hu & Liden,
	employee, while a negative relationship with a	2013)
	supervisor increases the chances of turnover,	
	burnout, and work stress.	
Between co-	Positive relationships with other co-workers can	(Benitez et al., 2019)
workers	contribute positively to employee wellbeing and	
	influence their perception of organisational and	
	managerial practices.	

Table 3. Workplace relationships under analysis within the social exchange theoretical framework

On the one hand, the quality of relationships between the employee and the supervisor or direct manager is considered an important influencer on work stress and employee wellbeing (Graen & Uhl-Bien, 1995). Social exchange theory suggests that a more positive relationship with a supervisor will contribute positively to the wellbeing of the employee (Hu & Liden, 2013). Building on the idea that the perception of supervisorial practices and policies influences

employee wellbeing, social exchange theory proposes that employees tend to respond with higher performance and extra-role behaviours which are not part of their formal job requirements when they perceive that the organisation is investing and believing in them (Cropanzano & Mitchell, 2005). However, social exchange theory does not expand on how employees respond to these organisational social resources. To further develop this proposition, it is essential to examine the team resources that influence employee wellbeing and that can be supported by the organisation. Many positive organisational researchers point out that this has been developed using the conservation of resources theory (e.g., Hakanen et al., 2018; Halbesleben et al., 2014; van Woerkom et al., 2016; Xerri et al., 2022). Social exchange theory is limited in its ability to explore what is considered a benefit or a cost when maintaining or finalising a relationship. For this reason, organisational resources that impact employee wellbeing are analysed in section 2.3. along with an exploration of conservation of resources theory.

Similarly, social exchange theory also suggests that relationships amongst co-workers can have an influence on employee wellbeing (Benitez et al., 2019). Studies suggest that a positive relationship with other co-workers can influence an employee's relationship with their supervisor and how managerial practices are perceived by employees (Bartel & Saavedra, 2000; Benitez et al., 2019). The understanding of the relationships amongst co-workers using the lens of social exchange theory explains human behaviour and the social structure of relationships from a sociological and psychological perspective (Meira et al., 2021). Yet again, social exchange theory does not provide a full understanding of how the dynamics of these interactions can influence one's wellbeing levels. Studies that use the social exchange theoretical framework to understand the relationship between co-workers have mainly analysed those behaviours from the individual perspective (e.g., Hussain et al., 2020; Salem et al., 2022), without considering that employees belong to a team and that the team may also have an influence on its employees' wellbeing. Moreover, one of the main criticisms of this theory is that it is so broad that it can explain almost any type of relationship, at least post hoc (Cropanzano et al., 2016). It is for this reason that within this thesis, emotional contagion theory is also drawn on to study the team antecedents of employee wellbeing. Emotional contagion theory is a multilevel theory that explains how humans tend to synchronise feelings, emotions, and beliefs with those around them (Hatfield et al., 1993). In the workplace, those around them would be in general, other co-workers, and therefore, it is also important to consider this effect as a moderator between work stress and self-perceived wellbeing levels. The emotional contagion theory will be further explored in section 2.2.

Social exchange theory describes relationships as result-oriented social behaviours. The theory predicts that people choose to enter and maintain relationships in order to maximise the benefits of these relationships while minimising the costs (Blau, 1964; Cropanzano & Mitchell, 2005). It

is similar to the cost-benefit economic analysis that businesses use to analyse which decisions to make and which to forgo. However, in terms of relationships and their cost-benefit analysis, researchers are describing their observations when analysing human actions and not arguing that an economic metaphor should be used to make these decisions (Cropanzano & Mitchell, 2005). In the workplace, employees are observed to make decisions based on the prospective outcomes and to consider the costs (Emerson, 1976). Costs from a relationship can include anxiety, stress, embarrassment, and time, while benefits can include an increase in wellbeing, engagement, achievement, or finding a sense of belonging.

When evaluating the rewards that an employee receives from their organisation through the lens of social exchange theory, Tsui et al. (1997) suggest that support for employee wellbeing and investment in employees' careers are essential. As such, relationships between employees, managers, and co-workers are fundamentally important. If this relationship is one of high-quality social exchange, it is theorised to lead to trust (Cropanzano & Mitchell, 2005), low levels of job stress (Farr-Wharton et al., 2022a), and high levels of employee wellbeing (Danna & Griffin, 1999; Milner et al., 2015; Xerri et al., 2022). However, if the relationship does not become beneficial, it may end, leading to higher turnover and burnout rates (Danna & Griffin, 1999). Studies suggest that the best workplace relationships are those with a mutual investment approach i.e., both employees and their employers invest in the relationship (Tsui et al., 1997). These relationships are, in general, also long-term relationships that provide greater benefits for all parties involved (Blau, 1964).

Social exchange theory has received several points of criticism that cannot be ignored. The main criticism is that the theory is so broad that it could likely explain any interaction (Cropanzano et al., 2016). Conservation of resources and emotional contagion theories help in overcoming this limitation. Some other criticisms of social exchange theory include the lack of clarity when developing constructs (Cropanzano et al., 2016). This challenge has been addressed in this study by further framing the study with other theories that are more specific and build additional pillars for this research. One last important limitation that has been mentioned in the literature is the assumption of bipolarity (Martinescu et al., 2021; Medina-Craven & Ostermeier, 2021). In this case, bipolarity is due to the assumption that positive actions or emotions are the same as the absence of negative actions or emotions. One of the main reasons for this assumption is that negative emotions are more prominent in the literature as these are the ones that have kept humanity alive (Ekman et al., 1983; Harker & Keltner, 2001). This criticism of social exchange theory is addressed in section 2.4. in the discussion of the conceptual framework where employee wellbeing is defined as a construct itself and not as the absence of stress or psychological distress.

2.2. The collective influence on employee wellbeing

Employees are not isolated units within a company – they belong to a team or a department, and the interaction between employees has an impact on employees' behaviours and emotions (Petitta et al., 2019). Several studies have analysed how team members can develop mutually shared emotions and moods when working together (e.g., Barsade, 2002; Totterdell et al., 1998; Xerri et al., 2022). In this study, emotional contagion theory helps to explain how teams influence members' individual and collective wellbeing. The emotional contagion theory is a multilevel model that defines emotional contagion as "the tendency to automatically mimic and synchronise movements, expressions, postures, and vocalizations with those of another person and, consequently, to converge emotionally" (Hatfield et al., 1993, pp. 153-154). Many studies have observed that as soon as people are born, they tend to mimic and synchronise their emotional expressions, both explicitly and implicitly, with those of others (Hatfield et al., 1993; Hoffman, 2002).

Work teams (this thesis refers to work teams as teams) can be defined as a group of employees with complementary skills who work together to achieve specific goals and under the same direct manager (Costa, 2003). Most workplaces are encouraging teamwork, as it has several benefits (Bartel & Saavedra, 2000; Staines, 2010). Numerous studies in different disciplines such as biology, social science, and psychology hold that individuals function better in groups than by themselves (Burgoon et al., 1995; Caporael, 1997). Organisational researchers have argued that emotional contagion between team members can improve group performance and organisational goals (Bartel & Saavedra, 2000; Schneider et al., 2013; van Maanen & Kunda, 1989).

There are several mechanisms associated with emotional contagion. Early studies suggested that emotional contagion included conscious reasoning and analysis, yet more recent studies have found that this process is more automatic and subtle than previous ones had suggested (Hatfield et al., 1993). Hatfield et al. (1993) suggested three propositions to support the idea that emotional contagion is more ubiquitous and automatic than what was suggested earlier. These include mimicry, feedback, and contagion, which are summarised in Table 4. Mimicry is the first step and happens when people automatically mimic the movements of those around them. In the workplace, this synchronisation has important implications. Several studies point out that if an employee is picking up others' emotions and mannerisms, it may indicate that they are comfortable and engaged in their job (Bartel & Saavedra, 2000). In the second phase, feedback happens when people make inferences about their own emotions and adapt them to the movements and behaviours they are mimicking. The link between facial expressions and emotions is, in general, spontaneous. For instance, when people display a facial expression of sadness or anger, they are likely to feel the emotions associated with those expressions (Ekman et al., 1983). Finally, contagion happens when "from moment to moment, people tend to catch other people's emotions" (Hatfield et al., 1993, p. 99). In a workplace environment, these emotions can be from other colleagues, supervisors, managers, or clients. This can have both positive and negative implications. For instance, if an employee's colleagues are experiencing high wellbeing, the employee may also catch it. However, more frequently than not, employees with high levels of emotional contagion tend to converge emotionally with those that are at their same level such as colleagues or co-workers (Xerri et al., 2022).

Proposition	Description	Work Implications	References
Mimicry	In conversation, people automatically and continuously mimic and synchronise their movements and instrumental behaviours with those of other people.	Employees spend a significant amount of time at work, and the people they work with have a considerable influence. If an employee is picking up others' emotions, it may indicate that they are comfortable and engaged in their job.	(Bartel & Saavedra, 2000; Hatfield et al., 1993; Lee & Pennings, 2002)
Feedback	Subjective emotional experience is affected moment to moment by the activation of and feedback from facial, vocal, postural, and movement mimicry.	As employees mimic others around them, they are also likely to experience the emotion associated with the mimicry.	(Ekman et al., 1983; Hatfield et al., 1993)
Contagion	Consequently, people tend, from moment to moment, to catch others' emotions.	In a workplace environment, people tend to catch others' emotions. These can be from other colleagues, supervisors, managers, or clients. This can have either a positive or negative impact. For instance, if an employee's colleagues are experiencing high levels of wellbeing, the employee may also catch them.	(Barsade, 2002; Hatfield et al., 1993; Petitta et al., 2019)

Table 4. Propositions of emotional contagion theory

According to Hatfield et al. (1993), there are two types of emotional contagion, which are summarised in Table 5: implicit and explicit. Implicit emotional contagion is the most common type, as it happens automatically and unconsciously. This process happens following the three propositions described above. According to Schoenewolf (1990), the speed of the emotional contagion depends on the energy used when expressing the emotion. In the workplace, this contagion is common not only with face-to-face conversations but also with phone calls, emails, and other forms of communication. On the other hand, direct or explicit emotional contagion happens when people's emotions are being explicitly manipulated to be converted to a specific emotion. This may happen, for example, if a leader wishes to motivate the team and gives a

motivational speech or offers a reward or treat to encourage team members. In the workplace, this manipulation has been studied for emotional labour, which, according to Hochschild (2012; 2013), is defined as positions that require the employee to display certain emotions to meet the requirements of the job. Some examples of positions requiring emotional labour include customer service employees, servers, nurses, and childcare workers. However, this type of emotional contagion has only been studied for those employees that engage in emotional labour, but not in an industry such as the professional services. Yet, some suggest that these studies would benefit both the academic literature and organisational practices (Xerri et al., 2022).

Туре	Description	Work Implications	References
Implicit	Automatic and	Implicit emotional contagion	(Schoenewolf,
	unconscious contagion of	happens in the workplace with	1990; Tee,
	emotions.	face-to-face conversations and	2015)
		with other forms of	
		communication, such as phone	
		calls or emails.	
Explicit	This happens when	Direct emotional contagion is	(Hochschild,
	people's emotions are	common in emotional labour	2012; 2013;
	being explicitly	positions, such as nursing or	Tee, 2015)
	manipulated to be	childcare.	
	converted to a specific		
	emotion.		

Table 5. Types of emotional contagion

According to Bartel and Saavedra (2000), several resources determine the level and magnitude of emotional convergence in an organisation and more specifically, in a work team. Bartel and Saavedra argue that "convergence in members' moods and emotions is determined by task and social interdependence, membership stability, and mood-regulation norms" (2000, p. 197). Task and social interdependence refer to particular duties that require high levels of social interactions. When this interdependence happens, employees are more likely to converge emotionally. Membership stability happens with group members who are familiar with each other and who have worked together for a period of time. This stability could also increase the emotional contagion levels. Finally, Bartel and Saavedra (2000) suggest that catching others' emotions also depends on social norms. Social norms are unwritten rules and behaviours that are considered appropriate in a particular setting. These norms could also influence employees in converging emotionally with co-workers.

Research on workgroups demonstrates that performance and engagement improve when team members mimic and synchronise their emotions, feelings, and behaviours (Hackman, 1992). Regardless of these facts, few studies have examined employee wellbeing using a multilevel approach from the emotional contagion theory perspective. A summary of these studies and their main findings is presented in Table 6.

Implications for employee wellbeing		Reference
Employee well-being profiles and service quality: A unit-		(Benitez et al., 2019)
level analysis using a multilevel latent profile approach		
How store managers' employee climate perceptions affect	No	(Deitz et al., 2021)
frontline employee, customer and store performance		
outcomes: An examination in the small-store setting		
The development of a model of employee wellbeing that	No	(Ilies et al., 2007)
considers both personal and situational predictors and		
both work and nonwork wellbeing indicators.		
Positive implications of family management in contrast to	Yes	(Kammerlander et al.,
other types of management on employee wellbeing.		2017)
How fairness assessments and organisational structure	Yes	(Spell & Arnold, 2007)
relate to employee wellbeing and mental health.		
How coaching leadership and positive management boost		(Wang et al., 2017)
employee wellbeing with the mediating effect of		
psychological capital.		

Table 6. Summary of empirical literature examining employee wellbeing from the emotional contagion theory perspective

Note. MLA is multilevel analysis

The third proposition of emotional contagion theory proposes that people tend to converge emotionally with the people they have some contact. Following this proposition, employees tend to catch emotions from co-workers. Therefore, it could be argued that emotional contagion moderates the negative relationship between work stress and employee wellbeing. Several studies suggest that there is a moderating effect of co-workers' emotional contagion on employee wellbeing for employees that engage in emotional labour (Xerri et al., 2019; 2022). Emotional labour refers to those employees who need to manage emotional expressions with other parties as part of their professional role (Parks & McKay, 2022). Some examples of industries in which their employees are engaged in emotional labour include public administration, health care, hospitality and education (Wharton, 2009). Although this moderation effect that influences employee wellbeing has been partially tested with employees who engage in emotional labour, this research project expects to replicate similar findings for employees within the professional services industry in Australia and thus, hypothesises the following (see Figure 5 for the full hypothesised model):

• Hypothesis 1. Emotional contagion moderates the relationship between work stress and employee wellbeing

This hypothesis aims to answer the first research question which asks if emotional contagion among team members has an influence on work stress and employee wellbeing. This is a crosslevel moderation between the emotional contagion at the team level that influences the employeelevel relationship between work stress and employee wellbeing. This positive moderation effect of emotional contagion could assist employees with higher levels of work stress if they are in a supportive and positive team. Emotional contagion theory proposes that employees with high levels of work stress and emotional contagion could see a less dramatic impact on their wellbeing levels if they are surrounded by colleagues with high levels of employee wellbeing (Xerri et al., 2022). In contrast, co-workers with low levels of wellbeing could negatively impact employees with high levels of emotional contagion as they would experience even lower levels of employee wellbeing.

2.3. Identifying organisational resources that influence employee wellbeing and job stress

The antecedents of employee wellbeing have been studied through different theoretical lenses. While this thesis has used conservation of resources theory, as will be explained shortly, it is necessary to mention that some have also drawn upon the theoretical lens of job demands-resources theory. Job demands-resources theory developed by Bakker and Demerouti (2014) explains the dynamic between the job demands of a particular position and the resources (e.g., organisational, personal or others) available to the employee. In summary, job demands-resources theory argues that when job demands are high and the employee does not have many resources to draw upon, they are likely to experience higher levels of stress, which is likely to cause low levels of employee wellbeing (Bakker & Demerouti, 2014). However, the second research question guiding this thesis asks to what extent organisational resources impact the level of employee wellbeing. As such, it is necessary to go back a step to understand how employees' value organisational resources and which ones are considered essential.

Conservation of resources theory has often been used as a stress theory in contexts where a loss of important resources causes instability and, therefore, stress (Halbesleben et al., 2014; Hobfoll, 1989; 2002). However, positive organisational researchers have recently been using this theory to explain how people protect their resources to maintain or improve their wellbeing (e.g., Hakanen et al., 2018; Halbesleben et al., 2014; Neto et al., 2016; Xerri et al., 2022). In this thesis, the conservation of resources theory is used to identify which team resources influence employee wellbeing and to analyse to what extent they influence it. It is necessary to understand in what manner the theory conceptualises the employees' consideration for team resources. This understanding will assist in identifying the resources that would theoretically influence the wellbeing of employees. Identifying these resources is essential before studying the social
relationship between employees and their supervisors or amongst co-workers. This exploration assisted in the construction of the research model that is used in this study.

The conservation of resources theory is a resource-oriented model "based on the supposition that people strive to retain, protect, and build resources and that what is threatening to them is the potential or actual loss of these valued resources" (Hobfoll, 1989, p. 513). Resources are anything that people value (Halbesleben et al., 2014; Hobfoll, 1989). In the workplace, resources comprise all of the things employees value in their position and job. The value of each resource varies amongst employees and depends on their personal experiences and life circumstances (Halbesleben et al., 2014). For instance, high managerial support may be a valued resource to one employee but a threat to another. Thus, it is important to study each team separately when analysing how they value their resources. The reason for this approach is that employees within a team are more alike than employees from different teams (Bakker, 2015; Ilies et al., 2007; Klein et al., 1994), and, therefore, they are also more likely to value resources similarly. This is theorised in the multilevel organisational theory touched upon later (see section 2.6.), but first, it is necessary to understand how resources are structured and which are considered social and organisational resources.

Maslow (1968) proposed that people seek resources hierarchically: physical resources first, then social resources, and, finally, psychological resources. Physical resources include tangible elements that take up space and, in general, they have a monetary value. These include tangible items, from raw goods to buildings and vehicles (Abraham, 2011). Social resources exist in a person's social network and social ties (Bradford, 2015; Lin, 1983). In the workplace, these can include relationships with co-workers or supervisors. Finally, according to Hobfoll (1989), psychological resources are defined as "those entities that either are centrally valued in their own right (e.g., self-esteem, close attachments, health, and inner peace) or act as a mean to obtain centrally valued ends (e.g., money)" (2002, p. 307). These psychological resources are generally not considered team resources, as they are usually specific to the individual and not controllable by the organisation (Bookwala & Fekete, 2009). Examples of team resources examined in the literature over the past decades can be found in Table 7. These are classified according to their category (physical, social, or psychological).

Category	Resource	Reference(s)
Psychological	Safety climate	(Dollard & Bakker, 2010; Kalshoven & Boon, 2012; Selenko et al., 2013; Farr-Wharton et al., 2022a)
Social	Supervisorial support	(Lee & Ashforth, 1996; Neveu, 2007; Wilson, 2010)

Table 7. Organisational and team resources by category and kind

Physical	Organisational	(Kalshoven & Boon, 2012)
	facilities	

According to the conservation of resources theory, organisational and team resources reduce work stress which enhances employee wellbeing (Salanova et al., 2005). The literature highlights that a safety climate and supervisorial support are two of the most important resources that reduce work stress and support employee wellbeing (e.g., Dollard & Bakker, 2010; Kalshoven & Boon, 2012; Selenko et al., 2013; Farr-Wharton et al., 2022a). However, there is little research on how these resources influence employee wellbeing, and there is even less in which the influence of the work team is considered. In order to fill this knowledge gap and identify the team antecedents of employee wellbeing, this research project considers the following two resources: safety climate and supervisorial support. Both resources are exhaustively explored in section 2.5. However, before exploring the resources that operationalise this thesis, it is essential to focus on the principles of conservation of resources theory.

Organisational facilities and physical resources are identified resources that could influence employee wellbeing. However, several studies also mention that these resources only have an indirect influence on employee wellbeing (Abraham, 2011; Bosmans et al., 2015). Those studies suggest that physical resources influence two different aspects. First, physical resources have an influence on social and psychological resources. Second, they influence stress and wellbeing. If those resources have an indirect influence, they are not the primary focus of this project which tries to identify the team influencers on employee wellbeing and to measure the influence.

The conservation of resources theory has several principles with certain workplace implications, which are reflected in Table 8. The first principle is the primacy of resource loss, which suggests that a resource loss is psychologically more harmful than a resource gain (Hobfoll, 2001). According to many empirical organisational behaviour studies, a loss of resources at work can explain work stress and strain (Halbesleben & Buckley, 2004; Hobfoll, 2001), depression (Kessler et al., 2011), burnout (Shirom, 1989), cardiovascular disease (Melamed et al., 2006), high blood pressure, and excessive levels of cortisol (de Vente et al., 2003). However, more recently, positive organisational scholars have analysed how a resource gain can be highly beneficial for organisations, teams, and employees, as positive emotions are as significant as negative emotions (Seligman et al., 2005). This would mean that team resources can have a significant impact on employee wellbeing.

Emotional contagion theory also posits that in the situation of resource shortfalls or when there is a recession of resources due to an external or internal cause, resource gains will have a greater impact on the wellbeing of employees. For example, during the pandemic of COVID-19 several organisational and team resources were reduced (Podubinski & Glenister, 2021). However, according to studies that had analysed previous resource shortfalls, it can be concluded that a resource gain in this setting will have a greater impact on employee wellbeing (Halbesleben et al., 2014; Stein & Cropanzano, 2011).

The second principle is resource investment, which suggests that people tend to invest resources to avoid a loss of other resources or to gain new resources (Hobfoll, 2001). Coping is a clear example of an investment in resources to avoid further resource losses (Ito & Brotheridge, 2003; Vinokur & Schul, 2002). According to Lazarus and Folkman's (1984) coping model, people use resource investment to cope only as an attempt to limit stress. Many studies have analysed how resource losses affect work intensity, engagement, performance, emotional exhaustion, or compassion towards co-workers (Halbesleben & Bowler, 2007; Halbesleben et al., 2014; Hochwarter et al., 2008; Wheeler et al., 2013). Employee participation in decision making is considered a resource investment, as it can be a way to gain new resources, but it also incurs a risk of losing other resources (Lee & Ashforth, 1996; Neveu, 2007; Ng & Feldman, 2012).

Principle	Description	Work Implications	References
The	Resource loss is	Employment-related losses will	(Cacioppo & Gardener,
primacy of	more relevant	have a more significant impact	1999; Halbesleben et al.,
resource	than resource	than similar resource gains,	2014; Seligman et al.,
loss	gain.	causing stress, burnout, and	2005; Stein &
		several types of disease.	Cropanzano, 2011;
		Resource gains at work will have	Taylor, 1991; Kahneman
		a greater meaning in the	et al., 1982; Vinokur &
		situation of resource shortfalls.	Schul, 2002; Wells et al.,
			1997)
Resource	People tend to	Employees may want to invest	(Halbesleben &
investment	invest resources	resources to gain certain	Wheeler, 2008;
	to avoid a loss	resources and protect themselves	Halbesleben et al., 2009;
	of other	from losing others or to recover	2014;
	resources or to	after a resource loss.	Ng & Feldman, 2012;
	gain new ones.		Vinokur & Schul, 2002)

Table 8. Principles of conservation of resources theory

Within the resource investment process, Hobfoll (1989; 2001) describes four corollaries, which are summarised in Table 9. More specifically, these corollaries are used to explain the complexity of the principle of resource investment. The first corollary states that employees rich in resources are better positioned, as they can invest those resources to protect themselves from losing other resources or to gain new ones (Halbesleben et al., 2014). On the other hand, employees who do not have as many resources tend to be more conservative, as they are more likely to experience resource losses. The second and third corollaries explain opposite "spirals". Corollary 2 describes the resource loss spiral, in which an employee who loses resources tends to be more prone to

continue losing resources; on the contrary, an employee who gains resources has more options to continue gaining resources because they can invest their current resources (Hobfoll, 2001). The final corollary, resource protectiveness, states that an employee is more likely to protect their resources more defensively if they are short on resources or currently losing resources (Benight et al., 1999; Halbesleben & Bowler, 2007; Halbesleben et al., 2014; Hobfoll, 1989). In sum, these corollaries explain the order of events that occur when an individual is losing or gaining resources.

Corollaries	Description	Work Implications	References
Resource	People with	Employees with more resources	(Demerouti et al., 2004;
richness	more	are better positioned, as they can	Mäkikangas et al., 2010;
	resources are	invest current resources in gaining	Whitman et al., 2014)
	better	new resources and preventing the	
	positioned.	loss of others. On the other hand,	
		employees with fewer resources	
		tend to be at a higher risk of losing	
		their resources and experiencing	
		job burnout, stress, anxiety, or low	
		job engagement.	
Resource loss	Resource	An employee who loses resources	(Demerouti et al., 2004;
spiral	losses lead to	tends to be more prone to continue	Halbesleben et al., 2014;
	future resource	losing resources.	Hobfoll, 2001)
	losses.		
Resource gain	Resource	An employee who gains resources	(Hakanen et al., 2011;
spiral	gains advance	has more options to continue	Halbesleben & Wheeler,
	further	gaining resources because they can	2015; Mäkikangas et al.,
	resource gains.	invest their current resources.	2010; Xanthopoulou et al.,
			2009)
Resource	Lack of	An employee is more likely to	(Halbesleben, 2010;
protectiveness	resources	protect their resources more	Halbesleben & Bowler,
	leads to	defensively if they are short on	2007; Halbesleben &
	extreme	resources or currently losing	Wheeler, 2011)
	resource	resources.	
	conservation.		

Table 9. Corollaries of the conservation of resources theo	Table 9. (Corollaries	of the	conservation	of resources	theory
--	------------	-------------	--------	--------------	--------------	--------

In the conservation of resources theory, a gain in team resources can have a positive influence on employee wellbeing. Freud introduced the idea that humans tend to pursue what will increase their wellbeing and seek to create a pleasurable world (Hobfoll, 1989). Following the principles of social exchange theory, humans tend to conduct a cost-benefit analysis of their relationships and choose those that have positive outcomes. The conservation of resources theory also incorporates this idea, as people tend to retain, protect, and build resources that reduce their stress and enhance their wellbeing (Halbesleben et al., 2014). In the workplace, employees are more likely to experience a greater sense of wellbeing when they have a surplus of resources such as managerial support or a valuable relationship with their supervisor. The conservation of resources theory employees theory suggests that positive resources, such as perceived managerial priority, help employees

obtain more resources, and this leads to a positive spiral that can positively influence wellbeing (Kalshoven & Boon, 2012).

On the other hand, resource losses cause harm to employee wellbeing, causing stress. Per the second principle of the conservation of resources theory, employees tend to invest resources to prevent the loss of other resources, which causes stress. According to Lazarus and Folkman (1984, p. 19), stress is defined as "a particular relationship between the person and the environment that is appraised by the person as taxing or exceeding his or her resources and endangering his or her wellbeing". Abusive management control can trigger a quick loss of resources at work, which causes feedback avoidance to reduce emotional exhaustion and avoid further loss (Whitman et al., 2014).

There are few studies in the empirical literature exploring employee wellbeing from the perspective of the conservation of resources theory, and there are even fewer that are multilevel studies. Remarkably, all these studies are very recent; in general, most of them have been conducted in the last five years. A summary of these studies with their main findings is presented in Table 10. For instance, Bayhan Karapinar et al. (2019) used multilevel analysis to examine the link between workaholism and employee wellbeing with the moderating effect of instrumental spousal support and the mediating effect of work-family conflict. They found that instrumental spousal support acts as the primary resource of tangible support, which is invested in gaining wellbeing and reducing work-family stress. Another example is a multilevel analysis that studied the impact of authentic leadership, rewards, and meaningful work on employee wellbeing (Salmee et al., 2020). The results demonstrated the positive impact of authentic leadership, non-financial rewards, and meaningful work on employee wellbeing, while financial rewards did not always contribute to positive employee wellbeing.

Resources	Implications for employee wellbeing	MLA	Reference
Employee wellbeing	The link between workaholism and	No	(Bayhan
• Work-family time	employee wellbeing, the mediating role of		Karapinar
 Spousal support 	work-family conflict, and the moderating		et al., 2019)
	role of instrumental spousal support.		

Table 10. Summary of empirical literature examining employee wellbeing from the perspective of the conservation of resources theory

• Employee wellbeing	The positive impact of authentic	Yes	(Salmee et
• Authentic leadership	leadership, non-financial rewards, and		al., 2020)
• Rewards	meaningful work on employee wellbeing.		
Meaningful work			
• Employee wellbeing	A multilevel analysis of employee	Yes	(Bakker,
 Job demands and 	wellbeing needs, including job demands,		2015)
crafting	job resources, job crafting, employee		
• Job resources	engagement, and work performance.		
• Employee engagement			
• Performance			
Psychological capital	Influence of workplace incivility,	Yes	(Chang et
 Perceived service 	psychological capital, and perceived		al., 2019)
climate	service climate on employee's emotional		
• Tenure	exhaustion (wellbeing).		
• Perceived health status			
 Emotional stability 			
 Social support 			
• Employee wellbeing	The impact of workplace incivility on	No	(Akhtar et
 Emotional stability 	employee wellbeing with the mediating		al., 2017)
• Tenure	effect of emotional exhaustion.		
 Social support 			
Organisational	The relationship between organisational	No	(Salanova
resources	resources and work engagement on		et al., 2005)
• Employee engagement	performance (wellbeing) and customer		
Performance	loyalty.		
• Customer loyalty			
• Corporate	The role of corporate entrepreneurship on	No	(Hoque et
entrepreneurship	performance and the mediating role of		al., 2017)
• Employee engagement	employee engagement (wellbeing).		
• Performance			
• Leader-member	A comparison of the role of managerial	No	(Xerri et
exchange	support and emotional contagion on		al., 2022)
 Psychological capital 	psychological capital and wellbeing		
	within the healthcare sector.		

2.4. Theoretical blending as theoretical framework

The three underpinning theories that drive this study have been presented: social exchange theory, emotional contagion theory and conservation of resources theory. The nuances of human relations impose a particular complexity when analysing the impact of colleagues and organisational resources on employees' wellbeing. As such, this project draws on the intersection between three different theories to set the theoretical framework. This process of using several theories is understood as "theoretical blending" (Cornelissen & Durand, 2012), which helps understand and study a problem through different theoretical lenses which give us a broader perspective on complex topics such as this one. It has been identified that there has been little research on the antecedents of employee wellbeing, and even less if considering those antecedents that could be accessed by the organisation to promote and support employee wellbeing. This subsection explores the intersection or the theoretical blending of these three theories (see Figure 2) to further understand the theoretical background that frames this project and that has assisted in developing the research questions and research model.

This project aims to understand the influence of a team on its employees' wellbeing and explore its organisational antecedents. Therefore, this investigation examines the value of workplace interactions and how these interactions impact the level of employee wellbeing. Social exchange theory explains most workplace behaviours (Cropanzano & Mitchell, 2005) and describes relationships as result-oriented social behaviours (Blau, 1964). It emphasises the importance of studying the relationships that occur within the workplace to identify how to support and promote employee wellbeing. However, several studies mention that social exchange theory has important limitations such as its ability to explain almost every workplace behaviour (Babič et al., 2019; Cropanzano et al., 2016; Meira et al., 2021). Some suggest that emotional contagion theory explains social interaction between team members with a deeper focus on how co-workers tend to synchronise their emotions and feelings (Stelman, 2016; Tee, 2015). This is strongly related to the fact that employees who belong to the same team are more alike than those from other teams (Kozlowski & Klein, 2000). All this suggests that in order to analyse how the employees' wellbeing is influenced, it is necessary to analyse clusters of teams instead of employees individually.

Social exchange theory establishes that the two main types of relationships to study are those between the employee and their supervisor and among co-workers (Mitchell et al., 2012; Vine, 2004). Employees choose to enter and maintain relationships when they receive something in exchange (Emerson, 1976). In a workplace, employees maintain relationships with co-workers or a supervisor if they feel supported or if their wellbeing increases. Social exchange theory suggests that if these relationships involve high-quality social exchange, it will result in higher levels of

trust (Cropanzano & Mitchell, 2005) as well as high levels of employee wellbeing (Milner et al., 2015). However, if it does not become a beneficial relationship, the relationship may end, leading to higher turnover and burnout rates (Danna & Griffin, 1999). Therefore, it has been suggested that better relationships with supervisors and co-workers lead to higher wellbeing. Yet, there are very few studies that analyse these relationships and take the team influence into account. The team influence is important as it is theorised to have an influence on how the employees perceive their workplace relationships.

Studying the value of these workplace interactions is essential in answering the research questions guiding this study. Employees tend to synchronise their emotions with those around them (physically or virtually). Relationships amongst co-workers can influence employee wellbeing (Benitez et al., 2019). Some studies have concluded that employees can develop mutually shared emotions and moods when working together in a team (Xerri et al., 2022). This has been studied in emotional contagion theory in concert with social exchange theory. Moreover, emotional contagion theory does not only argue that team members tend to synchronise emotions and moods with each other but also suggests that by converging emotionally with co-workers, employees can experience higher or lower levels of wellbeing depending on the co-workers' emotions.

According to the conservation of resources theory, employees will contribute their best efforts towards building, retaining, and protecting their resources in order to reduce their work stressors and support their wellbeing. Conservation of resources theory points out several resources to be considered when analysing the influences on employee wellbeing. The literature also points out that the two most prominent team resources that can influence employee wellbeing are a safety climate (Farr-Wharton et al., 2022a) and the perception of supervisorial support (Xerri et al., 2022). Moreover, these two team resources are strongly related to the two types of relationships that have been identified as essential when analysing the antecedents of employee wellbeing.

Conservation of resources theory helps leverage emotional contagion theory. Both theories are beneficial to be examined in tandem and guided by social exchange theory. An important consideration is that according to emotional contagion theory, the perception of how the team, supervisor, and organisation prioritise their safety and wellbeing is influenced by the employee's relationship with their supervisor (Blau, 1964). In most cases, relationships with greater trust tend to be considered a valuable resource or provide a higher benefit to the parties, while those marked by distrust or fear tend to have a lower value and may end sooner. Studies suggest that a positive relationship with other co-workers can influence the relationship with the supervisor and how managerial practices are perceived by employees (Bartel & Saavedra, 2000; Benitez et al., 2019). Social exchange theory proposes that a positive relationship with a direct supervisor will

contribute positively to the wellbeing of the employee (Graen & Uhl-Bien, 1995). Furthermore, the conservation of resources theory identifies positive and constructive supervisorial support as a team resource that contributes positively to employee wellbeing. The literature suggests that this is usually studied with the analysis of the perception of supervisorial support from the employee perspective using the Leader-member exchange concept (Graen & Uhl-Bien, 1995; Xerri et al., 2022)

Through the social exchange theory lens, when an employee perceives there to be a positive safety climate within their organisation and ideal levels of supervisorial support, their wellbeing will be higher (Bond et al., 2010; Dollard & Bakker, 2010). In contrast, if an employee does not feel that the organisational policies and practices support their psychological and physical health or does not perceive support from their direct supervisor, their wellbeing could be negatively impacted. In other words, the intersection between the three theories propose that employees tend to respond with higher performance and extra-role behaviours when they perceive that the organisation is investing and believing in them (Cropanzano & Mitchell, 2005).

2.5. Employee wellbeing

2.5.1. From happiness to wellbeing

Happiness has been studied since the time of renowned Ancient Greek philosophers such as Democritus (460 BCE - 370 BCE), known as the "laughing philosopher" because of his emphasis on the value of cheerfulness (Annas, 2002; Berryman, 2016). Epicurus (341 BCE - 270 BCE) had a hedonistic view of happiness and defined it negatively as the absence of pain and fear (Bergsma et al., 2008). More profoundly, Plato's (428-7 BCE - 348-7 BCE) *Republic* highlights the relationship between happiness and morality, or "the idea that one's own happiness is dependent upon acting for the sake of one's community" (Emond, 2006, p. 1). Aristotle (384 BCE - 322 BCE) introduced the concept of eudaimonia (translated as happiness or, more accurately, human flourishing (Anscombe, 1958; Fowers, 2016; Robinson, 1989), wherein happiness is achieved by pursuing that which gives life meaning and cultivating unique human endowments of virtue (Vittersø, 2016). Aristotle defined virtue as the mean between deficit and excess (Brink, 2000). In his work, Aristotle stated that the goal of any human is to reach a state of eudaimonia (Anscombe, 1958; Fowers, 2016; Robinson, 1989).

Many other philosophers have also linked the concepts of happiness, morality, and virtue. Examples include well-known Ancient Greek philosophers such as Antisthenes (Prince, 2006), Stoics such as Zeno of Citium (Annas & Wang, 1989; Engstrom & Whiting, 1993), philosophers from the Middle Ages such as St. Thomas Aquinas (Elders, 2019), and philosophers from the Modern Age such as Jeremy Bentham (Burns, 2005). The study of happiness has also attracted

attention from relatively contemporary academics like Robert Nozick (Rachlis, 1978). However, more recently, the study of happiness has been increasingly replaced with the study of wellbeing as some researchers have suggested that wellbeing is a broader concept that includes happiness (Diener, 1984; Dodge et al., 2012). Some of the most critical studies on wellbeing include those of Diener et al. (1999), Wright and Cropanzano (2000), and Dodge et al. (2012). Before exploring the definition of wellbeing, the definition and components of happiness should also be outlined.

A person can make himself happy, or miserable, regardless of what is actually happening "outside", just by changing the contents of consciousness. We all know individuals who can transform hopeless situations into challenges to be overcome, just through the force of their personalities. This ability to persevere despite obstacles and setbacks is the quality people most admire in others, and justly so; it is probably the most important trait not only for succeeding in life, but for enjoying it as well. To develop this trait, one must find ways to order consciousness so as to be in control of feelings and thoughts. It is best not to expect shortcuts will do the trick. (Csikszentmihalyi, 1991, p. 24)

Many have studied the pursuit of happiness and how can it be sustained. Csikszentmihalyi (1991) states that happiness is not strongly correlated with people's income after basic needs have been met. After that point, to increase happiness, the content of consciousness has to be changed by putting oneself in a stage of optimal experience called *flow*. Csikszentmihalyi (1991) understands happiness as engagement, and flow is a state where concentration is so intense that one cannot pay attention to anything else, and one's sense of time becomes distorted. Flow is the balance of challenge and skills (see Figure 3). If a challenge is too high for one's abilities, anxiety sets in, and the person feels stressed and alert. On the other hand, if an activity is not challenging enough and the person's skills are too great, they become relaxed and bored, which can lead to depression (Csikszentmihalyi, 1991). Middle points could lead to arousal when the challenge level is high, but the skill level is not as advanced as it should be. Or boredom if the challenge level is not high enough for the individual's set of skills.

Figure 3. Csikszentmihalyi's model of flow



Note. Mental state depending on the difficulty of the faced challenged and the skill level. Adapted from *Flow: The psychology of optimal experience* (1st Ed.), by M. Csikszentmihalyi, 1991, Harper & Row. Copyright 2000 by Global Learning Communities.

Another study has suggested that a person's happiness level has three major contributors: genetic set-range, life circumstances, and intentional or voluntary activities (Lyubomirsky et al., 2005) (see Figure 4). Genetic set-range is a mix of biological, psychological, and societal influences that characterise a person's thoughts and actions throughout their lives. Indeed, there is consensus that genetic set-range determines around 50% of a person's happiness (Braungart et al., 1992; Diener et al., 1999; Lyubomirsky et al., 2005; Tellegen et al., 1988), although there are several studies that propose that this could be as high as 80% (Lykken & Tellegen, 1996). Another 10% is determined by a person's life circumstances such as their economic status, age, physical attractiveness, the place they live, or other life events like getting married or having children (Argyle, 1999; Diener et al., 1999; Lyubomirsky et al., 2005).

The remaining 40% is determined by one's intentional activities (Gupta & Singh, 2017; Lyubomirsky et al., 2005), or actions whereby a person invests effort into building the architecture for sustainable happiness. Many scholars have analysed which activities can boost happiness. Lyubomirsky et al. (2005) have classified these activities into three different types: behavioural, cognitive, and volitional. Behavioural activities include physical exercise or healthy eating (Keltner & Bonanno, 1997; Magen & Aharoni, 1991). Cognitive activities include gratitude journals, writing about life goals, or actively reasoning (Emmons & McCullough, 2003; King, 2001; Seligman, 1998). Last but not least, volitional activities can include self-concordance and goal attainment (Sheldon & Houser-Marko, 2001) or working on meaningful tasks (Ferrari et al., 2007).

Figure 4. Pie chart of the sustainable happiness model



Note. Pie chart that represents the sustainable happiness model, where 10% of someone's happiness is due to personal circumstances, 50% due to their genetic set point, and the other 40% due to their intentional activities. Adapted from "Pursuing happiness: The architecture of sustainable change", by S. Lyubomirsky, K. M. Sheldon, & D. Schkade, 2005, *Review of General Psychology*, 9(2), p. 116. (https://doi.org/10.1037/1089-2680.9.2.111). Copyright 2005 by The American Psychological Association Division 1 (Society for General Psychology).

Happiness is typically understood as a state of mind which is purely philosophical or a life that is desirable and goes well, including psychologically (Argyle, 2001). Over the last two decades, the majority of articles have referred to the second case using the term wellbeing (Diener, 1984; Dodge et al., 2012).

What is wellbeing? Wellbeing is a nebulous term that is difficult to define. Wellbeing includes three dimensions: physical, social, and psychological (Grant et al., 2007). Crucially, it is necessary to have "a complete state of physical, mental, and social wellbeing and not merely negative as the absence of disease or infirmity" (Grad, 2002, p. 981). All three dimensions are equally important and needed to have a good level of wellbeing. According to Maslow (1998), to achieve strong wellbeing, these resources are pursued hierarchically, beginning with physical resources, then social resources, and, finally psychological resources. Physical resources include shelter, food, water, and clothing and, secondarily, healthcare and mobility. Social resources include "participating in the community, being accepted in public, and helping others" (Grant et al., 2007, p. 5). Finally, psychological resources include self-esteem, self-respect, recognition, freedom, self-actualisation, and a sense of belonging (Maslow, 1998).

These three wellbeing dimensions are included to varying extents in the two approaches that many scholars have used to define wellbeing, namely hedonism and eudaimonia (Peterson et al., 2005; Seligman, 2002). The hedonic approach defines wellbeing as life satisfaction – or how people feel and show their emotions and feelings (Anand, 2016) – and focuses on happiness, pleasure, 36

positive affect, and low negative affect (Bradburn, 1969; Diener, 1984; Kahneman et al., 1999; Lyubomirsky & Lepper, 1999). Happiness has been extensively defined in the previous paragraphs and is just one component of wellbeing. Positive affect refers to positive emotions such as cheerfulness, contentment, energy, and joy, while negative affect refers to negative emotions such as anger, anxiety, sadness, depression, or fear. The other wellbeing approach is the eudaimonic perspective, and it emphasises the fulfilment of finding meaning and positive psychological functioning through, for example, achieving personal career goals (Rogers, 1961; Ryff, 1996; Waterman, 1993). As mentioned, the eudaimonic approach has been studied since Aristotle, and its version of wellbeing is also known as the good, ultimate, and engaged life (Bauer et al., 2008; Deci & Ryan, 2008; Kashdan et al., 2008).

More recent studies have suggested that wellbeing is a multi-dimensional construct that includes both the hedonic and the eudaimonic perspectives (Brunetto et al., 2022a; Kim & Kim, 2021; Xerri et al., 2022). There are both theoretical and practical reasons to approach wellbeing as a multi-dimensional construct (Farr-Wharton et al., 2022b). Theoretically, wellbeing is comprised of various aspects, including physical, psychological, and social facets (Grant et al., 2007; Seligman, 2012). On the practical side, statistics have demonstrated that unidimensional measures are strongly affected by a respondent's mood at the time of the study (Kern et al., 2015). Some scholars have argued that wellbeing is a stable condition (Dodge et al., 2012) or that it focuses on authentic happiness (Bognar, 2010), but more recent studies have suggested that it is a multidimensional, dynamic concept that goes beyond what has been defined as wellbeing thus far (Brunetto et al., 2021).

2.5.2. From wellbeing to flourishing

Over the past two decades, there has been increasing recognition of the value of positive emotions (Seligman et al., 2005). It has been posited that positive emotions are as authentic as negative emotions and that they enhance wellbeing (Aspinwall & Brunhart, 1996; Danner et al., 2001; Harker & Keltner, 2001). Additionally, one study found that employees with more positive emotions had better performance evaluations, showed greater endurance, and chose higher goals (Barry et al., 1994). Emotional contagion theorists have pointed out that positive emotions can be as contagious as negative ones (Barsade, 2002; Xerri et al., 2022). Similarly, according to cognitive theory, people can create a more desirable present and future by choosing positive emotional responses to past and present experiences (Sharot & Sunstein, 2020). Positive emotional responses include gratitude (Diener et al., 2020; Sturm et al., 2020) and forgiveness (Keltner & Cowen, 2021).

The broaden and build theory of positive emotions suggests that "certain discrete positive emotions, although phenomenologically distinct, all share the ability to broaden people's momentary thought-action repertoires and build their enduring personal resources, ranging from physical and intellectual resources to social and psychological resources" (Frederickson, 2001, p. 219). Studying the non-linear dynamics model of team performance, Keyes (2002) and Fredrickson and Losada (2005) developed the psychological concept of *flourishing*, defined as "path-breaking work that measures mental health in positive terms rather than by the absence of mental illness" (Frederickson & Losada, 2005, p. 678).

Seligman (2012) also incorporated the concept of flourishing into his theory of wellbeing. He proposes that wellbeing is a dynamic construct that goes beyond happiness, which, when considered on its own, hides the real and complex nature of human flourishing. Seligman (2012) identifies a move in the literature from a concern with happiness to a concern with wellbeing. This is due to an increase of interest in increasing flourishing. Seligman (2012) introduced the PERMA model of flourishing, which has five domains: Positive emotions (P), Engagement (E), positive Relationships (R), Meaning (M), and Accomplishment (A). Positive emotions refer to the hedonic feeling of happiness, such as feeling joyful or content, and can only be assessed subjectively. Engagement refers to the eudaimonic concept associated with finding meaning, a positive psychological connection with the performed activities, and the presence of a flow state. Relationships include the hedonic aspect of positive affect and low negative affect and refer to friends, family, intimacy, or social connections. Meaning is the feeling of contributing to something greater than oneself. Accomplishment refers to having a sense of achievement and making progress towards goals, as per the eudaimonic component of wellbeing, even when this progress involves no positive emotions, meaning, or relationships.

Frederickson (1998; 2001; 2003; 2005) has extensively studied positive emotions. Examples of positive emotions include amusement, compassion, forgiveness, gratitude, hope, interest, joy, or love. Conservation of resources theory states that resources are everything that is valued by the individual, including objects, conditions, social support, energy, and so on (Halbesleben et al., 2014). According to Frederickson (1998; 2001; 2003; 2005), the benefits of positive emotions include intellectual, physical, social, and psychological resources. Intellectual resources help in the acquisition of new information and problem-solving skills. Physical resources help to develop coordination, strength, and cardiovascular health. Social resources solidify and create bonds. Finally, psychological resources help to develop resilience, optimism, a sense of identity, and goal orientation. As mentioned, Csikszentmihalyi (1991) studied the meaning of engagement and flow experience (also known as "being in the zone"). Flow is a mental state of full engagement

in which the skill level of the employee is balanced with how challenging the task is, as seen in Figure 3.

Positive relationships are also an essential aspect of PERMA. In the workplace, this includes interactions with other employees, co-workers, managers, and supervisors as well as with clients. A key to positive relationships is active constructive responses that include enthusiastic support, eye contact, and authenticity (Gable et al., 2004). The active constructive response is the opposite of a passive destructive response, which turns the focus inward, from the respondent to the interlocutor, and tends to avoid and ignore the speaker.

Seligman also stressed the need for finding meaning to achieve flourishing. The concept of finding meaning was introduced as a consequence of suffering in 1959 by Viktore Frankl, who initially tested his theory as an inmate in a Nazi death camp during World War II (Frankl & Lasch, 2006). However, as positive organisational scholars have been paying special attention to positive emotions, they have moved from the analysis of suffering to finding meaning in the positive to achieve flourishing. In organisational studies, several researchers have demonstrated how meaning can be created in the workplace, and three different orientations have been suggested (Wrzesniewski et al., 1997): 1) job orientation, which focuses on material rewards; 2) career orientation, where employees find meaning in their position and tasks and are motivated by success; and 3) calling orientation, which happens when people find their work inherently fulfilling.

2.5.3. Employee wellbeing and work stress

General individual wellbeing is difficult to define and measure, and employee wellbeing is similarly complex (Dodge et al., 2012; Juniper, 2011; Kiran, 2021). Employee wellbeing has attracted attention in organisational research (e.g., Alotaibi, 2022; Brunetto et al., 2022a; Farr-Wharton et al., 2022b; Roche et al., 2022), but very little has been done from an individual perspective, and almost no research has involved multilevel analysis (Tortia et al., 2022; Wang et al., 2022). The lack of agreement on the definition and measurement of wellbeing complicates its utility and blurs perspectives on its antecedents (Adamou et al., 2020; Chen et al., 2021). Without an agreed-upon theoretical framework, it is challenging to create a body of knowledge because there are no defined principles to question (Cook & Campbell, 1976).

According to the Australian Fair Work Ombudsman, an employee is "a person that's hired to provide a service to a company either on a full-time, part-time or casual basis in exchange for payment" (2022, para. 1). In Australia, full-time employees are engaged to work 38 hours per week, with some exceptions depending on contracts. A part-time employee works regular hours that are fewer than 38 hours per week. A casual worker works by the hour or day when needed

and is paid on a casual basis. Volunteers are not generally considered employees as they are not paid.

According to the conservation of resources theory, employees would try anything to minimise their stress levels and to protect their wellbeing, as a good level of wellbeing is the final goal of an employee (Halbesleben et al., 2014). To protect their levels of employee wellbeing, workers are likely to be willing to invest additional resources to maintain or reduce their stress level and support their wellbeing (Loh et al., 2018). Employee wellbeing focuses on workers' physical, social, and psychological wellness at work. The physical aspect includes employees' health and the performance of physical activities and exercise to reduce the risk of developing diseases such as obesity, type 2 diabetes, or cardiovascular disease (de Cieri et al., 2019; Loh et al., 2019). The social aspect is characterised by the quality of workplace relationships and professional affect. Finally, the psychological aspect includes job engagement and psychological functioning (Brunetto et al., 2011).

Some scholars have suggested that workplace wellbeing only includes the psychological component (Edgar et al., 2015; Vanhala & Tuomi, 2006; Wright & Cropanzano, 2000) and that it is characterised by psychological safety, as measured by the Team Psychological Safety Scale developed by Edmondson (1999). However, this definition neglects other elements of wellbeing as employee wellbeing does not only focus on the safety or the psychological component. It also includes physical and social components (Brunetto et al., 2011; Diener, 2009; Grant et al., 2007).

Employee wellbeing includes both hedonia and eudaimonia.

As with general wellbeing, psychological, physical and social dimensions of wellbeing are included in both the hedonic and eudaimonic approaches to wellbeing (Peterson et al., 2005; Seligman, 2002). Multiple studies define employee wellbeing as a multi-dimensional construct, combining both hedonia and eudaimonia (e.g., Brunetto et al., 2022a; Huppert & So, 2013; Xerri et al., 2022). The hedonic approach defines employee wellbeing in terms of job satisfaction, happiness, pleasure, positive affect, and low negative affect with co-workers, managers, supervisors, and clients (e.g., Baselmans & Bartels, 2018; Berezan et al., 2020; Diener, 1984; Kahneman et al., 1999; Lyubomirsky & Lepper, 1999). The eudaimonic approach emphasises the fulfilment of finding meaning in one's job, engagement, achievement, and positive psychological functioning through the achievement of personal career goals (e.g., Oliver, 2022; Ryff, 2019; Ryff et al., 2021; Waterman, 1993).

Several studies have analysed the cost of employee wellbeing and how it can be contagious in workgroups. Importantly, low wellbeing can negatively affect work teams (Baptiste, 2008).

Potential causes of poor employee wellbeing include personal, physical, or emotional suffering (Holman, 2002); stress and job-related anxiety (de Jonge & Schaufeli, 1998); and organisational issues such as downsizing, harassment, poor culture, or politics (Farr-Wharton et al., 2022b; Hall et al., 2016; Xerri et al., 2019). According to the Australian Institute of Health and Welfare (2022), during 2019-2020, the Australian government spent \$11 billion on mental health-related services. This represents a 3% increase since the Australian expense on mental health in 2015-2016. This poses a considerable cost for organisations as mental illness is associated with poor health behaviour and the loss of other important resources such as work-life balance (Doran & Kinchin, 2017). In 2021-2022, the cost of absenteeism for the Australian economy was more than \$44 billion, while the cost of presenteeism was around \$35 billion (Asare et al., 2022). Several authors have argued that a reduction in workplace bullying and the use of sick leave along with an increase in business profitability are the most important consequences of good employee wellbeing (Edgar et al., 2015). Other scholars have also noted overall societal benefits of strong employee wellbeing (Johnson et al., 2020).

Nonetheless, there is a call for academic studies that focus on cost-effective interventions to support organisations in their employee wellbeing programs (Murphy et al., 2023). Accordingly, the aim of the present study is to develop a deeper understanding of the extent to which employee wellbeing is influenced by work teams from a multilevel perspective. This project assumes that employees are not isolated from other workers and supervisors and that these interactions influence employee wellbeing. Employee wellbeing is affected by personal traits and circumstances as well as by organisational resources, policies, and practices and the workplace relationships between co-workers, managers, and clients (Xerri et al., 2022). As this study focuses on the team antecedents of employee wellbeing, the relationship between employees and clients or other stakeholders is not considered because such relationships are not a team component that can be managed by the organisation.

Employee wellbeing is influenced by the levels of work stress, which is influenced by the employees' perceptions of managerial priority and support (Farr-Wharton et al., 2022b). Moreover, employees with low levels of work stress, tend to experience higher levels of employee wellbeing (Gulzar et al., 2020). In contrast, employees with high levels of stress, are likely to experience lower levels of employee wellbeing. As such, this study hypothesises that:

• Hypothesis 2. Work stress is negatively associated with employee wellbeing.

This research study hypothesises that there is a negative relationship between work stress and employee wellbeing at both levels of study: the team- and the employee-level. The employeelevel relationship is the one impacted by Hypothesis 1, which was presented in section 2.2.

2.5.4. Wellbeing measures

Before exploring the proposed antecedents of employee wellbeing, it is necessary to obtain a deeper understanding of how employee wellbeing is quantified. This section presents a review of current wellbeing measures that compares, and contrasts approaches to identify an effective measurement for employee wellbeing. There are numerous psychometric and non-psychometric scales that have been proposed and applied over the past 50 years. The objective of this subsection is to gain a deeper understanding of available measures and to determine if any of these are compatible with this project's definition of employee wellbeing.

Table 11 lists 27 wellbeing measures classified according to their use, applicability in the workplace, and origin (i.e., academic or professional). Their use in the workplace is essential as employee wellbeing differs from general wellbeing. Additionally, a review of all the scales (including their definitions, scale items, and statistics from the initial study and from subsequent studies using the scales) is included in the Appendix. In some cases, this list also includes correlations found between the studies.

The number of wellbeing measures indicates the level of interest in applied research on the topic (Kun et al., 2016). However, the lack of cohesion obscures the utility and applicability of these studies for the workplace. Most of these measures are subjective self-reporting indicators that ask questions about individuals' feelings, experiences, and judgements of their lives (National Economic Foundation, 2014). Self-reporting is the most commonly used method for measuring pleasure, happiness, and wellbeing, and it is therefore an excellent way to study employee wellbeing. However, some of these instruments, such as the Subjective Happiness Scale (Lyubomirsky & Lepper, 1999), only reflect the hedonic approach to wellbeing, while others only capture the eudaimonic perspective, such as the Questionnaire for Eudaimonic Wellbeing (Waterman et al., 2010). Very few measures encompass both approaches as a multi-dimensional construct. This study focuses on analysing these measures as they comply with the proposed definition of employee wellbeing.

Measure	Reference	Origin	Items	Application to work
Affect Balance Scale	(Bradburn, 1969)	Academic	9	Yes
Authentic Happiness Inventory	(Peterson et al., 2005)	Academic	18	Yes
Emotional Exhaustion Subscale	(Maslach & Jackson, 1981)	Academic	9	Yes

7	ahle	11	Measures	of we	llheino
1	ubie	11.	weusures	UJ WEI	ubeing

Employee Psychological	(Brunetto et al.,	Aaadamia	4	Var
Wellbeing	2011)	Academic	4	1 05
Fordyce Happiness Measure	(Fordyce, 1988)	Academic	2	No
Friedman Well-being Scale	(Friedman, 1992)	Academic	20	Yes
High Performing Workplace	(South Australia	Covernment	51	Vac
Index	Government, 2015)	Government	54	105
Index Sustainable Economic	(Cobb & Daly 1989)	Academic	NA	Ves
Wellness	(Cooo & Dary, 1909)	readenne	1 1 2 2	105
Occupational Hardiness	(Moreno-Jiménez et	Academia	15	Vac
Questionnaire	al., 2014)	Academic	15	105
Oxford Happiness Inventory	(Argyle et al., 1989)	Academic	29	No
PERMA	(Seligman, 2012)	Academic	23	Yes
PFRMA_Profiler	(Butler & Kern,	Academic	23	Ves
	2016)	readenne	23	105
PANAS	(Watson et al., 1988)	Academic	20	Yes
Psychological Well-being Scale	(Diener, 2009)	Academic	NA	No
Psychological Well-being	(Ryff & Keyes,	Acadamia	20	No
Scales	1995)	Academic	20	INO
Psychosocial Safety Climate 12	(Hall et al., 2010)	Academic	12	Yes
Psychosocial Safety Climate 26	(Dollard & Kang,	Academic	26	Vac
1 Sychosocial Safety Chinate 20	2007)	Academic	20	1 05
Questionnaire for Eudaimonic	(Waterman et al.,	Aaadamia	21	Vac
Wellbeing	2010)	Academic	21	105
Satisfaction with Life Scale	(Diener, 1984)	Academic	5	No
Subjective Hanniness Scale	(Lyubomirsky &	Academic	4	Ves
Subjective mappiness Searc	Lepper, 1999)	readenne	-	105
Tension Index	(Lyons, 1971)	Academic	9	Yes
Warwick-Edinburgh Mental	(Clarke et al. 2011)	Academic	14	No
Wellbeing		Academic	17	INU
Well-being@Work Index TM	(Deloitte, 2017)	Professional	NA	Yes
Well-being360TM	(Vitality Works,	Professional	116	Ves
wen-being500	2019)	TIOICSSIOIIAI	110	105
Work Well-being	(Parker & Hyett,	Academia	50	Vas
Questionnaire	2011)	Acadelliic	50	1 55
Work-life Balance	(Brough et al., 2014)	Academic	4	Yes

Work-Related Well-being	(Orsila et al., 2011)	Academic	56	Yes
World Health Organisation- Five Well-Being Index	(Topp et al., 2015)	Professional	5	Yes

The review was performed by identifying several key indicators as criteria. The first criterion was that they are peer-reviewed in an academic journal. This left us with 24 measures. The second criterion was that they have been tested and applied on employee wellbeing and not just on general wellbeing as the definition is not the same. This left us with 18 scales. Finally, a deeper analysis was performed to analyse which definition they used to measure employee wellbeing. The results of the analysis against this last criterion was that most of these measures only included one of the two factors (hedonic and eudaimonic) that are included in the definition according to this thesis. This left us with two scales: employee wellbeing by Brunetto et al. (2011) and PERMA-Profiler by Butler & Kern (2016). However, an important factor when trying to maximise survey responses is the length of the survey, and the second scale has 23 items. For this reason, the results of the review revealed that the only wellbeing measure that fits all the criteria of this project is the employee wellbeing scale developed by Brunetto et al. (2011). The scale includes both eudaimonic and hedonic components and has been applied to public and private organisations in Australia, New Zealand, Canada, and the UK.

The employee wellbeing measure includes the following four items and is answered on a 6-point Likert scale ranging from strongly disagree (1) to strongly agree (6).

- Overall, I am reasonably happy with my work life.
- Most days I feel a sense of accomplishment in what I do at work
- I feel content with my work
- I get a sense of joy from my work

Brunetto et al.'s (2011) original scale was specifically suited to nursing, but a 2019 redevelopment (Farr-Wharton et al., 2019) aimed to make it suit any employee in any industry. The re-development of the scale consisted of the substitution of two of the original items ("Overall, I fulfil an important purpose in the work that I do" and "I get enough time to reflect on what I do in the workplace") by two new items ("I feel content with my work" and "I get a sense of joy from my work").

2.6. Antecedents of employee wellbeing

According to conservation of resources theory, two of the most important resources that act as antecedents of employee wellbeing are safety climate and managerial support (e.g., Dollard &

Bakker, 2010; Kalshoven & Boon, 2012; Selenko et al., 2013; Farr-Wharton et al., 2022a). Yet, there is little research on how these resources influence work stress and employee wellbeing from a multilevel perspective. To operationalise the aim of this thesis, this study focuses specifically on the role of social resources on employee wellbeing. Previous research has identified specific variables that impact individual wellbeing (Brunetto et al., 2020; Dollard et al., 2012; Tuan, 2016). Thus, it stands to reason that these variables are also instrumental antecedents to employee wellbeing. The first variable is employee perception of the value that team management grants to employees' psychological health and safety, and the second variable is employee assessment of the relationship between the employee and their supervisor.

This section presents the two team antecedents that are considered in this research project. Psychosocial safety climate (PSC) is an organisational resource characterised by prioritising employee psychological health (Becher & Dollard, 2016; Dollard & Bakker, 2010; Dollard et al., 2012), while Leader-member exchange (LMX) is a social exchange resource that is defined as the two-way relationship between leaders and followers (Rita Yi Man et al., 2019).

2.6.1. Managerial priority: Psychosocial safety climate

Employees' perceptions of how an organisation or team's management prioritises their wellbeing influences employee wellbeing. According to Becher and Dollard (2016), psychosocial safety climate is defined as an organisational resource that prioritises employee psychological health. Psychosocial safety climate can also be defined as "the shared belief held by workers that their psychological safety and wellbeing is protected and supported by senior management" (Bond et al., 2010, p. 49). In other words, it refers to the employees' perception that their organisation and team policies, practices, and procedures protect their psychological health and safety (Dollard & Bakker, 2010).

Hall et al. (2010, p. 355) defined the four domains of psychosocial safety climate as:

(1) senior management support and commitment for stress prevention through involvement and commitment; (2) managerial priority to psychological health and safety versus productivity goals; (3) organizational communication, that is, the organization listens to contributions from employees; and (4) organizational participation and involvement, for example, participation and consultation occurs with unions and occupational health and safety representatives. (Hall et al., 2010, p. 355)

However, several studies have claimed that psychosocial safety climate is a team or departmental resource rather than an organisational resource given its focus on management, and, in most cases, each team has a different manager and potentially a different culture (Idris et al., 2012).

Not surprisingly, positive psychosocial safety climate has been correlated with other health outcomes such as lower depression rates, lower psychological distress, and higher morale (Becher & Dollard, 2016; Bond et al., 2010; Dollard et al., 2012). Several studies have analysed how organisations and specifically teams can enhance employee wellbeing by creating a psychologically safe environment (Dollard & Bakker, 2010; Law et al., 2011). According to Hall et al. (2010), if psychosocial safety climate is effectively measured, it can highlight potential team interventions that can positively impact employee wellbeing. Therefore, this study hypothesises that psychosocial safety climate is positively associated with employee wellbeing (see Figure 5 for the full hypothesised model):

• Hypothesis 3. Psychosocial safety climate is positively associated with employee wellbeing

As with conservation of resources theory, earlier studies of psychosocial safety climate focused on employee stress and how psychosocial safety climate is negatively correlated with stress. However, positive organisational scholars have also positively related psychosocial safety climate to employee wellbeing. In theory, a better perception of how the organisation prioritises employees' psychological health has a positive impact on employee wellbeing. At the individual level, psychosocial safety climate is understood as the individual perception of management practices and policies that protect and support employees' psychological health. However, the literature also argues that psychosocial safety climate is a within unit of agreement among team members, termed the psychosocial climate strength (González-Romá & Hernández, 2014), and that it impacts both work stress and employee wellbeing (Kalshoven & Boon, 2012). At both levels, it is theorised that psychosocial safety climate has a positive influence on employee wellbeing and a negative influence on work stress. These points lead to the following hypotheses:

• Hypothesis 4. Psychosocial safety climate is negatively associated with work stress

Perceptions of managerial support are impacted by perceptions of managerial support (Becher & Dollard, 2016). According to Zhou and Jian (2015), safety climate impacts the relationship between leader-member exchange and safety behaviour. High-quality leader-member exchange relationships have been identified as a positive organisational factor that can contribute to work engagement and psychological empowerment, which can lead to a psychologically safe work environment (Oktavio, 2020). Similarly, this thesis predicts that psychosocial safety climate is positively associated with leader-member exchange:

• *Hypothesis 5. Psychosocial safety climate is positively associated with Leader-member exchange.*

These hypotheses aim to partially answer the second research question which asks to what extent organisational resources impact the level of employee wellbeing. This research study hypothesises that there is a positive relationship between psychosocial safety climate and employee wellbeing at both levels of study: the team- and the employee-level. At the employee-level, this relationship means that employees who perceive that their organisation prioritises their wellbeing are likely to experience higher levels of employee wellbeing. In contrast, employees who believe that their organisation prioritises outputs above their wellbeing are likely to experience lower levels of wellbeing. At the team-level, this relationship can be described as teams who perceive that their organisation prioritises their wellbeing are more likely to experience higher levels of collective wellbeing and vice versa.

This study also hypothesises that there is a negative association between psychosocial safety climate and work stress at both levels of study. In summary, if accepted, this hypothesis would mean that employees with higher levels of psychosocial safety climate also experience lower level of work stress. Finally, psychosocial safety climate is hypothesised to have a positive association with Leader-member exchange at both levels of analysis. Such an association would be described as those employees (and teams at the between level) who perceive that their organisation is prioritising their wellbeing are more likely to experience higher levels of managerial support. In contrast, employees (and teams at the between level) who perceive that their organisation is not prioritising their wellbeing above performance would also perceive lower levels of managerial support.

2.6.2. Managerial support: Leader-member exchange

The Supervisor-employee relationship also influences employee wellbeing. Dansereau et al. (1973) first analysed the Supervisor-employee relationship through the concept of vertical dyad linkage, which then became Leader-member exchange (Graen et al., 1982). Leader-member exchange is a form of social exchange describing the two-way relationship between a supervisor and employee from the employee's perspective (Rita Yi Man et al., 2019). According to Dulebohn et al. (2012), Leader-member exchange is affected by both the leader's and the follower's characteristics and their interpersonal relationship. It is theorised that Leader-member exchange has a strong impact on employee wellbeing (Brunetto et al., 2016).

Leader-member exchange was initially developed as part of roles theory, wherein employees accomplish their tasks through roles. However, more recently, studies have noted that employees accomplish their work within teams or departments (Babič et al., 2019; Hu & Liden, 2013; Kim & Yi, 2019). Dienesch and Liden (1986) argued that Leader-member exchange is a multilevel variable that functions at both the employee and the team levels. This aligns with multilevel

organisational theory, which underlines the importance of studying workplace behaviours and team influence because employees are not separate units but instead are part of a team. Theoretically, the Supervisor-employee relationship has a positive influence on employee wellbeing. Leader-member exchange is a useful construct in analysing this relationship and is used to answer the research questions and build the research model. At the team level, the perception of managerial priority and support is generally understood as the team climate and is strongly linked to team culture (Loh et al., 2019; Spell & Arnold, 2007). These points lead to the following hypothesis:

• Hypothesis 6. Leader-member exchange is positively associated with employee wellbeing

Employees who have poor quality relationships with their leaders often experience high levels of stress (Oktavio, 2020). Therefore, this thesis hypothesises that leader-member exchange is negatively associated with work stress:

• Hypothesis 7. Leader-member exchange is negatively associated with work stress.

These hypotheses aim to partially answer the second research question which asks to what extent organisational resources impact the level of employee wellbeing. This research study hypothesises that there is a positive relationship between Leader-member exchange and employee wellbeing at both levels of study: the team- and the employee-level. At the employee-level, this relationship means that employees who perceive higher levels of managerial support are likely to experience higher levels of employee wellbeing. In contrast, employees who perceive lower levels of managerial support are likely to experience lower levels of wellbeing. At the team-level, this relationship can be described as teams who perceive higher levels of managerial support are more likely to experience higher levels of collective wellbeing and vice versa.

This study also hypothesises that there is a negative association between Leader-member exchange and work stress at both levels of study. If accepted, this hypothesis would mean that employees (and teams) who perceive higher levels of managerial support also experience lower level of work stress. Moreover, this model has a double or chain mediation in the relationship between psychosocial safety climate and employee wellbeing. The first mediation is from psychosocial safety climate to work stress by Leader-member exchange. This is explained by the fact that managers are considered agents of their organisations in their role of providing organisational resources to their employees (Shi & Gordon, 2020). The second mediation happens from Leader-member exchange to employee wellbeing via work stress.

The supervisor-employee relationship is important not only for its influence on employee wellbeing but also for team cohesion. Various studies have found that the quality of this relationship affects a team's access to information, support, and other resources (Brunetto et al, 2011; Gerstner & Day, 1997; Mueller & Lee, 2002). Leader-member exchange also impacts turnover intentions and job performance (Dulebohn et al., 2012). It is agreed that high-quality Leader-member exchange can promote other valued resources, such as participation in decision-making (Birdi et al., 2008) and access to extra support (Brunetto et al., 2014). On the other hand, low-quality Leader-member exchange tends to be related to low performance and role ambiguity, whereby employees do not fully know what is expected from them (Brunetto et al., 2011). Additionally, employees with supportive supervisors are more likely to have positive work-oriented behaviours (Graen & Uhl-Bien, 1995), which can transfer from the individual to the collective and influence team wellbeing.

According to emotional contagion theory, employee perceptions of the Supervisor-employee relationship can also be influenced by the perceptions of other co-workers (Tee, 2015) because employees tend to synchronise their feelings and emotions. Likewise, in multilevel organisation theory, employees from the same team tend to share more similarities than other employees (Kozlowski & Klein, 2000). Hence, emotional contagion not only frames the theoretical background of this project but also acts as a construct that moderates the influence of both psychosocial safety climate and Leader-member exchange on employee wellbeing. The following section builds on this idea and develops the concept of emotional contagion as a moderator in the research model.

Some argue that managers serve as organisational agents and mediate the relationship between organisational resources and work stress (Shi & Gordon, 2020). These points lead to the following hypothesis:

• *Hypothesis 8. Negative mediation between psychosocial safety climate and work stress through Leader-member exchange*

Work stress has also been proven as a bridge between organisational resources and employee wellbeing (Oktavio, 2020). First, a balanced workload leads to higher perceptions of managerial support which lead to controlled and manageable levels of work stress which finally lead to higher levels of employee wellbeing. Therefore, this thesis hypothesises that work stress positively mediates the relationship between leader-member exchange and employee wellbeing.

• *Hypothesis 9. Positively mediation between Leader-member exchange and employee wellbeing through work stress*

Both hypothesis 8 and hypothesis 9 build on the total mediation between managerial priority and employee wellbeing through managerial support and work stress. Therefore, the final hypothesis under study is:

• *Hypothesis 10. Positive mediation between managerial priority and employee wellbeing through Leader-member exchange and work stress*

The model also reflected a double mediation in the relationship between psychosocial safety climate and employee wellbeing. First, the relationship between psychosocial safety climate and work stress is fully mediated by Leader-member exchange at the within level. This means that employees' perceptions of how their organisation prioritises their wellbeing is fully mediated through employees' perceptions of managerial support. At the between level, this is only partially mediated since the direct relationship between psychosocial safety climate and work stress is significant. Second, the relationship between Leader-member exchange and employee wellbeing is fully mediated by work stress at both levels of analysis (between and within). This means that employees' perceptions of managerial support are fully mediated through their self-reported work stress level.

2.7. Multilevel organisational theory. The need to study work teams as a cluster

This chapter has explored the theoretical and conceptual frameworks guiding this research project. However, thus far the framework does not completely address the need to study this using a multilevel perspective. Employees are not isolated units, in most cases, employees are part of a work team, and their relationships and interactions with other employees also have an influence on their wellbeing (Kozlowski & Klein, 2000). The literature suggests complementing the theoretical framework with multilevel organisational theory. This theory was first published in 2000 by Klein and Kozlowski and has been increasingly used to understand workplace behaviours in addition to already developed theories. Therefore, it is necessary to study multilevel organisational theory in detail. The following section explores multilevel organisational theory and helps to address the current knowledge gap and the methodology of this study.

In most cases, employees are part of a work team and do not work alone. Most studies mention the importance of considering the team influence on workplace behaviours, yet few of them incorporate advanced statistical analysis in their method in such a way as to account for the team influence. Most organisational and workplace research is done from the individual standpoint, analysing individuals separately without considering that employees belong to a work team or group and those groups belong to an organisation. The problem with not analysing employee wellbeing from a multilevel perspective is that the results may not reflect the reality and therefore, the recommendations derived from the results may not achieve the expected aim. There is a very big gap from individual to multilevel analysis, and yet, humans do not operate as isolated units.

Thus far, the general theoretical framework has been presented – identifying the team antecedents that are under study in this thesis. Social exchange theory has pointed out the need to consider that social interactions are the drivers of workplace behaviours. Additionally, the conservation of resources theory has established how employees value their workplace resources. Finally, emotional contagion theory has argued for the importance of considering the influence that social interactions may have on the wellbeing of employees. Therefore, this section builds on this last point to describe multilevel organisational theory, which also frames the chosen methodology, under a post-positivist research philosophy as defined by Kozlowski and Klein (2000).

Multilevel theory in organisational studies was first published on in 2000 by Klein and Kozlowski to attempt to move towards a more integrated methodology in organisational science that would be able to explain both micro and macro perspectives. Organisations have been framed as multilevel systems in previous studies, but this has not materialised beyond such identification (Dansereau et al., 1999; House et al., 1995; Klein et al., 1994; Seron et al., 1980). According to Kozlowski and Klein (2000), organisational studies have never fully integrated different levels of investigation. In most studies the three different levels: individual employees, work teams or departments, and the overall organisation, are studied separately. This division has resulted in organisational researchers not integrating the various organisational levels, which has therefore created different approaches, theories, and methods (Kozlowski & Klein, 2000). However, organisations are multilevel systems, and they should be studied as such.

The perspective of organisations being divided into levels was first mentioned by Lewin (1951) and was later reviewed by James and Jones (1976). James and Jones (1976) analysed the characteristics of the organisational context and stated that work climate perceptions are the result of both situational and individual influences. They found that the aggregation of each individual perception, as well as the context, results in the overall organisational climate. Similarly, individual employees' wellbeing can be aggregated at the team level to result in the overall team wellbeing, or be measured as collective wellbeing if there are appropriate measures for this.

Multilevel organisational theory has five principles that are organised around the "what, how, where, when, and why (and why not)", and they have been summarised in Table 12. The *what* principle attempts to define the focus of the project. According to this principle, the focus should be on the endogenous or dependent variable. In this case, employee wellbeing is the endogenous variable that the theory and research attempt to understand and explain. A psychosocial safety climate and Leader-member exchange are the exogenous variables that are studied as the team

antecedents of employee wellbeing. Within this principle, Kozlowski and Klein (2000) add that even though multilevel models are relevant to the vast majority of phenomena, there are cases where they are not necessary. In summary, multilevel analysis is unnecessary when the dependent variable is not influenced by higher-level units, does not influence lower-level units, or when it has been infrequently explored in the organisational literature. However, this project does not fall within these cases, and employee wellbeing should be studied at a multilevel perspective. Few studies have analysed how employee wellbeing, the dependent variable, is influenced by teams or how a team can influence its employees' wellbeing (e.g., Bayhan Karapinar et al., 2019; Brunetto et al., 2014; 2016; Farr-Wharton et al., 2021; Salmee et al., 2020).

Principle	Description	Implications for this research
What	Theory building should begin	Employee wellbeing is the endogenous
	with the designation and	construct of interest. It is the "what" or the
	definition of the theoretical	phenomenon that the theory attempts to
	phenomenon and the	understand.
	endogenous construct(s) of	The exogenous variables in the research model
	interest.	under study are the psychosocial safety climate
		and Leader-member exchange.
How	A multilevel model must	In this case, there are both top-down and
	specify how phenomena at	bottom-up links between a work team and its
	different levels are linked.	members regarding wellbeing. Top-down
	These links can be top-down or	processes include employees embedded within
	bottom-up.	teams, and bottom-up processes are
33.71	· · · · · · · · · · · · · · · · · · ·	represented by collective wellbeing.
Where	The unit specification should	Work teams will be at the top level, as these
	be driven by the theory of the	are the groups that interconnect. The bottom
	phenomena in question.	level comprises employees, as they are who
		will treat their wellbeing as a resource and
		attempt to conserve or acquire it per the
		conservation of resources theory. It is this
		closer relationship between work team
		contagion
When	The temporal scope as well as	The temporal scope will be determined by the
w nen	the point in the life cycle of a	data collection process which took place
	social entity affect the apparent	during 2021
	origin and direction of many	during 2021.
	phenomena in such a way that	
	they may appear variously top-	
	down, bottom-up, or both.	
Why and	Multilevel theoretical models	This literature review of the theoretical
why not	must provide a detailed	framework justifies the need for a multilevel
•	explanation of the assumptions	analysis, as employee wellbeing and the
	undergirding the model. These	antecedents under investigation are influenced
	should answer "why" and "why	at both the individual and collective level.
	not".	

Table 12. Principles of multilevel organisational theory

(Kozlowski & Klein, 2000)

The following principle establishes that every multilevel model must specify *how* the dependent variable is linked with each level analysed. There are top-down processes, also known as contextual influences, which in this study involve the employees embedded within teams. Bottom-up processes, also called *emergence*, happen when there is a collective measure. There are two types of collective measures: first, the aggregation of individual aspects, and second, the collection of a collective measure itself. In this case, the aggregate of employee wellbeing could form the team wellbeing or could be measured as collective wellbeing. Additionally, it is important to know *where* these processes originate and terminate, and this should be driven by theory. For this project, work teams will be at the top level, as these are the groups that interconnect and involve relationships between members (Sales-Prado et al., 2007; Simon & Laird, 2019). It is this closer relationship between work team members that is necessary for emotional contagion (Hatfield et al., 1993; Stelman, 2016). The bottom level is defined as each employee, as employees treat their wellbeing as a resource and attempt to conserve or acquire it per the conservation of resources theory (Halbesleben et al., 2014; Hobfoll, 2001).



Figure 5. Hypothesised model

Note: Hypothesised multilevel research model with the between and within effects as well as the observed variables.

The *when* principle confirms that time is essential to incorporate into any model, although it has been neglected in many studies (House et al., 1995). According to Kozlowski and Klein (2000, p. 23), "the temporal scope, as well as the point in the life cycle of a social entity, affect the apparent origin and direction of many phenomena in such a way that they may appear variously top-down, bottom-up, or both". The final principle is the *why and why not*. This principle states that

multilevel models must justify why they are multilevel as well as provide an explanation on why the relationship between levels is bottom-up (or top-down) and not top-down (or bottom-up).

It is equally important to take into consideration several principles that Kozlowski and Klein (2000) have structured for model specification. Initially, it is important to define, justify, and explain the level of each construct that builds the model. In this project, the definition and justification for each construct is developed in the following section, the conceptual framework. A construct is a mental abstraction used to express aspects such as ideas, events, organisational characteristics, and people attributes (Austin & Vancouver, 1996). For this study, the primary construct is employee wellbeing (further details are described in the coming sections).

The level of measurement must also be considered. In this study, as employee wellbeing is an individual-level construct, data is collected from individuals (or employees). Another important factor to take into account is the type of multilevel model that will be used. According to Kozlowski and Klein (2000), there are two types of multilevel models: single level and cross level. Single-level models study the relationship between the dependent and independent constructs at a single level, while cross-level models study the relationships between constructs at different levels. Finally, the sampling process and analysis process must be considered. These are further described within the methodology section of this document.

2.8. Professional services industry in Australia

This last section of Chapter 2 presents a description of the professional services industry in Australia as the context in which this thesis examines the team influence on employee wellbeing. While this thesis did not aim at any industry or sector in particular, there was still a need to operationalise the study. Chapter 3 contains an in-depth presentation of the recruitment strategy that was undertaken to get participants. Due to the complexity of multilevel data and the need to identify the work team to which each participant belongs as well as at least two group members of each team, the engagement strategy targeted private sector organisations within the professional services industry in Australia. Moreover, the data collection process took place during the Australian lockdowns caused by the pandemic of COVID-19, and employees within the professional service industry were still able to work from home and operate in remote teams.

The professional services industry is the third largest industry in Australia after health care and retail (ABS, 2022). Professional services include occupations that require special training, specific qualifications and/or licences (Nishikawa & Orsato, 2021). This industry includes professional services such as accountants, engineers, marketing specialists, and business consultants amongst others. Within the professional services industry, there are teams that operate within the same area (e.g., accountants) with a manager and tend to work closely together as part

of their daily duties. This context is particularly important since the employees within the professional services industry do not have to engage in emotional labour on a frequent basis. As this chapter has explored, most of the research papers on employee wellbeing or emotional contagion study public sector employees and those who engage in emotional labour (e.g., Brunetto et al., 2021; 2022a; Farr-Wharton et al., 2021; 2022; Xerri et al., 2019; 2022). Moreover, these employees are also impacted by red tape and public sector bureaucracy (Baig et al., 2021), as well as austerity driven policies and scarcity of organisational resources (Farr-Wharton et al., 2021). Instead, the participants in this research project belong to private sector organisations within the professional services industry in Australia.

2.9. Conclusion of the chapter

This section has analysed the four theories that frame the research project. The aim of this research project is to develop a deeper understanding of the team antecedents of employee wellbeing and to analyse how these antecedents influence employee wellbeing. This theoretical background has helped to develop the research questions under study. The intersection of these four theories is the foundation of the theoretical framework for this project and is where the current knowledge gap lies. The theoretical blending of the first three theories: social exchange theory, conservation of resources theory, and emotional contagion, forms the theoretical underpinning of this project. However, they present an additional limitation, as they all consider that employees cannot explain the effect of the team and the individual. Therefore, a fourth theory, multilevel organisational theory, has also been included. The theoretical framework was followed by the conceptual framework and an in-depth examination of wellbeing and its antecedents. The chapter concluded with the presentation of the industry in which the data collection took place to contextualise the findings of this project.

Chapter 3. Methodology

The core objective of academic research is to provide answers to theoretical questions and develop an aspect for a particular body of knowledge that had not been done before (Shaheen, 2021). With this aim, academic research is guided by a set of established principles and theoretical assumptions. The context for this thesis was presented in Chapters 1 and 2, and the theoretical and conceptual framework were outlined in Chapter 2. This research project aims to contribute new, empirically derived knowledge regarding the impact of work teams and managerial support on employee wellbeing. Given this objective and in accordance with multilevel organisational theory, the study was developed using a multilevel statistical analysis approach, within a postpositivist research philosophy, as defined by Kozlowski and Klein (2000). The methodology chapter defines the research approach and the research methods developed as part of the thesis. This chapter commences with a presentation of the various approaches to employee wellbeing research. An overview of the different research paradigms as well as an outline of the philosophy guiding this research project follows, along with a description of the quantitative methods adopted. The subsequent section details the data analysis techniques employed to test the theorised model and its hypotheses in order to answer the research questions. Chapter 3 concludes with an overview of the ethical considerations and potential risks of the study taken into account by the researcher. While this is an extensive chapter developing the methodology of this study, it has been necessary to use this space to develop the complexity of the multilevel statistical analysis.

3.1. Approaches to employee wellbeing research

Employee wellbeing has been extensively explored and researched, as elaborated in Chapter 2. Despite ample research, maintaining good levels of employee wellbeing remains a key challenge for many organisations (Lawless & Willocks, 2021; Xerri et al., 2022). Employee wellbeing has been explored using both quantitative and qualitative research methods. While from a qualitative perspective, employee wellbeing has been explored using both single and multilevel analysis, it has not been the same for quantitative research. Most quantitative studies on the antecedents or consequences of employee wellbeing have focused on the employee level and have not considered multilevel influence. As a consequence, the potential variance caused by higher level influencers such as the team are not being considered. As we know, most of the time, employees work in a team. It is known that members of a team tend to be more alike than those from different teams. According to emotional contagion theory, those who work together tend to converge emotionally. Therefore, the observations (or responses) are not independent of each other. Nonetheless, there are still very few multilevel statistical studies that examine the antecedents of employee wellbeing. The lack of multilevel studies is mostly due to the to the fact that statistical software

packages that allow for multilevel statistical analysis did not exist until recently. The first version of Mplus to include multilevel modelling was version 2.1, which was launched in February 2001 (Muthén & Muthén, 2022). Moreover, multilevel data collection processes are much more demanding, and there tends to be a scarcity of resources (funds and time) to support researchers (Irani, 2019). Additionally, there is a high level of complexity in the development of the syntax associated with a multilevel statistical analysis. For all these reasons, quantitative post-positivist research has focused on single-level analysis when studying factors contributing to employee wellbeing as well as its outcomes and consequences. Table 13 presents some of the studies that have analysed the antecedents of employee wellbeing using a single-level structural equation modelling analysis from January to October 2022.

Table 13. Research studies that have analysed the antecedents of employee wellbeing using single-level structural equation modelling analysis between January and October 2022

Title	Reference
Flexibility and Saudi employees' perceptions of job satisfaction: A	(Alotaibi, 2022)
multisector study.	
The role of social responsibility and ethics in employees'	(Bocean et al., 2022)
wellbeing.	
Psychosocial safety climate, psychological capital, healthcare	(Brunetto et al., 2022b)
SLB's wellbeing and innovative behaviour during the COVID 19	
pandemic.	
Managing emotional labour: the importance of organisational	(Brunetto et al., 2022a)
support for managing police officers in England and Italy.	
An integrative review on job burnout among teachers in China:	(Cheng et al., 2022)
Implications for human resource management.	
Transformative leadership and organizational citizenship behavior	(Deusdedit & Alazeezi,
in UAE.	2022)
Transformational leadership and project success: The mediating	(Fareed et al., 2022)
role of trust and job satisfaction.	
Work harassment in the UK and US nursing context.	(Farr-Wharton et al.,
	2022b)
The impact of working from home during COVID-19 on time	(Gunasekara et al., 2022)
allocation across competing demands.	
The impact of human resources environment and organizational	(Hameed et al., 2022)
identification on employees' psychological well-being.	

The influence of technostress, work-family conflict, and perceived	(Harunavamwe & Ward,
organisational support on workplace flourishing amidst COVID-	2022)
19.	
Do workers respond differently to learning from supervisors and	(Huo & Boxal, 2022)
colleagues? A study of job resources, learning sources and	
employee wellbeing in China.	
Lean production, work intensification and employee wellbeing:	(Huo et al., 2022)
Can line-manager support make a difference?	
Significant task activates trait gratitude for organizational	(Li et al., 2022)
citizenship behaviors: The mediating role of psychological	
availability.	
Self-determination theory and accountant employees'	(Lin et al., 2022)
psychological wellbeing: The roles of positive affectivity and	
psychological safety.	
Family support and posttraumatic growth among tourism workers	(Luu, 2022)
during the COVID-19 shutdown: The role of positive stress	
mindset.	
Organizational justice and employee well-being in India: Through	(Majumdar & Kumar,
a psychological lens.	2022)
Mediating role of meaningful work between sustainable human	(Nasir & Waheed, 2022)
resource management and employee engagement: A study on	
banking sector of Pakistan.	
Mindfulness older workers and relational leadership.	(Roche et al., 2022)
The influence of pandemic-related workplace safety practices on	(Subramony et al., 2022)
frontline service employee wellbeing outcomes.	
Antecedent and employee well-being outcomes of perceived	(Xiao et al., 2022)
benefits schemes: a two-wave study.	

As shown, there are numerous papers on the antecedents of employee wellbeing that have used structural equation modelling at the single level within a specific timeframe. Although their findings contribute to understanding how organisations can further support employees' mental and physical health, many of these articles state the need for a multilevel analysis that also considers team influence (e.g., Brunetto et al., 2022a; Fareed et al., 2022; Li et al., 2022). Employees are not isolated units unaffected by other factors such as co-workers or managers, yet there has been very little research published on collegial and managerial influence on employee wellbeing using a multilevel quantitative approach. Statistical packages that support multilevel

structural equation modelling have been available for several years and still, there is a lack of multilevel studies. Thus, probably due to the complexity in both data collection and analysis, very few studies have incorporated such advanced statistical processes in their methodology and findings. Table 14 presents studies that have analysed employee wellbeing using a multilevel statistical methodology in ten months of 2022. There is a large difference in the number of single-level studies and multilevel projects, with there only being two studies in the latter category. Not surprisingly, many researchers have emphasised the urgent need for multilevel research to understand the organisational antecedents of employee wellbeing.

Table 14. Research studies that have analysed the antecedents of employee wellbeing using multilevel structural equation modelling analysis between January and October 2022

Title	Reference
A human growth perspective on sustainable HRM practices,	(Tortia et al., 2022)
worker well-being and organizational performance.	
Servant leadership and employee wellbeing: A cross-cultural	(Wang et al., 2022)
investigation of the moderated path model in Canada, Pakistan,	
China, the US, and Brazil.	

To address the research questions and test the hypotheses of this project, a multilevel quantitative analysis with a post-positivist research philosophy was deemed as being necessary. The present study incorporated a multilevel research analysis to study managerial and collegiality influence on employee wellbeing. This research used the available advanced statistical packages as well as documentation on multilevel analysis to establish the research design and the data analysis process. As such, the findings obtained can provide human resources professionals and line managers with valuable and novel insights for informing and designing programs and interventions that support employees' wellbeing.

3.2. Research paradigm

A research paradigm (also called *philosophical paradigm*) can be defined as "universally recognized scientific achievements that, for a time, provide model problems and solutions to a community of practitioners" (Kuhn, 1970, p. viii). Philosophical paradigms are different ways of understanding and viewing the world from a philosophical perspective and possess a distinctive standpoint regarding what constitutes acceptable knowledge, the nature of reality and the role of values in the research project (Rehman & Alharthi, 2016). An understanding of the different philosophical paradigms is important in determining and justifying the research design and methodology. Normally, the philosophical worldviews of the researcher are linked with the selected strategies of inquiry and the chosen research methods. These three different aspects (philosophical worldviews, selected strategies of inquiry, and research methods) inform the

research design of a project as seen in Figure 6. The philosophical assumptions that best align with this research project and its research questions are the positivist and post-positivist paradigms as defined by Kozlowski and Klein (2000).

Figure 6. What informs a research design?



Note. The interconnection between philosophical worldviews, research methods and the selected strategies of inquiry. Adapted from *Research design: qualitative, quantitative, and mixed methods approach* (5th ed.), by J. W. Creswell and J. D. Creswell, 2018, SAGE Publications, p. 5. Copyright 2018 by SAGE Publications, Inc.

The spectrum of research paradigms and philosophical worldviews have significantly evolved over time. The two poles of the spectrum are positivism and interpretivism. Due to their disunited understanding of the world and what can be considered true, supporters of each side have historically questioned the methods and validity of the other side (Denzin, 2012). However, since this is a spectrum, several other research paradigms exist depending on the researcher's understanding of the world. The major paradigms are positivism, post-positivism, transformative approach, pragmatism, and constructivism or interpretivism. However, other philosophical paradigms may also exist, and new ones could arise over time. Research paradigms not only differ in methodology but also in epistemology, ontology, and axiology (Maarouf, 2019). Ontology, epistemology, and axiology are three different brands of philosophy that address the unspoken assumptions that researchers make about their surroundings (Prosek & Gibson, 2021). These assumptions support researchers in determining their methodological choices and analysing their results. Table 15 presents a summary of these five major research paradigms according to their logic, ontology, epistemology, axiology, and methodology.
	Positivism	Post-	Transformative	Pragmatism	Interpretivism
		Positivism			
Logic	Deductive	Mainly	Mainly	Abductive	Inductive
	(verify)	deductive	inductive	(converse)	(interpret/
		(predict)	(transform/		understand)
			emancipate)		
Ontology	One reality	One or	Realities are	Reality is	Subjectivism,
What is Real?	that is	multiple	socially	constant	socially
	external,	realities that	constructed	debated and	constructed
	objective,	are external,	entities that are	interpreted as	reality with
	and	objective, and	under constant	new	multiple
	independent	independent to	internal	situations	possibilities
	to the	the researcher,	influence	arise	
	researcher	but they			
	and can be	cannot be fully			
	found	found			
Epistemology	Findings are	Findings are	Knowledge is	Knowledge is	Knowledge is
What is True?	true and	probably true	socially	searched to	socially
	represent the	but could have	constructed	solve	constructed
	reality	limitations		problems	
Axiology	Research is val	lue-free.	Intrinsically	Values play a	Research is
What do we	Researcher is i	ndependent of	valuable aiming	large role in	value-bound.
Value?	the data and m	aintains an	to socially	interpreting	Researcher
	objective stanc	e.	emancipate	results;	cannot be
				researcher	separated from
				adopts both	the research,
				objective and	which is
				subjective	therefore
				points of	subjective
				view	
Methodology	Quantitative	Quantitative	Quantitative/	Quantitative/	Qualitative
How can we			Qualitative	Qualitative	
examine what					
is Real and					
True?					

Table 15. Summary of the five major research paradigms according to their logic, ontology, epistemology, axiology, and methodology

Ontology can be defined as the "nature of reality" or what it is understood as reality (Collis & Hussey, 2013, p. 59). Positivists argue that there is only one absolute truth and that it can be

discovered and measured by research (Alvesson & Skoldberg, 2009). Positivism is also known as "the scientific method" (Scotland, 2012). On the other side, interpretivists suggest that there is not an absolute truth but rather everyone has their own reality constructed via mutual understanding (Grant, 2022). For interpretivists, there is more than one reality as each person has their own experience of reality. Post-positivists argue that, even though there is one reality, there may be limits to human's ability to accurately capture it (Kankam, 2019; Ryan, 2006). The transformative approach suggests that there is no such thing as a true reality as the reality is a product of power relations (Creswell & Creswell, 2018). Lastly, pragmatic research agrees with the positivist and post-positivist idea that there is a reality but argues that it cannot be determined (Nunfam, 2021). Multilevel organisational theory follows a post-positivist research philosophy as it agrees that there is one reality but maintains that there are limitations to discovering it.

Epistemology denotes the "nature of knowledge", or the study of knowledge acquisition and how humans understand their environment and surroundings (Wilson, 2014, p. 9). For positivists, the object of study is distinct from the researcher (Scotland, 2012). Positivism is frequently used in mathematics or biology. For post-positivists, the object of research is also distinct from the researcher, and the research builds an approximation of the object of study (Creswell & Creswell, 2018); this approach is normally used in psychology and medicine. Constructivists or interpretivists, conversely, believe that knowledge is co-constructed, and that people can only learn by engaging and sharing (Alharahsheh & Pius, 2020). Traditionally, this paradigm has been used in psychoanalysis and education. Interpretivists take a subjective perspective regarding the object of study. Traditionally, this paradigm has been used in psychoanalysis and education. The transformative approach argues that knowledge is political and value-laden. Some have suggested that this is the approach of gender research, politics, and some sociological studies (Creswell & Creswell, 2018). From a pragmatic perspective, knowledge is both subjective and objective depending on the phase of the research (Teddlie & Tashakkori, 2009). Within this research project, the researcher is independent from the participants, which aligns with the positivist and post-positivist paradigms.

Axiology refers to "the role of values and ethics" within the research project (Saunders et al., 2019, p. 134) and the origin of those values. Positivists focus on logical, disengaged, and absolute knowledge, whereas post-positivists agree that research is dispassionate, universal, and motivated by particular reasons (Creswell & Creswell, 2018). On the other end, constructivists argue that knowledge is contextual and there are biases caused by the researcher, who cannot be separated from the object under study (Alharahsheh & Pius, 2020). For those who see the world through the transformative lens, their values sit within critical analysis, social justice, and a transformative belief system (Creswell & Creswell, 2018). Finally, pragmatism focuses on solving practical

problems in the world (Feilzer, 2010; Kaushik & Walsh, 2019). Lately, pragmatism has been defined as more aligned with social justice (Morgan, 2014) and the idea that the values of the researcher are not independent from the interpretation and discussion of the results (Kaushik & Walsh, 2019). Pragmatism strongly aligns with this project as its final goal reaches beyond contributing to the understanding of employee wellbeing and intends to provide further evidence on how organisations can support employee wellbeing and promote healthier workplaces.

3.2.1. Positivism and post-positivism

Quantitative studies have traditionally been under the positivist or post-positivist research paradigm (Creswell & Creswell, 2018). Positivists and post-positivists believe that the researcher is independent from the participants under study (Alvesson & Skoldberg, 2009). While both approaches believe that there is one truth, there is an important difference between them. From a positivist approach, this one truth can be found through quantitative research methodology such as experiments or surveys. Conversely, from a post-positivist approach, it is not possible to find the one truth, but it is possible to obtain an approximation of reality using inference statistics and other quantitative methods. This research project aligns with the latter research paradigm.

Creswell and Creswell (2018) argue that the biggest difference between the pragmatic and postpositivist approach is that for pragmatists there is not only one singular reality. In fact, Metcalfe (2014) argues that pragmatism is an extension of post-positivist and interpretivist approaches. The pragmatist paradigm draws on both objective and subjective knowledge. While the researcher is still external from the research, there is not a unique reality – instead, the reality is relative to the perspective of the researcher and the interpretation of the results.

3.3. Quantitative methods

This research project studies the influence of work teams on employee wellbeing and job stress. The chosen methodology was developed using a post-positivist quantitative approach in which the researcher was independent of the research and the analysis sought objectivity. However, the findings do not necessarily verify a reality but rather take a dialectic perspective whereby the results juxtapose ideas with the objective of resolving a conflict. The research takes a multilevel statistical analysis approach and a post-positivist research philosophy, as defined by Kozlowski and Klein (2000).

A significant innovation of this study relates to the use of multilevel statistical analysis. While multilevel analysis is somewhat established in the qualitative approach, it is still nascent within organisational quantitative methods (Ward et al., 2022). Multilevel statistical analysis seems strangely undervalued and often argued against by reviewers despite the obvious flaw involved in measuring between level constructs at the within level (Jong & Ford, 2019). Multilevel

structural equation modelling seeks to understand the influence of variables at different levels of analysis on an outcome (dependent) variable. In quantitative organisational studies, multilevel analysis has only been developed over the past decade and has not been used to analyse the emotional contagion among co-workers on employee wellbeing. In application, multilevel analysis can be used to understand how, for example, organisations and/or teams can impact individual employees and vice versa. This contrasts with traditional, single-level analysis approaches, which are limited as they only consider statistical relationships at the same level of analysis (e.g., an employee's personality and corresponding work engagement).

In standard general linear models or standard structural equation models, it is assumed that the observations are independent of each other. In other words, these models assume that participants or responses are independent of each other. Therefore, the model assumes that the residuals are normal, independent, and identically distributed. However, this is rarely the case in social sciences. These assumptions could be violated if there are multiple observations per participant, as occurs with longitudinal data, but this is not the case for this research project. Another case where single level structural analysis could be applied is when participants are clustered in groups, as in this research project. Employees belonging to a work group tends to introduce similarity among group members. As such, the residuals are not independent from each other.

The aim of using multilevel structural equation modelling was to account for, rather than ignore group variance, as happens with standard single-level models. In this way, the results could separate the variance into two groups: the within-variance explaining the individual effect on each employee, and the between-variance explaining the often-ignored team effect.

3.4. Research design

This section details the research design. Multilevel analysis is used when the observed data has a hierarchical or clustered structure (Kozlowski & Klein, 2000). In this research, there are employees nested within teams. These clusters tend to be more alike in their characteristics than employees not studied within their workgroups (Kozlowski & Klein, 2000). Multilevel models recognise the existence of this hierarchy and allow some residual variance in each level (Ward et al., 2022). The residual variance can then be divided into a between-team component, which is the variance of the team-level residuals, and a within-team component, which is the variance of the employee-level residuals. The team residuals are called the *team effects*, and the employee residuals are called *employee effects*. This allows researchers to identify the influence of the team level variance, as occurs in traditional single-level analysis. One area of significance in this research project is the multilevel statistical analysis. The results provide the individual and the team level

variance, which offers valuable information for organisations that want to implement programs to support their employees' wellbeing. Equipped with deeper understanding of the extent to which teams influence employee wellbeing and work stress, organisations could be more successful when promoting specific interventions or training and development programs that support employee wellbeing.

This type of advanced statistical analysis is a contemporary approach that has only been in organisational studies and academic journals for the past decade. Multilevel theory in organisational studies was first published in 2000 by Klein and Kozlowski, who aimed to move organisational science research towards a more integrated methodology to explain both microand macro-perspectives. Organisations have been framed as multilevel systems since the early 1980s, but this multilevel approach has not materialised beyond a simple presentation (e.g., Dansereau et al., 1999; House et al., 1995; Klein et al., 1994; Seron et al., 1980). According to Kozlowski and Klein (2000), organisational studies have never been fully integrated because most projects study organisations at three separate levels: individual employees, work teams or departments, and the overall organisation. This division has prevented organisational researchers from integrating the different levels, which has created different approaches, theories, and methods (Kozlowski & Klein, 2000). The creation of different approaches, theories, and methods is particularly problematic when trying to reduce the readiness gap that exists between valuing and supporting the wellbeing of employees. Without an agreement on a definition, a measurement or even on what the antecedents and consequences of employee wellbeing are, it is not possible to create a robust body of knowledge (Cook & Campbell, 1976). However, organisations are multilevel systems, and they should be studied as such.

There are a number of reasons for using a two-level model in this project. First, the research questions concern the grouping for individual outcomes as the study aims to measure team influence on employee wellbeing. As a result, the multilevel analysis can yield the correct inferences. In traditional multiple regressions, the regression coefficients are underestimated if there is a hierarchical structure influencing the variance. This project did not analyse the third level of analysis (the organisational level) or any potential higher levels such as the industry or country levels. There are several reasons for not including higher levels. First, the sample size does not allow for higher levels of analysis. Second, once analysed, the organisational level effect is very small as the variables included are team-level and/or individual-level variables. Nevertheless, key considerations were made to ascertain the validity of the results without these higher levels. What is more, the potential for higher level research projects opens the door for further research through important findings that could impact the business world.

After discussing the methodological implications of this study, this chapter details the application of the multilevel organisational design to investigate the influence of teams on the relationship between employee wellbeing and work stress.

3.4.1. Population sampling

Classic statistical inference processes such as difference analysis, factor analysis, or regression analysis are based on the assumption that the cases used are obtained from a simple random sampling procedure (Harradine et al., 2011). However, this is not the case for most quantitative studies in the social sciences, including applied psychology or management, and this research project is no exception. The employees who participated in the study were not randomly selected. They were part of a work team, just as a group of students would be part of a class or a group of nurses form a hospital ward. The participants of this study were nested, or clustered, within a work team. Hence, the data is hierarchically structured in a two-level hierarchy – employees nested within work teams.

Generally, individuals within a cluster are more alike than those from different clusters (Kozlowski & Klein, 2000). For example, employees who work in the accounts department of an organisation tend to be accountants, bookkeepers, or have an interest in numbers. Similarly, those employees who work in a kitchen are all chefs, cooks, or kitchen assistants, and they are all interested in cooking or gastronomy. Other authors such as Humphrey and LeBreton (2019) or Sierra (2011) also explain how individuals within the same group tend to have similar characteristics than those from different groups. In these cases, the precision of inferences around single-level parameter estimates is reduced. If hierarchically structured data is treated as a simple random sample and the hierarchical structure is ignored, the standard errors around parameters measured at the higher level (in this case, the team level) will be underestimated.

For this sampling process, one of the multilevel statistical implications is that individual (in this case, employee) data must be nested within one's team or workgroup. Specifically, this meant that team members needed to identify that they belonged to a specific team. This had implications for the sample size of the study because it required that there be more than one respondent in each team; otherwise, there would have been no team-level effect as both the team and the employee would have been the same. Some researchers have argued that four is the minimum desired number of respondents to generate skewness and kurtosis statistics for each group (Blanca et al., 2013). These statistics analyse the shape of the distribution of variables. If a group has fewer than four respondents, it is challenging or even impossible to ascertain valid representations of the group's central tendency to undertake statistical calculations. Skewness measures the symmetry of the distribution and whether there is some skewness from the central tendency (see Equation

1). Kurtosis is a measure for analysing how much outliers affect the distribution by rendering the distribution leptokurtic (positive) or platykurtic (negative; see Equation 2). Others have argued that having at least two group members (Hallquist, 2017) is sufficient to measure team influence and thus find the between-team and within-team residual variances (also called team and employee effects).

Equation 1. Skewness formula

$$Skewness = \frac{\sum_{i}^{n} (Y_{i} - \bar{Y})^{3}}{(n-1)\sigma^{3}}$$

Where:

n is the number of variables in the distribution Y_i is the variable of the distribution \overline{Y} is the mean of the distribution σ is the standard deviation

Equation 2. Kurtosis formula

$$Kurtosis = n \frac{\sum_{i}^{n} (Y_{i} - \overline{Y})^{4}}{\left(\sum_{i}^{n} (Y_{i} - \overline{Y})^{2}\right)^{2}}$$

Where:

n is the number of variables in the distribution Y_i is the variable of the distribution \overline{Y} is the mean of the distribution

Equally, there must be a minimum number of groups to yield enough samples at the team level to robustly compare group properties. While 25 is the minimum number of groups that other multilevel organisational researchers have used (e.g., Bakker, 2015; Benitez et al., 2019; Chen et al., 2019; Dollard et al., 2012; Ilies et al., 2007; Kammerlander et al., 2017; Salmee et al., 2020; Terpstra-Tong et al., 2020; Wang et al., 2017; Xie et al., 2021), this low sample size can make it difficult to achieve robust goodness of fit indices for the computational models involved in multilevel analysis. Indeed, computational models benefit from having 40 or more groups and become very robust when there are more than 96 groups. This is because statistical measures of central tendency become much more reliable with larger sample sizes. However, at the team/group level, achieving large sample sizes in multilevel research is challenging and resource-intensive. For this reason, most published studies in management/organisational research domains

have between 25-55 teams. Table 16 presents the number of individuals at the within level and the number of groups at the between level that other recent organisational multilevel studies have used when undertaking research. Elorza et al. (2016) posit that the minimum number of groups necessary to run multilevel structural equation modelling is 30. This threshold was determined after noting that previous studies consistently presented valid estimation and model fit with between-level samples greater than 30. Additionally, González-Romá & Hernández (2017) review multilevel research and its applications in various fields. The authors point out the importance of determining the ideal sample of teams and individuals to precent estimate bias and other errors. González-Romá & Hernández (2017) also state that the minimum of 30 groups is ideal to obtain reasonable standard errors, although several studies have achieved it with as few as 10 groups of 5 individuals. This research project collected data from 41 groups, which fits within the previously mentioned threshold of 30 groups to be able to have a valid estimation and appropriate model fit indices.

Within-level sample size	Between-level sample size	Reference
40,000+	400+	(Kammerlander et al., 2017)
368	57	(Wang et al., 2017)
396	91	(Benitez et al., 2019)
542	36	(Farr-Wharton et al., 2021)
274	43	(Farr-Wharton et al., 2022a)
343	30	(Mohd Salleh et al., 2015)

Table 16. Number of individuals at the within level and the number of groups at the between level of recent organisational multilevel studies

The target population for this research project was individuals working as part of a team (excluding solo employees) in Australia. The unit of analysis was not a specific industry or group of organisations within a particular industry. This research project aimed to study how teams influence the levels of employee wellbeing without targeting any particular industry.

The study employed a variety of sourcing strategies to recruit companies and work teams that would be interested in the findings or that would benefit from the output of this research project. The main recruiting strategy involved contacting human resource managers and other managers directly via LinkedIn. The utilisation of social media accounts to promote research projects has increase notably during the last decade (Cawcutt et al., 2019; Kapoor et al., 2017). According to Al-Daihani et al. (2018), it has opened the doors for more and better research as it has offered more visibility and a broader impact for research projects. Other sources were used as well. The researcher contacted organisations that promote mental and physical health across Australia such as Beyond Blue, which is a non-profit organisation that provides mental health support and

services (Beyond Blue, 2022). In addition, other professional and industry associations were contacted such as the Australian Human Resources Institute, the National Institute of Accountants, the Real Estate Institute of Australia, and the Economic Society of Australia. Professional or industry associations are a cooperative group of organisations and individuals that represent several professions. By contacting these organisations, the researcher could reach a wide number of organisations and gain sponsorship. Support from these institutions is important to highlight the relevance of this project for individual private organisations. Moreover, sponsorship from professional associations increases the participation rate (Heijmans et al., 2015).

In total, over 600 Australian organisations were contacted. The researcher initially shared an information brochure and a YouTube video with potential participants. Both the recruitment flyer and the video link can be found in Appendix 2. Both presented the key objectives of the study and a call-to-action link where users could decide to engage with the study. After several rounds of emails and communications, 14 Australian organisations agreed to participate, but valid data was collected from only nine of those organisations. The others did not have sufficient participants to be able to run a multilevel statistical analysis with the available statistical software.

Once the data collection process closed, the data from 237 completed surveys from 41 work teams was stored, prepared, and analysed. The 41 teams belonged to nine companies in the professional services industry. It is important to reiterate that the unit of analysis was not a specific industry or group of organisations within an industry but rather work teams. The research examined team dynamics within Australian organisations. Teams from multiple organisations can be compared as long as one of the following two approaches is taken into consideration: 1) if the organisational effect (the third level of analysis) is proven to be non-existent or very small, the third-level analysis is insignificant; 2) and/or the organisational effect is controlled for when performing the multilevel statistical analysis. The organisational effect is the influence that each organisation has on each employee. So long as this effect is statistically insignificant or controlled when performing the multilevel analysis, it is appropriate to compare different teams even if they do not belong to the same organisation.

3.4.2. Data collection

Quantitative studies focus on the collection of numerical data to study the relationship between constructs (Creswell, 2014). The study of such relationships is particularly useful as such connections can be applied practically and have an impact beyond research. The most common means of collecting numerical data is by using a survey. Questionnaires are a robust method of research as they use uniform definitions to set the questions (Hagino, 2002). This is particularly important to ensure that all participants are asked the same questions in an identical way (Hagino,

2002). Questionnaires allow for the collection of same data from numerous respondents (Akinci & Saunders, 2015). Moreover, surveys are impersonal as participants are unable to build a relationship with the investigator (Gliner et al., 2017) – thus, aligning with the positivist and postpositivist paradigms in which the researcher is independent from the participants. Surveys can be administered in numerous modalities such as face-to-face, traditional post, telephone, or online. While online questionnaires have been used for decades, the COVID-19 pandemic enhanced the use of web technology to avoid close contact between people. Better web technology has allowed researchers to reach a wider audience and to collect data remotely, thereby reducing costs and increasing the number of responses and the speed at which data is collected (Mertler, 2020). Participants are normally also given the autonomy to complete the survey in their own time and at their own pace, which increases participation rate.

In this study, data was gathered directly from employees using a survey-based, self-report strategy (Ghauri et al., 2020). The online survey software that has been used to manage and distribute the survey is Qualtrics. The online survey software that was used to manage and distribute the survey is Qualtrics.

The survey development for this project included four parts: consent, team identification, constructs, and demographics. The consent block includes the participant information sheet, all the relevant information regarding the survey, the usage of the responses, and contacts in case participants felt distressed. Team identification was an essential part as the research examines team influence on employee wellbeing. Team identification was achieved by asking each employee to write the three first letters of their manager's first name and the three first letters of their manager's surname.

The team identification questions were followed by the scales of the constructs, including employee wellbeing, psychosocial safety climate, Leader-member exchange, work stress, and emotional contagion. The scales were generated from the extant literature and presented using matrix questions with statements rated on a Likert-type scale. These measures are further discussed in Section 3.4.4.

Finally, general demographic questions were included on gender, year of birth, and tenure. Demographic questions were placed at the end of the survey as they are usually off-topic and sometimes unnerve respondents who feel that such questions are intrusive (Green et al., 2000). They could also be placed at the beginning of the survey, as some argue that this helps to ease the participants into the survey and increase their confidence (Teclaw et al., 2012). They can also be placed at the beginning if they are used for screening (Teclaw et al., 2012). However, this study did not require particular demographics and did not need initial screening of participants.

Additionally, and according to the Human Resources Ethics Committee at the University of Technology Sydney, the survey included questions that have a potentially psychologically disturbing effect as some questions could be a trigger for psychological distress. For these reasons, only demographic questions that were essential for the study were asked, and these were placed at the end of the survey after the participants had a chance to see what the survey asked.

In addition to the survey development, a recruitment strategy was established to engage with organisations willing to participate in the study. Over 600 Australian organisations were contacted via email and LinkedIn. Of these, 14 agreed to participate in the study, but only nine had a sufficient number of complete responses to be included in the research. Data collection began after Human Research Ethics Committee approval in August 2021. Each participant organisation received a personalised survey link to distribute amongst their employees. The data collection process was finalised in January 2022. After the survey was closed, all data was securely stored to begin the data analysis process.

The most common research design when using questionnaires is a cross-sectional study, as discussed by de Vaus (2014). Cross-sectional correlational surveys are intended to study the relationship between independent and dependent variables (Cooper & Schindler, 2014; Cassell et al., 2018). These projects involve collecting data at one point in time (Cassell et al., 2018), which allows the researcher to identify if a relationship exists between the various dependent and independent variables. However, it is important to note that the researcher cannot infer whether the dependent variables are caused by the dependent variables (Cassell et al., 2018).

Fourteen Australian organisations within the professional services industry agreed to participate, and valid data was collected from only 9 of those organisations. The other five did not have sufficient participants to be able to run a multilevel statistical analysis with the available statistical software. The details of the survey and the questionnaire link were shared via email with a representative of each participant organisation or work team. In general, these representatives were human resource managers or wellbeing officers within the companies and/or teams. For safety purposes and on request from the Human Research Ethics Committee, the organisations and/or their wellbeing or human resource managers were the ones who distributed the survey amongst their organisation or work team. Several reminders were sent out to ensure that sufficient data for a multilevel statistical analysis was collected. According to van Mol (2017), survey reminders are the most effective way to engage with the targeted audience and increase sample sizes.

When the data collection process finished, a total of 237 responses from 41 different work groups were received. The data preparation process began shortly after the survey was closed.

3.4.3. Survey considerations

An online survey was used for data collection as surveys are mostly used to determine perceptions and assess attitudes (Sackett & Larson, 1990). Rating scales are commonly used in social sciences and in the majority of quantitative research projects within this field (Croasmun & Ostrom, 2011), and scales were also used in the present study.

To maximise the success of the data collection process, it is important to consider several key factors of the survey design. For this, the researcher should undertake extensive research on survey design to ensure that the participants' levels of confusion, frustration, and potential distress are minimised. This research project relied on a cross-sectional web questionnaire where participants were required to indicate their agreement level on several scales. It was also essential to collect team identification data in order to perform multilevel statistical analysis. Additionally, important demographic information such as gender, age, and tenure were also collected. A copy of the questionnaire is provided in the Appendix 1 and a copy of the participant information sheet was included in the introduction of the survey (see Appendix 3). The information sheet offered more details about the research project.

According to Balch (2010), the target population, distribution speed, and participation rate are considerations that should be considered when undertaking survey research. When they are correctly designed and distributed, surveys are a very robust data collection instrument (Quinlan et al., 2019). They collect meaningful responses for the researcher, but they do not require a lot of effort from the responder (Quinlan et al., 2019). Questionnaires also allow large datasets to be collected while minimising costs and time (Mertler, 2020) and allow the researcher to make inferences about a population without having to collect data from every individual (Fraenkel et al., 2014).

However, if overused, surveys can cause survey fatigue (Buckingham & Saunders, 2004; de Koning et al., 2021: Le et al., 2021). There are two types of survey fatigue that should be avoided at all costs. First, survey request fatigue happens when an individual receives too many surveys, causing frustration and exhaustion. Second, survey-taking fatigue occurs when individuals experience exhaustion because a survey is too long (generally, more than 20 minutes), difficult to understand, or not user-friendly. When undertaking survey research, it is essential to reduce potential survey fatigue as its consequences have a high cost and could even lead to having to discard a dataset. Some of the key consequences include lower response rate, skewed results, waste of resources such as time and money, or even damage to the researcher's institution or academia overall (de Koning et al., 2021).

Keeping the survey engaging, as well as short, is also key in collecting as many valid responses as possible. Some suggest that having different types of scales and question types keeps the audience more active (Le et al., 2021). All the central scales were presented using a matrix with the items as the rows and the Likert scale as the columns. The Likert-type scale was described and developed by Rensis Likert (1931), and it is the most common attitude score in social sciences (Croasmun & Ostrom, 2011). Likert scales provide a range of responses to a series of statements that then become aggregated following different methodologies to build a construct. Generally, there are between five and seven categories of response. Some researchers have argued that the best scale has five items, ranging from strongly agree to strongly disagree and with a neutral point in the middle (Croasmun & Ostrom, 2011). Others prefer a scale with 7 or more items as this may increase the reliability of the instrument (Simms et al., 2019). Still others have argued that the reliability and validity of an instrument is not related to the number of scale points used (Boateng et al., 2018).

The various scales and variables used for this research consist of different scale end points. For some constructs, the 5-point Likert scale was used while for others, the 7-point Likert scale was used. All the scales that were added to the survey used the original number of items as per the original developed construct. Moreover, not all scales followed the strongly disagree to strongly agree format; some ranged from rarely to most of the time, depending on the nature of the question. This also allowed respondents to remain engaged with the survey as well as allowing researchers to identify unengaged responses so that the final sample was more robust.

There is an important limitation to the use of self-report surveys: this approach can present bias as participants may answer what they believe the researcher wants to see or may "respond in a way that makes them look as good as possible" (Donaldson & Grant-Vallone, 2002, p. 247). To address this issue, Donaldson and Grant-Vallone (2002), suggest that the sensitivity of the questions as well as the characteristics and potential conflicts of participants are taken into account. All the scales utilised in this project were selected from a review of the extant literature, as outlined in section 3.4.4. All scales were used and validated in previous peer-reviewed research projects. Moreover, important ethical considerations were considered to minimise the potential distress that these questions may cause to respondents. These ethical considerations are further discussed in section 3.5.

One last methodological concern with survey collected data is the possibility of common-method variance (CMV) or common-method bias. CMV is "the amount of spurious correlation between variables that is created by using the same method – often a survey – to measure each variable" (Craighead et al. 2011, p. 578). In other words, it is the variance caused by using a survey to

measure a construct that cannot be measured in real life (e.g., wellbeing). If variables are affected by CMV, the intercorrelations amongst them would not represent reality. Some researchers have analysed the sources of CMV. Podsakoff et al. (2003) argue that there are three different causes of CMV. First, when responses are self-reported and come from the same respondent. This cause is inevitable in a research project that looks at the relationship between variables in a two-level analysis. Second, the way items are presented as well as their context can cause CMV. In this study, this has been addressed by following the available survey design tools presented in this section and accounting for potential limitations due to this being an unfunded project with time constraints. Third, contextual factors such as location and time can produce CMV.

During the last decade, several articles have been published describing different ways to test and address CMV. According to Craighead et al. (2011), two thirds of the research articles published between 2001 and 2009 did not note how CMV was taken into consideration for their research. However, most methodological papers find testing and reporting on CMV crucial to publish meaningful results (e.g., Jordan & Troth, 2020). On the other hand, some authors have suggested that CMV does not pose a great thread to validity of findings with constructs that are measured using multiple items (Fuller et al., 2016).

For this project, CMV was addressed via two different approaches. First, the Variance Inflation Factor (VIF) test was applied (Kock, 2015; Salmerón Gómez et al., 2018). This statistic represents the severity of multicollinearity using an index that measures how much the variance of an estimated regression coefficient is increased because of collinearity. This statistic is discussed in detail in Chapter 4, along with the results of the multilevel statistical analysis. Second, Chang et al. (2010), propose using different scale types to minimise CMV as this reduces the likelihood of having consistent responses across different constructs. This approach has also been used by having different scale sizes (5 and 7 Likert items) or items (strongly agree to strongly disagree and rarely to most of the time).

3.4.4. Questionnaire scales

A print version of the survey can be found in Appendix 1. The survey consisted of three different sections. The first section comprised the introduction with the consent question and information sheet. A copy of the participant information sheet is available in Appendix 3. The following section describes all the relevant scales for this research project, including employee wellbeing, psychosocial safety climate, Leader-member exchange, colleague emotion-sharing, and work stress.

As employee wellbeing, colleague emotion-sharing, work stress, managerial support, and psychosocial safety climate are not readily observable by direct means, latent measurement tools

were needed (El-Den et al., 2020; Song et al., 2013). A latent variable is a variable that cannot be directly observed or measured using standard tools, but instead, it is inferred from other observed variables (Ranalli, 2016). Though these constructs are intangible variables, a psychometric tool like a Likert scale can capture the levels of employee wellbeing, work stress, colleague emotion-sharing, Leader-member exchange, and psychosocial safety climate.

To summarise, five relevant scales to measure the five constructs included in the model were also included in the questionnaire. A summary of these can be found in Table 17, and each of these scales is described in the following subsection. The survey closed with three demographic questions: gender, year of birth, and tenure.

Concept	Reference	Items	Likert-Scale
Employee wellbeing	(Brunetto et al., 2011)	4	6-Likert scale
Work stress	(Cullen et al., 1985)	5	5-Likert scale
Emotional contagion	(Siebert et al., 2007)	6	6-Likert scale
Psychosocial safety climate	(Hall et al., 2010)	4	5-Likert scale
Leader-member exchange	(Graen & Uhl-Bien, 1995)	7	5-Likert scale

Table 17. Latent constructs included in the survey

3.4.4.1. Employee wellbeing scale

The literature review included an extensive review of the theoretical and conceptual frameworks presented in Chapter 2. In addition, Chapter 2 included a review of 27 different psychometric and non-psychometric scales that have been developed and applied over the past 50 years with the objective of measuring employee wellbeing. The scales were assessed according to their applicability in the workplace and their definition of employee wellbeing. The details of this review can be found in Chapter 2 with a review of the different definitions of employee wellbeing and how to measure it. The conclusion was that the best available measure for employee wellbeing was the one developed by Brunetto et al. (2011). This scale used the same definition as this project, only had four items, and has been tested in Australian and international organisations. Moreover, this tool has been used in several multilevel studies (Farr-Wharton et al., 2021; 2022a; Xerri et al., 2019).

Brunetto et al.'s (2011) original scale was specifically suited to nursing, but a 2019 redevelopment (Farr-Wharton et al., 2019) aimed to make it suit any employee in any industry. The re-development of the scale consisted of the substitution of two of the original items ("Overall, I fulfil an important purpose in the work that I do" and "I get enough time to reflect on what I do in the workplace") by two new items ("I feel content with my work" and "I get a sense of joy from my work").

The final four scale items used in this study are: (EWB_1) "Overall, I am reasonably happy with my work life", (EWB_2) "Most days I feel a sense of accomplishment in what I do at work", (EWB_3) "I feel content with my work", and (EWB_4) "I get a sense of joy from my work". The responses were obtained on a 6-point Likert scale ranging from strongly disagree to strongly agree. High scores represent perceptions of high levels of employee wellbeing. The reliability of this scale can be found on section 3.4.5.8.

3.4.4.2. Work stress scale

Work-related stress is a significant workplace issue, as presented in Chapter 2. It is defined as a psychological feeling of tension, strain, and anxiety caused by work tasks and duties. In other words, it is the physical, mental, and emotional reaction of employees who perceive that their work demands exceed their abilities and their available resources. It can thus negatively impact the performance and behaviour of employees.

Job stress was measured using a 5-item scale developed by Cullen et al. (1985). This scale has been validated and used to measure work stress in many different research projects. The studies that have used this measurement between January and October 2022 can be found in Table 18.

Table 18. Journal articles in English that referenced and used the job stress scale developed by Cullet et al. (1985) between January and October 2022.

Title	Reference
Understanding who is hired to work in U.S. prisons and why it matters:	(Burton et al., 2022)
A call for research.	
What it takes to be a "good" correctional officer: Occupational fitness	(Cassiano et al.,
and co-worker expectations from the perspective of correctional	2022)
officer recruits in Canada.	
Mental health outcomes among public social workers in the occupied	(Easton et al., 2022)
Palestinian territories.	
Suffering in silence: Violence exposure and post-traumatic stress	(Ellison & Jaegers,
disorder among jail correctional officers.	2022)
Just another day's work: The nexus between workplace experiences	(Ellison et al., 2022)
and Post-Traumatic Stress Disorder (PTSD) in jail settings.	
"That doesn't leave you": Psychological dirt and taint in prison	(Garrihy, 2022)
officers' occupational cultures and identities.	

"Anything can happen at any time": Perceived causes of correctional	(Goulette et al., 2022)
officer injuries.	
"Lockdown probation leaders" and fundamental human needs.	(Herzog-Evans &
	Sturgeon, 2022)
Pursuing a correctional career: The motivations and reasons for	(Higgins & Swartz,
staying.	2022)
How conflict "bleeds over" for correctional staff: Exploring work-	(Higgins et al.,
family conflict through correctional subculture.	2022b)
"We keep the nightmares in their cages": Correctional culture, identity,	(Higgins et al.,
and the warped badge of honor.	2022a)
Testing the job demands-resources model with organizational trust	(Keena et al., 2022)
among prison staff.	
Testing the job demands-resources model for organizational	(Lambert et al.,
commitment among Indian correctional officers.	2022b)
Testing the job demands-resources model in explaining life	(Lambert et al.,
satisfaction of Nigerian correctional staff.	2022a)
What matters most? Comparing the impact of individual, job, and	(Mack &
organizational factors on job stress and job satisfaction among juvenile	Rhineberger-Dunn,
justice personnel.	2022)
COVID-19's impact on black, female correctional officers and justice-	(Martin-Howard et
involved individuals at Rikers Island jail.	al., 2022)
Teachers' war against terrorism: A mediated moderation model.	(Memon et al., 2022)
The creation of a correctional officer trainee self-efficacy index: An	(Miller et al., 2022)
application of item response theory.	
Workplace safety: Perceived dangerousness versus experienced fear	(Rhineberger-Dunn
among community corrections personnel.	& Mack, 2022)
An evaluation of a yoga program designed for correctional	(Smith et al., 2022)
administrators and officers.	
"We are all humans and deserve a decent way to go": Examining	(Steely Smith et al.,
professional's experiences with providing end-of-life care in	2022)
correctional institutions.	
The development of prison officers' job satisfaction and its impact on	(Stinglhamber et al.,
depersonalization of incarcerated persons: The role of organizational	2022)
dehumanization.	

Can't shake the prison guard blues: Examining the effects of work	(Worley et al., 2022)
stress, job satisfaction, boundary violations, and the mistreatment of	
inmates on the depressive symptomatology of correctional officers.	
Determinants of job dissatisfaction and its impact on the	(Yean et al., 2022)
counterproductive work behavior of university staff.	

The five items included in this scale are (WS_1) "A lot of the time my job makes me very frustrated or angry", (WS_2) "I am usually under a lot of pressure when I am at work", (WS_3) "When I am at work I often feel tense or uptight", (WS_4) "I am usually calm and at ease when I'm at work" (reversed item), and (WS_5) "There are a lot of aspects of my job that make me upset". The responses were obtained on a 5-point Likert scale ranging from strongly disagree to strongly agree. High scores represent perceptions of high levels of work stress. The reliability of this scale can be found in section 3.4.5.8.

3.4.4.3. Emotional contagion scale

Emotional contagion was measured using a 6-items scale developed by Siebert et al. (2007). A published study has used a workplace adaptation of this scale. This adapted scale is the most commonly used tool for measuring emotional contagion in the workplace as it was specifically designed for employees (Jung & Yoon, 2019; Petitta et al., 2017). A collection of the studies that used this scale in 2022 is presented in Table 19.

Table 19. Journal article in English that referenced and used the emotional contagion scale developed by Siebert et al. (2007) between January and October 2022.

Title	Reference
Modelling maternal depression: An agent-based model to examine the	(Benny et al., 2022)
complex relationship between relative income and depression.	

The six scale items are (EC_1) "When there is excitement happening around me at work, I remain cool" (reversed item); (EC_2) "At work, if colleagues appear to be worried about something, I generally remain calm" (reversed item); (EC_3) "If colleagues around me feel depressed, I also tend to feel depressed"; (EC_4) "If a colleague felt upset, I would also feel upset"; (EC_5) "At work, I would become nervous if colleagues around me appear nervous"; and (EC_6) "The colleagues around me have a strong influence on my mood". The responses were obtained on a 6-point Likert scale ranging from strongly disagree to strongly agree. High scores represent perceptions of high levels of emotional contagion amongst co-workers. The reliability of this scale can be found in section 3.4.5.8.

3.4.4.4. Psychosocial safety climate scale

Employees' perceptions of organisational priority for their mental and physical health as well as their satisfaction with team policies and practices were measured using a 4-item scale (PSC-4), which is a component of a bigger scale (PSC-12). The PSC-4 consists of the four items that build the managerial priority section of PSC-12. This factor was selected as it represents the interactions between managers and employees and how employees perceive these interactions as authentic. PSC-4 was developed by Dollard and Bakker (2010) and has been used in several organisational studies. A summary of these studies is presented in Table 20. According to Dollard (2019), PSC-4 is the most commonly used short scale for measuring psychosocial safety climate and has the best psychometric properties of all the instruments reviewed.

Table 20. Journal articles in English that used the psychosocial safety climate scale developed by Dollard and Bakker (2010) between January and October 2022

Title	Reference
Psychosocial safety climate moderates the effect of demands of	(Alshamsi et al.,
hospital accreditation on healthcare professionals: A longitudinal	2022)
study.	
A minute in time saves minds: Psychosocial safety	(Dollard et al., 2022)
climate (PSC-4) cut-points for early detection of	
mental distress.	
Facilitating stress prevention in micro and small-sized enterprises:	(Engels et al., 2022)
Protocol for a mixed method study to evaluate the effectiveness and	
implementation process of targeted web-based interventions.	
Emergency service workers: The role of policy and management in	(Farr-Wharton et al.,
(re)shaping wellbeing for emergency service workers.	2022a)
The Swedish HealthPhys study: Study description and prevalence of	(Hagqvist et al.,
clinical burnout and major depression among physicians.	2022)
"I was given three marks and told to buy a Porsche" - Supervisors'	(Sjöblom et al., 2022)
experiences of leading psychosocial safety climate and team	
psychological safety in a remote academic setting.	

The four items included in this scale are (PSC_1) "Managers show support for stress prevention through involvement and commitment"; (PSC_2) "Management clearly considers the psychological health of employees to be of great importance"; (PSC_3) "There is good communication here about psychological safety issues which affect me"; and (PSC_4) "In my organisation, the prevention of stress involves all levels of the organisation". The responses were obtained on a 5-point Likert scale ranging from strongly disagree to strongly agree. High scores

represent high levels of perception of managerial priority for employees' physical and mental health. The reliability of this scale can be found in section 3.4.5.8.

3.4.4.5. Leader-member exchange scale

Employees' satisfaction with the quality of the quality of their Supervisor-employee relationship was measured using a 7-item scale (LMX-7) developed by 1. According to Brunetto et al. (2014), the LMX-7 scale is the most commonly used tool for measuring Leader-member exchange quality and has the best psychometric properties of all the instruments reviewed. In fact, over 500 research papers published in 2022 used this scale. A summary of 2022 Australian studies is presented in Table 21.

Table 21. Journal articles in English that used the Leader-member exchange scale developed by Graen and Uhl-Bien (1995) with Australian data between January and October 2022

Title	Reference
A contemporary view of interpersonal aggression and cyberbullying	(Akram et al., 2022)
through ICT: Multilevel insights from LMX differentiation.	
Mediating-moderating effect of employee creativity and team potency	(AlMazrouei et al.,
on expatriate innovative work behaviour.	2022)
Leader-follower interpersonal behaviors, emotional regulation and	(Ayoko et al., 2022)
LMX quality.	
Work harassment in the UK and US nursing context.	(Farr-Wharton et al.,
	2022b)
Examining the effect of entrepreneurial leadership on employees'	(Hoang et al., 2022)
innovative behavior in SME hotels: A mediated moderation model.	
The role of transformational leadership on firm performance:	(Huo et al., 2022)
Mediating effect of corporate sustainability and moderating effect of	
knowledge-sharing.	
From empathic leader to empathic leadership practice: An extension to	(Jian, 2022)
relational leadership theory.	
Leader favorable feedback and withdrawal behavior: A moderated	(Jiang & Qu, 2022)
mediation model of gratitude and Leader-member exchange.	
Psychological contract, leadership, and job satisfaction: An empirical	(Megheirkouni,
investigation into the non-profit sports sector.	2022)
Linking responsible leadership and green innovation: The role of	(Shahzad et al., 2022)
knowledge sharing and Leader-member exchange.	
Comparisons draw us close: The influence of Leader-member	(Tang et al., 2022)
exchange dyadic comparison on coworker exchange.	

Does middle leaders' learning-centred leadership matter in promoting	(Thien et al., 2022)
teacher professional learning? A partial least squares analysis.	
Leader crisis communication and salesperson resilience in face of the	(Tuan, 2022)
COVID-19: The roles of positive stress mindset, core beliefs	
challenge, and family strain.	
Catching emotions: The moderating role of emotional contagion	(Xerri et al., 2022)
between Leader-member exchange, psychological capital and	
employee well-being.	

The seven items included in this scale are (LMX_1) "Do you know where you stand with your direct supervisor, and do you usually know how satisfied your direct supervisor is with what you do?"; (LMX_2) "How well does your direct supervisor understand your job problems and needs?"; (LMX_3) "How well does your direct supervisor recognise your potential?"; (LMX_4) "Regardless of how much formal authority your direct supervisor has built into his or her position, what are the chances that your direct supervisor would use his or her power to help you solve problems in your work?"; (LMX_5) "Again, regardless of the amount of formal authority your direct supervisor has, what are the chances that he or she would 'bail you out' at his or her expense?"; (LMX_6) "I have enough confidence in my direct supervisor that I would defend and justify his or her decision if he or she were not present to do so"; and (LMX_7) "How would you characterise your working relationship with your direct supervisor?". Responses were obtained on a 5-point Likert scale ranging from strongly disagree to strongly agree. High scores represent perceptions of high levels of employee satisfaction with their relationship with their line manager. The reliability of this scale can be found in section 3.4.5.8.

3.4.5. Data analysis

In January 2022, after the data collection process finished and all surveys were closed for responses, the data analysis process begun. After the data preparation, 237 completed surveys were collected from 41 different work teams. All the data was securely stored following University of Technology Sydney Human Resource Ethics Committee advice, after which the data preparation process began. The data analysis process featured several important steps prior to testing the hypotheses of the theorised multilevel research model. The results of the data analysis are presented in Chapter 4 along with an in-depth discussion in Chapter 5. This section explores the data analysis process, including the data preparation, treatment of missing data, and analysis of outliers.

3.4.5.1. Data preparations

Data preparation is a key process within quantitative analysis for gathering, combining, organising, and structuring all data so that it can be used to test research hypotheses (Corrales et al., 2018). Data preparation includes addressing all the necessary aspects to ensure that the dataset is ready for the subsequent analysis. It includes the treatment of missing data and outliers, cleansing, structuring, validation of the data and the scales, and, if needed, transformation (Pallant, 2020).

To simplify the data wrangling process, each company received a different Qualtrics link with access to a different survey, although the content of the surveys was the same. Once all data was collected, all surveys were downloaded in a .sav file and stored safely in a secure electronic folder. The software used to access and prepare the data was the Statistical Package for the Social Sciences (SPSS) 28. The data management and storage information processes are discussed in section 3.6. After storing the raw data, a master SPSS document (.sav) was created with all the responses as well as an identifier for the organisation and another identifier for the work team that each employee (entry) belonged to.

That master file was screened to ensure that all the data was in a suitable condition for the rest of the analysis. According to DeSimone and Harms (2018), data screening can significantly improve the trustworthiness and quality of the data and the study results. Some of the key benefits of data screening include the identification of data issues that need to be addressed at an early stage. The master file had 293 responses in 41 different work teams, of which 56 responses were deleted as they were incomplete and presented missing data problems (see section 3.4.5.2).

Once all data was collected in the master file, an initial exploration of the data was undertaken. The aim of this initial data profiling was to identify patterns, relationships, and other potential attributes in the data as well as anomalies, inconsistencies, missing values, and outliers that could impact the data analysis (DeSimone & Harms, 2018).

3.4.5.2. Missing data

Missing data happens when an observation does not have an assigned value (Mirzaei et al., 2022), for example, if a particular question has no response because the respondent skipped or did not answer that question. When collecting data using a questionnaire, it is common to have missing data (Mirzaei et al., 2022), though there are options to avoid it to some extent. Missing data is particularly problematic in multilevel structural equation modelling with latent variables. At least up to August 2022, the best available software, Mplus, was not able to run the multilevel statistical analysis when there was untreated missing data (Winter & Depaoli, 2022). Therefore, it is

essential to address missing data as early as possible to avoid further problems with the data analysis.

Missing data can be classified in two different categories. On the one hand, there can be missing data when respondents leave certain items or questions of the survey unanswered (Mirzaei et al., 2022). When a participant leaves an item unanswered, it may be because they did not want to answer and skipped it on purpose or because of an involuntary error. In this study, all questions included in the survey were mandatory, and it was not possible to skip a question as Qualtrics has the option to bar participants from continuing without giving a response, thereby avoiding missing data. The second category of missing data happens when a participant leaves the survey unfinished by only completing the first questions of the survey (Mirzaei et al., 2022). This second type is particularly problematic for multilevel structural equation modelling, but it is also problematic for other types of analysis since a significant amount of missing data may cause problems with model fit or jeopardise the significance of the findings (Zhang & Savalei, 2022).

Within missing data, there are three types of missing values: missing completely at random (MCAR), missing at random (MAR), and missing not at random (MNAR) (Pham et al., 2022). MCAR refers to completed data that cannot be distinguished from missing data. In other words, the causes of missing data are not related to the dataset. Data that presents MAR problems is identified following a pattern that can be linked with other items in the same dataset. This second type is more frequent than MCAR. If the missing data cannot be identified as MCAR or MAR, then it is data that is MNAR (also known as not missing at random, or NMAR). MNAR happens when there is no pattern to the missing data.

A screen check of the SPSS master file was undertaken to identify missing data. According to Karanja et al. (2013), there are two common strategies to work with missing data. These techniques are called first generation and second generation. The first-generation technique consists of removing the entries that have missing values. Researchers who use this technique propose that hypothesis testing should be run with data that is observed and not estimated (Awan et al., 2021; Kwak & Kim, 2017). Therefore, removing those entries with missing data would be the most appropriate technique to treat missing data. The first-generation method is the most common technique to deal with missing data (Awan et al., 2021). Those researchers who disagree with the first-generation technique argue that it can significantly reduce the sample size and reduce data representativeness (Kwak & Kim, 2017). The second-generation technique consists of a longer process that aims to estimate the missing values of missing data by employing other values within the dataset (Carter, 2006). As such, there would not be more missing data since all missing values would be estimated using other values in the same dataset. This method uses

imputation analysis to replace missing values using the mean, median, mode, or even a value obtained from previous statistical models (Dubey & Rasool, 2021). This is commonly used when there are missing items within the same variable (Dubey & Rasool, 2021). However, some have argued that if the proportion of missing data is larger than 10% of the total entry, imputation analysis would not be reliable and could present high levels of bias (França et al., 2021).

This research project used the first-generation technique because within the data collected there were no missing values that could be predicted using other available data. Some of the 56 entries that were removed had completed less than 80% of the survey, making it is impossible to predict what the other 20% of the responses that the participant left blank would have been. The final master data file contained 237 responses clustered within 41 different work teams.

3.4.5.3. Outliers

A statistical outlier is a data point that significantly differs from the central tendency of a variable (Kwak & Kim, 2017), such as when all employees of an organisation earn an annual salary within a range except an employee who earns significantly more than the rest. According to Aguinis et al. (2013), when using Likert scales, those who respond on either end of the scale should not be considered outliers. However, outliers could exist in data collected using questionnaires when the respondent is unengaged with the survey (Barnett, 1994; Liu et al., 2010). Respondent disengagement could be due to lack of interest or survey fatigue, amongst other reasons (de Koning et al., 2021). In this case, participants enter random responses that then follow a pattern and, in some cases, may even contradict each other.

Another cause of outliers in survey data that asks participants to self-report behaviours is response bias, also known as survey bias (Beaumont & Rivest, 2009). Response bias occurs when participants respond with an inaccurate answer or do not respond truthfully (Giromini et al., 2022). Research shows that participants are more likely to over-report good behaviours and underreport bad or negative behaviours (Vashistha et al., 2018).

In this project, to check for potential outliers in each work group, the researcher completed a first visual evaluation using a boxplot. A boxplot is a method to graphically analyse the spread of the distribution for each work group and potential skewness. See, for example, Figure 7, which represents the boxplot of the first item of the employee wellbeing scale (EWB_1) by work group.



Figure 7. Boxplot of the responses for employee wellbeing by work group

Note. The graph represents a boxplot per work group for the survey item EWB 1.

Following the analysis of the boxplots, the standard deviation of the answers for all same-value responses of each participant was calculated using SPSS. If the standard deviation is between 0 and .25, this indicates that there is little or no variance in the responses, implying that the participant was unengaged and could therefore be removed from the dataset (Collier, 2020). After running the analysis, the present data did not contain standard deviations below .65. Hence, it can be assumed that all participants were engaged and that there were no outlier responses to be removed. The final dataset contained 237 responses grouped within 41 different work teams.

3.4.5.4. Multivariate normality

A normal distribution, which is also known as Gaussian or Laplace-Gauss distribution, is a continuous probability distribution for a real-valued random variable (Westfall & Henning, 2013). In other words, it is a distribution where a random variable normally deviates from the mean or another central point such as the median or the mode (see Equation 3; Stevens, 2009).

Equation 3. Probability density function for a normal distribution

$$f(x) = \frac{1}{\sigma\sqrt{2\pi}}e^{-\frac{1}{2}\left(\frac{x-\mu}{\sigma}\right)^2}$$

Where:

x is the random variable

 σ is the standard deviation of the distribution

 μ is the mean of the distribution or chosen the central tendency point

Before running a multilevel structural equation model, it is recommended to test for multivariate normality (Thode, 2002). There are different options for testing for normality. According to Thode (2002), a normally distributed variable needs to have a particular shape that can be plotted

using a histogram with a normal curve. Then the distribution of the variable is visually compared against the normal curve to check for potential normality problems. Additionally, the variable must also have a skewness and kurtosis between -2 and +2 (George & Mallery, 2010). Skewness and kurtosis statistics for all items of the survey can be found in Chapter 4.

3.4.5.6. Intraclass correlation

This study's two-level analysis includes the lower individual level of variance amongst employees and the higher level of the work team analysing the variance caused by the work team that the employees belong to. This study does not look at the influence of other higher levels on such variables or relationships. Another higher level that could be considered in future research is the organisational level.

Other potential levels of analysis include country and industry. However, for this study, these levels were not relevant. All the respondents belong to the same country, Australia. Only data from Australian organisations was collected, and thus the variance caused by the economy, or the country level cannot be studied. However, this could be studied if sufficient data were collected from many different countries. In this case, this was not possible due to time and money limitations. Equally, all participant organisations belong to the professional services industry, which encompasses all those occupations in the service sector that require specific training and licensing or a professional degree (Nishikawa & Orsato, 2021), such as architects, designers, accountants, engineers, lawyers, and teachers (Reid, 2008). As with the economy level, the industry level could also be studied if sufficient data from many different industries was collected. Moreover, three or four-level data analysis is more complex and requires high level technology to support it (Muthén & Muthén, 2009).

The organisational level could be a relevant level of analysis. However, in this case, this level was not considered for several reasons. First, the academic literature on emotional contagion theory, social exchange theory, and conservation of resource theory mentions that the relevant part of professional relationships is at the team level (Benitez et al., 2019; Kammerlander et al., 2017; Spell & Arnold, 2007; Wang et al., 2017). Barsade (2002) argues that emotional contagion happens via direct contact, whether face-to-face or online, with another person. As such, within large organisations, this interaction does not happen at the company level but at the team level through frequent contact between co-workers belonging to the same team.

Moreover, data from only nine organisations was collected. It is statistically impossible to perform multilevel data analysis with the available statistical software. Additionally, some argue that to have sufficient reason to undertake a multilevel analysis, the intraclass correlation (ICC) at that particular level needs to be above .1 to reduce estimation problems (Lüdtke et al., 2008).

González-Romá & Hernández (2017) suggest that the threshold should be .05. ICC is a statistical measure that describes the similarity between observation within a group. describes the variable strength within each cluster from 0 to 1, with 0 being no relationship and 1 being a strong intraclass relationship. ICC is calculated as the variance of the unobserved random effect over the total variance, which is calculated as the sum of the variance of the unobserved error (see Equation 4). It is commonly used in multilevel studies to understand the proportion of variance that resides at the team level, and additionally, the degree of dependence between two individual scores of the same unit or two different members of the same team (Kollo, 2008). Three level ICCs are presented in Chapter 4.

Equation 4. Intraclass correlation formula

$$ICC = \frac{\sigma_{\alpha}^2}{\sigma_{\alpha}^2 + \sigma_{\varepsilon}^2}$$

Where:

 σ_{α}^2 is the variance of the unobserved random effect shared by all values in the cluster σ_{ε}^2 is the variance of the unobserved error

3.4.5.7. Multilevel structural equation modelling

Multilevel structural equation modelling was chosen to test the hypotheses, which aligns with multilevel organisational theory and a post-positivist research philosophy (Kozlowski & Klein, 2000). This technique is used to study the relationships between observed and latent variables (Preacher et al., 2010). A latent variable is one that cannot be observed directly and instead needs to be indirectly inferred through another that can be observed. In this case, the concepts under the theoretical model, such as employee wellbeing levels or employees' perceptions of managerial priority, cannot be directly observed. Instead, these concepts are inferred (or observed) using psychometric questionnaires and a Likert scale that allows researchers to measure such constructs. Figure 8 represents the observed variables and how they are caused by variation in the latent variables at both the between and within levels.

Figure 8. Within and between variance in multilevel structural equation modelling using latent variables



Note. The observed variables are jointly caused by both the within and the between variation as latent variables. Adapted from *Multilevel SEM*, by M. Hallquist, 2017 Dec 7, https://psu-psychology.github.io/psy-597-SEM/15_multilevel/multilevel_sem.html. Copyright 2017 by Michael Hallquist.

Multilevel structural equation modelling investigates two types of relationships. First, the multilevel measurement model studies the relationship between the observed variable and the latent variables at the within and between levels. Second, the multilevel structural model analyses the causal relationship between latent variables (Preacher et al., 2010). As such, the goal of the analysis is to determine the statistical significance of the hypothesised model using the sample data.

The use of multilevel structural equation modelling offers four benefits according to Hallquist (2017). First, multilevel structural equation modelling uses path diagrams that support researchers in conceptualising complex multilevel structures. Second, it allows for mediation and moderation effects at both levels (within and between) (Preacher et al., 2010; 2016) as well as cross-level interactions (Aguinis et al., 2013). Third, standard multilevel modelling cannot incorporate models with latent variables, which can easily be implemented in multilevel structural equation modelling. Last, the incorporation of group mean centring within multilevel models could introduce an important bias at the between-level variance. With small sample sizes and intraclass correlations below .5, as is the case of the present study, multilevel structural equation modelling is the best statistical technique to test the theorised model.

Mplus is the best statistical package available to date to run multilevel structural equation modelling. As specified by Asparouhov and Muthén (2008), the formal equation of a multilevel structural equation model is specified in Equation 5.

Equation 5. Specification of multilevel structural equation model

$$Y_{ij} = \nu_j + \Lambda_j \eta_{ij} + K_j X_{ij} + \varepsilon_{ij}$$

Where:

 v_i are the intercepts of the variables that can vary according to the cluster

 Λ_i are the factor loadings

 η_{ii} are the factor scores

 X_{ii} are the influence of the exogenous covariate

3.4.5.7.1. Multilevel exploratory factor analysis

Exploratory factor analysis (EFA) is a statistical method that is used to determine the underlying structure of a set of observed variables (Watkins, 2018). This statistical technique analyses the underlying relationship between different survey items (Finch, 2020). Most journal articles that develop a new psychometric scale to measure a latent construct use EFA (Watkins, 2018). With this technique, researchers analyse the correlation rates between items by grouping them into factors. It is a statistical method that is not bound by theory. Instead, it is used to reveal correlations between items. Moreover, it helps to uncover validity issues that would be ignored by a confirmatory factor analysis (Costello & Osborne, 2005).

Because this research project uses clustered data within work teams, it is also necessary to use a multilevel EFA (D'Haenens et al., 2010). Traditional EFA also assumes that the observations are independent from each other, which is not the case with clustered data. A multilevel EFA is a more complex technique than the traditional single-level EFA, and it does not assume that the observations are independent (Huang & Cornell, 2016). Since 2009, Mplus has the programming code to run two-level EFA (Muthén & Muthén, 2022), which allows researchers to investigate the psychometric properties of instruments using grouped data.

Although this research project has used validated scales to measure the constructs included in the theorised model, the analysis of the measurement model began with a multilevel EFA. The main aim of running a multilevel EFA is to explore the collected data and determine if the multilevel theorised model makes sense from a statistical point of view (Dunn et al., 2015). If potential weaknesses are detected in the scales utilised to measure the research model, the researcher can address them earlier and make different decisions before proceeding with the following step of multilevel confirmatory factor analysis (Roesch et al., 2010).

3.4.5.7.2. Multilevel confirmatory factor analysis

Confirmatory factor analysis (CFA) is another statistical method used to verify the factor structure of a set of variables. This technique allows researchers to test the hypothesis that the underlying

latent construct exists and is indeed measured with that set of observed variables. Traditionally, EFA and CFA are run on different samples. However, some argue that they should be conducted using the same dataset (Kyriazos, 2018). If not, it is impossible to determine if any differences between the two are caused by methodological issues. Van Prooijen and van der Kloot (2001) point out that if both EFA and CFA are conducted using the same dataset, this would be the ultimate test for methodological explanations for any variances between both tests. Several research papers have adopted this approach since van Prooijen and van der Kloot's methodological paper was published in 2001 and report both EFA and CFA (e.g., Schmitt et al., 2018).

However, as with EFA, CFA also needs to be run at both the within and the between levels. According to Wu et al. (2017, p. 1), multilevel CFA "extends the power of CFA to accommodate the complex survey data with the estimation of the level-specific variance components and the respective measurement models". In other words, the hierarchical structure of the data can result in erroneous findings about the factor structure of the set of variables. As with EFA, the assumption that the observations are independent is violated due to their multilevel nature. According to Dyer et al. (2005, p. 151), there is also a need for a multilevel CFA "since the nature of the construct can differ across levels of analysis". As such, the factor analysis and discriminant validity results may be different across levels.

The traditional procedures that have been utilised to conduct multilevel CFA have been criticised. Some researchers have chosen to use the total covariance matrix derived from the full dataset to conduct the factor analysis. In so doing, they have ignored the multilevel structure of the data and therefore reached incorrect conclusions (Muthén, 1994). To prevent this, Muthén (1994) developed a five-step procedure to conduct multilevel CFA, and this is the procedure that has been adopted in this research project.

First, it is necessary to conduct a conventional CFA on the sample total covariance. Although the results may be biased as this step assumes that the observations are independent, the results are used as a baseline to compare with the following steps. The second step consists of estimating between-cluster level variation. The aim of this step is to understand if multilevel analysis is appropriate and needed for the collected dataset. The technique employed to determine if multilevel analysis is necessary is to check the ICC. The results of this analysis are presented in section 3.4.5.6. to justify a multilevel statistical analysis. Following this, the next step is to perform a factor analysis on the sample pooled-within covariance matrix. To this end, Muthén (1994) argues that it is beneficial to discretely analyse the within- and between-level sub-models. This step used a matrix with adjusted values that remove between-cluster differences. If the model

fit of the sample pooled-within covariance matrix is significantly better than that obtained in the first step, the within-level variance is higher than the between-level variance. Step 4 is to perform a factor analysis on the sample between-cluster covariance matrix. Results from this step would provide preliminary information about the appropriate group-level factor. However, as noted by Dyer et al. (2005), this is not possible with small group-level sample sizes, as is the case in this research project. Nonetheless, Dyer et al. (2005) argue that it is still possible to proceed to step 5 and finish the multilevel CFA without conducting the previous step.

The last step consists of performing the multilevel CFA. In this study, this step was run with two variables at a time using Mplus. Due to the small number of groups, a full multilevel CFA with all the variables included in the theorised model could present problems due to the small degrees of freedom. Results of the multilevel CFA can be found in Chapter 4.

3.4.5.8. Reliability and validity

Before analysing any dataset that uses multi-item psychological instruments, it is necessary to ensure that the instrument has a good reliability score. To this end, researchers should obtain and report reliability information of the instruments that are used in the research model. Reliability is often conceptualised as the internal consistency of the scale or the correlation among the latent and observed variables. It is the ratio between the true variance and the observed variance. The most common statistic for reporting reliability is Cronbach's alpha (α). Cronbach's alpha is also known as tau-equivalent reliability and measures internal consistency by multiplying the number of survey items with the average covariance between item-pairs and dividing this by the average variance plus the number of items minus one multiplied by the average covariance between item-pairs (see Equation 6).

Equation 6. Cronbach's alpha formula

$$\alpha = \frac{N \cdot \bar{c}}{\bar{v} + (N-1) \cdot \bar{c}}$$

Where:

N is the number of survey items \bar{c} is the average covariance between item-pairs \bar{v} is the average variance

However, Cronbach's alpha presents several limitations. It assumes both that the measures are unidimensional (single-level) and that they all have equal factor loadings, which is also known as essential tau equivalence. Both these assumptions pose problems for reporting multilevel reliability (Hayes & Coutts, 2020). First, reliability must be analysed using a tool that allows for

multidimensional measures. Second, essential tau equivalence appears to be rarely true in factor analysis (for both single-level and multilevel).

According to Hayes and Coutts (2020), the best alternative to Cronbach's alpha is McDonald's omega (ω). To find this omega, the latent and observed correlation must be directly estimated, and the ratio between the true variance and the total variance must be studied. The square root of omega is the true and total correlation between the latent and observed variables. To find McDonald's omega in Mplus, it first must be coded. In the present research, the analysis was completed following the code suggestions by Lai (2021).

In a latent variable (X) with three items (y_1, y_2, y_3) , each of these will have its own factor loading (λ_i) and its own error (ε_i) , as represented in Figure 9.

Figure 9. Path diagram of a latent construct with three items



Note. Path diagram of a latent construct (X) with three items (y_1, y_2, y_3) , each with its own error $(\varepsilon_1, \varepsilon_2, \varepsilon_3)$ and factor loading $(\lambda_1, \lambda_2, \lambda_3)$.

To calculate McDonald's omega, the first step is to fix the latent variance to 1 to obtain standardised values. Second, the true-score variance must be found by summing the freely estimated loadings and squaring them $(\Sigma(\lambda_i)^2)$. Then, the error variance must be identified; this can be done by summing the residual variance $(\Sigma(\varepsilon_i))$. To measure the omega, the true-score variance is divided by the total, which is the sum of the true-score variance and the error variance $(\Sigma(\lambda_i)^2 + \Sigma(\varepsilon_i))$, as per

Equation 7.

Equation 7. McDonald's omega formula.

$$\omega = \frac{\sum (\lambda_i)^2}{\sum (\lambda_i)^2 + \sum (\varepsilon_i)}$$

Where: λ_i are the freely estimated factor loadings ε_i are the residual variances

With a two-level model, the omega should be calculated at the within (individual) and at the between (team) level. This can also be done on Mplus by using *model constraint* syntax. The present study included two different omegas: one at the within and one at the between level. A sample of the Mplus code used to calculate the McDonald's omega for the employee wellbeing scale can be found in Appendix 5.

The McDonald's omega at the between levels tends to be very high due to the high correlation between variables at the cluster level. At the within level, traditional thresholds for other reliability metrics also apply. For example, Kline (2013) and Dunn et al. (2014) argue that the reliability index should be above .7 to be considered a reliable scale, while Wicke et al. (2022) claim that the minimum valid score is .8. Others have argued that lower reliability scores are also acceptable as long as the factor analysis validates that the factor and the latent variable represent the true score (Preacher et al., 2010). In a multilevel structural equation modelling, only the latent true score variance is used, while the error variance is not used. The McDonald's omegas for each of the utilised scales are presented in Chapter 4.

In the social sciences, and particularly in research projects that use a psychometric scale, it is common to have high correlations between constructs (Simonoff et al., 2013). As a result, it is necessary to test the discriminant validity of the scales to ensure that they measure only one construct. Different measures measuring different constructs cannot be strongly correlated; if they are, constructs may not be discriminating between each other, which can cause problems with the results of the model. In other words, if the variables are strongly correlated, the scales would be measuring the same concept, and therefore a significant regression between them would not be meaningful. According to Simonoff et al. (2013), correlation levels between variables above .7 present several complications. The method developed by Gorsuch (1983) to assess the discriminant validity of the variables was used in this model. This method has been extensively used in multilevel modelling (i.e., Farr-Wharton et al., 2021 and Sideridis et al., 2018). To test for discriminant validity, Gorsuch (1983) proposes to compare the model fit of the freely estimated multilevel CFA against a model where all correlations are constrained to 1. If the model fit of the constraint model is significantly worse than the freely estimated model, it can be

concluded that the discriminant validation of the tested variables is supported. Results of this comparison are presented in Chapter 4.

3.4.5.9. Multicollinearity

Multicollinearity is a statistical concept that happens when two or more of the independent variables of a regression model are highly correlated (Shieh & Fouladi, 2013). This high correlation amongst independent variables may or may not be casual, but it has important implications for analysis regardless. When regression or structural models suffer from multicollinearity, the effect of the independent variables on the dependent variables is overestimated and therefore less reliable (Alin, 2010).

For single-level structural equation models, different techniques have been developed to assess for multicollinearity. One of these is the VIF, which analyses the tolerance of a variable by the squared multiple correlation of the variable with all other predictor variables. However, with multilevel analysis, it is necessary to study the correlations between variables at both levels (within and between).

Correlation levels between variables at the within level tend to be low (Simonoff et al., 2013). However, at the between cluster level, multicollinearity issues are more persistent. Within social sciences, the correlation absolute value between items should be .5 for multicollinearity to be inexistent, but at the between level, the correlation absolute value between items of different constructs could be up to .7 for multicollinearity to be considered acceptable (Simonoff et al., 2013). Correlation levels at both levels of analysis are presented in Chapter 4.

3.4.5.10. Model fit

In structural equation modelling, *model fit* (also known as goodness of fit or global fit) can be defined as the analysis of how well the model represents the original dataset. To identify how good the model fits the population, there are several tests that can be used. These tests can be classified in two different groups: absolute and comparative. First, absolute fit indices which are a function of the test statistic. These absolute fit indices tend to quantify global fit using the model residuals or the population covariance structure. Second, comparative fit indices compare the specified model with a baseline model. The baseline model is a minimal model that normally only contains variances for observed endogenous variables. Moreover, the baseline model also assumes that there are no covariances among the different variables. As such, comparative fit indices compare the specified model with a model with a model where all variables are independent. This section presents the different model fit indices that have been used for both the measurement model and the specified multilevel structural model.

As the multilevel structural model is specified using random slopes and the between-level sample size is small, the estimator used is Bayes (see section 4.2.1). With Bayes estimator, the model fit indices that are generated by Mplus are not the frequently used statistics. These include the deviance (DIC) and the estimated number of parameters (pD). However, Bayes statisticians recommend to remove the random slope and use maximum likelihood estimator to report on model fit (Levy, 2011). It could be argued that when adding the random slope, the model fit will change but since the random slope can only improve the model, this is not a problem.

To test and report model fit, this study utilised the methodology developed by Anderson and Gerbing (1988). Hu and Bentler (1999) published the cut-off criteria for model fit indices in structural equation modelling. This thesis reports on five different model fit indices: Chi-square over degrees of freedom (CMIN/DF), comparative fit index (CFI), Tucker Lewis Index (TLI), standardised root mean square residual (SRMR), and the root mean square error of approximation (RMSEA).

First, Chi-square goodness of fit test is used to test if a set of variables come from a specified distribution or not. It is also used to determine if a sample dataset represents the population. The Chi-square is divided by the degrees of freedom. Degrees of freedom is the maximum number of independent values in the data set. A frequently used absolute goodness of fit statistic is Chi-square over degrees of freedom (CMIN/DF), which is considered good below 3, and acceptable below 5. However, this test has an important limitation. It is highly susceptible to the sample size. The bigger the sample size, the better results obtained. Therefore, it presents some problems with small samples, and it tends to be biased when using big datasets.

To overcome the sample size problem, the second goodness of fit test offers some solutions. Comparative fit index (CFI) is a model fit index that examines the discrepancy between the data and the hypothesised model and adjusts for the issue of sample size. The results of CFI vary from 0 to 1 and it is considered that the model has good fit when the CFI is above .95, and acceptable when it is above .9.

Third, Tucker Lewis Index (TLI) is a non-normed fit index that was developed by Tucker and Lewis (1973), and it also addresses the sample size issue that Chi-square presents. The results of TLI also vary from 0 to 1 and are considered good above .95, and acceptable above .9.

Fourth, Standardised Root Mean Square Residual (SRMR) analyses the average of standardised residuals between the observed and the hypothesised covariance matrices. A model is supposed to have an acceptable fit when the SRMR is between 0 and .08. Finally, the Root Mean Square Error of Approximation (RMSEA) tests for close or approximate fit instead of perfect or exact

fit. Testing for approximate fit has some benefits as it is not realistic to aim for a perfect fit with large populations and small sample sizes. RMSEA is considered acceptable below .08 and good below .05.

The results for all these model fit indices are presented for the measurement model and the structural model in Chapter 4.

3.5. Ethics and risks

Any research project faces ethical challenges (Clark et al. 2018; Hoque & Rana, 2018). Ethical principles and values are used to support the research project in reaching its goals while maintaining the rights of the research participants (Panter & Sterba, 2011). Internet-based research technologies have transformed the research environment and require extra considerations (Clark et al. 2018).

Ethical considerations are an essential part of any research to ensure data security and the safety of participants throughout the whole process. In this research, ethical considerations are especially important as primary data was collected directly from participants (employees), and some questions could cause risk for participants, even if the survey was designed to cause minimal disruption. It would be reasonable to expect that any distress experienced would be transient in nature for the majority of individuals. However, as it is important to consider the one participant who may react negatively to the questions, this section addresses all the potential risks. This project was categorised as high risk in the University of Technology Sydney (UTS) ethics assessment.

The ethical considerations associated with this project have been summarised in Table 22. These include the use of appropriate methods to achieve the project aim based on a thorough study of current literature, and the fact that the study was designed to ensure that respect for participants would not be compromised at any stage. All ethical considerations correspond to UTS ethics guidelines. As such, approval from the UTS Human Research Ethics Committee was sought prior to data collection and granted in August 2021. The COVID-19 pandemic did not have any major impact on the data collection as this was done remotely and online.

In addition, this project also aimed to be respectful to every participant, including the university and all researchers. For this reason, all participants were at liberty to make their own decision regarding participation. Consent was voluntary and given freely without coercion, pressure, or intimidation, and participants were fully informed using an information sheet in the survey. The participant information sheet included all relevant information as well as an independent contact from the UTS ethics research team in case any participant wished to make further comments. A
copy of the participant information sheet can be found in the Appendix 3. Participants were entitled to withdraw from the data collection at any stage, and all data collected was anonymous and kept in e-Research secure data storage that only designated researchers have access to in order to ensure privacy.

To identify which team a respondent belonged to, a unique identifier was needed for each team that every team member identified. As is common, the approach taken in this study was that employees generated a code based on their manager's first and last names. Collecting this data was necessary because it was to be a particular form of analysis that allowed for significant contributions and outcomes. While there was the potential for some identifiable information to emerge, the risk of this was mitigated by the storage processes that the researcher established. It is also important to note that the purpose of this study was not to collect identifiable information.

Ethical considerations	How have they been addressed?
National legislation and UTS	The researcher consulted with UTS ethics committee and
guidelines and policies.	gained UTS Human Research Ethics Committee approval
	prior to data collection.
UTS Ethics Application	The application was done in consultation with the Ethics
	Research team via email and in several Ethics Clinic
	sessions as well as with the support from the supervisors of
	the project to ensure that it included every ethical aspect that
	needed to be considered. UTS ethics committee provided
	approval before proceeding.
Use of the appropriate	The research which included human participant data, was
methodology	designed following UTS guidelines and policies as well as
	the literature, theory and with the supervision of my panel.
Information Security	The UTS data security classification was: Confidential.
Classification	
Gaining permission from	Any ethical standards that may have been required by the
organisations involved	organisations, as well as UTS, were consulted.
Collection and use of personal	Participation was voluntarily, free and a participant
information with consent of the	information sheet was provided at the beginning of the online
participant	survey. This PIS was designed following the UTS template.
	Additionally, participants could withdraw at any stage.

Table 22. Ethical considerations

Potential for participants to	Firstly, a PIS was provided at the start of the survey, and
experience harm, anything more	participants could withdraw at any stage. This ensured that
than discomfort	participation was voluntary, free, and fully informed. The
	survey had, in three different sections, the contact details of
	three organisations that could provide free support in case
	the survey evoked some issues: Beyond Blue, Lifeline and
	Head to Health. In the case of the participants experiencing
	some harm, they could reach out to any of these
	organisations. Also, the contact details of UTS Ethics
	Research team were available as an external party.
Ensuring data security	The Research Data Management Plan was in place to avoid
	any data breach. The data was collected through UTS
	Qualtrics, and all data was stored in a UTS e-Research
	secure location which had been organised with the UTS IT
	team.
Sharing of the research results	team. This was done through a benchmark report where the data
Sharing of the research results with the organisation	team. This was done through a benchmark report where the data was grouped in teams and anonymity of participants was
Sharing of the research results with the organisation	team. This was done through a benchmark report where the data was grouped in teams and anonymity of participants was assured. Data was not identifiable.
Sharing of the research results with the organisation Infringement of the privacy or	team. This was done through a benchmark report where the data was grouped in teams and anonymity of participants was assured. Data was not identifiable. Infringement of the privacy or professional reputation of the
Sharing of the research results with the organisation Infringement of the privacy or professional reputation of	team. This was done through a benchmark report where the data was grouped in teams and anonymity of participants was assured. Data was not identifiable. Infringement of the privacy or professional reputation of the participants, their managers, or the organisation was very
Sharing of the research results with the organisation Infringement of the privacy or professional reputation of participants, their managers, or	team. This was done through a benchmark report where the data was grouped in teams and anonymity of participants was assured. Data was not identifiable. Infringement of the privacy or professional reputation of the participants, their managers, or the organisation was very unlikely. Any potential identifiable data would be destroyed
Sharing of the research results with the organisation Infringement of the privacy or professional reputation of participants, their managers, or the organisation	team. This was done through a benchmark report where the data was grouped in teams and anonymity of participants was assured. Data was not identifiable. Infringement of the privacy or professional reputation of the participants, their managers, or the organisation was very unlikely. Any potential identifiable data would be destroyed upon usage or if any survey responses were disclosed to the
Sharing of the research results with the organisation Infringement of the privacy or professional reputation of participants, their managers, or the organisation	team. This was done through a benchmark report where the data was grouped in teams and anonymity of participants was assured. Data was not identifiable. Infringement of the privacy or professional reputation of the participants, their managers, or the organisation was very unlikely. Any potential identifiable data would be destroyed upon usage or if any survey responses were disclosed to the participant organisation nor any other party.
Sharing of the research results with the organisation Infringement of the privacy or professional reputation of participants, their managers, or the organisation	team. This was done through a benchmark report where the data was grouped in teams and anonymity of participants was assured. Data was not identifiable. Infringement of the privacy or professional reputation of the participants, their managers, or the organisation was very unlikely. Any potential identifiable data would be destroyed upon usage or if any survey responses were disclosed to the participant organisation nor any other party. The survey requested demographic information. This
Sharing of the research results with the organisation Infringement of the privacy or professional reputation of participants, their managers, or the organisation Identification of participants through data triangulation.	team. This was done through a benchmark report where the data was grouped in teams and anonymity of participants was assured. Data was not identifiable. Infringement of the privacy or professional reputation of the participants, their managers, or the organisation was very unlikely. Any potential identifiable data would be destroyed upon usage or if any survey responses were disclosed to the participant organisation nor any other party. The survey requested demographic information. This information was only collected to be used as mediators or
Sharing of the research results with the organisation Infringement of the privacy or professional reputation of participants, their managers, or the organisation Identification of participants through data triangulation.	team. This was done through a benchmark report where the data was grouped in teams and anonymity of participants was assured. Data was not identifiable. Infringement of the privacy or professional reputation of the participants, their managers, or the organisation was very unlikely. Any potential identifiable data would be destroyed upon usage or if any survey responses were disclosed to the participant organisation nor any other party. The survey requested demographic information. This information was only collected to be used as mediators or moderators of the research model. It was not to be presented
Sharing of the research results with the organisation Infringement of the privacy or professional reputation of participants, their managers, or the organisation Identification of participants through data triangulation.	team. This was done through a benchmark report where the data was grouped in teams and anonymity of participants was assured. Data was not identifiable. Infringement of the privacy or professional reputation of the participants, their managers, or the organisation was very unlikely. Any potential identifiable data would be destroyed upon usage or if any survey responses were disclosed to the participant organisation nor any other party. The survey requested demographic information. This information was only collected to be used as mediators or moderators of the research model. It was not to be presented to the organisation. This avoided any individual being

3.6. Conclusion of the chapter

This chapter presented the methodology adopted for this thesis. The aim of this chapter was to justify and explain the adopted research methods and to provide an extensive description of the research design. It began with the exploration of the approaches to employee wellbeing research. Employee wellbeing has been studied using both quantitative and qualitative research methods. However, to answer this project's research questions, this study focuses on a quantitative multilevel approach under a post-positivist research philosophy. Post-positivism and pragmatism

have also been extensively explored in this chapter. This research project sits under the umbrella of these two research paradigms. The logic used is mainly deductive and while there is a reality, there also needs to be a combination of different perspectives to help interpret the data. Therefore, the interpretation of the results takes a subjective approach that is impacted by the researcher's set of beliefs.

The chapter follows with an introduction to quantitative methods and in particular, to multilevel statistical analysis. Then, there is an in-depth presentation of this project's research design which has several parts. First, the research design begins with the population sampling and the target population under study. Second, the data collection process and the different steps that were followed are presented. The data was collected using an online survey programmed using Qualtrics. Collecting data using an online survey has some important implications that are explored in the third section. Fourth, the section was followed by a comprehensive presentation of the different questionnaire scales that were adopted for the different constructs included in the theorised model: employee wellbeing, work stress, emotional contagion, psychosocial safety climate, and Leader-member exchange. Fifth, the chapter continued with the complex data analysis process that was undertaken including the data preparation, how missing data and outliers were addressed, the sample demographics, multivariate normality, intraclass correlation, the different steps to conduct a multilevel structural equation modelling, reliability and validity of the scales used, multicollinearity and model fit indices. The chapter concluded with the ethical considerations and the risk management plan adopted in this study.

Chapter 4. Results

This chapter presents the results and findings from the analysis of the survey data. The purpose of this chapter is to present the results of the data analysis that tested the hypothesised relationships between employee wellbeing, work stress, psychosocial safety climate, Leadermember exchange, and emotional contagion. However, before proceeding with the structural equation model, this chapter presents the variable descriptive statistics: minimum, maximum, mean, and standard deviation. This is followed by the sample demographics: gender, year of birth and tenure. Finally, the chapter presents the results of the model. Within this study, there are two categories of statistical models. Firstly, the measurement model, also called the outer model. And second, the structural model, or inner-model. Results of both are presented in this chapter. The measurement model section presents the results of the multivariate normality test, the intraclass correlations that justify a two-level analysis, the exploratory factor analysis and the multilevel confirmatory factor analysis, where the relationship between the latent variables and their scales were examined. This part is followed by the results of the multilevel structural equation modelling. The multilevel structural equation model presents the relationship between the latent variables, including the cross-level moderating effect of emotional contagion on the relationship between work stress and employee wellbeing and the double or chain mediation between psychosocial safety climate and employee wellbeing by Leader-member exchange and work stress. Finally, this chapter concludes with a summary of key findings resultant from the data analysis.

4.1. Descriptive statistics

Descriptive statistics help researchers understand the dataset before running any other statistical analysis techniques. After data preparation, addressing missing data, and checking for outliers, the final dataset had 237 employees within 41 work teams. This section presents the minimum, maximum, mean and standard deviation of each variable included in the hypothesised model. All the descriptive statistics have been rescaled from their Likert-scale to a scale from 0 to 1 to simplify their readability. The results of these descriptive statistics are presented in Table 23.

The minimum for employee wellbeing was .2 and the maximum 1. Employee wellbeing had a mean statistic of .61 and a standard deviation of .16. This suggests that overall, the sample experience moderate levels of employee wellbeing. The minimum work stress level was 0 and the maximum .75. Work stress had a mean statistic of .45 and a standard deviation of .17. These results suggest that the sample of employees experience moderate levels of work stress. For psychosocial safety climate, the minimum score was 0 and the maximum 1. The mean statistic was .65 with a standard deviation of .21. This suggests that the cohort perceives a slight prioritisation of their wellbeing above productivity by management. The Leader-member

exchange minimum value was 0 and the maximum was 1. The mean statistic was .67 with a standard deviation of .2. These results suggest that the sample perceived a slight level of support from their line manager. Finally, the minimum value of emotional contagion amongst the participants was 0 and the maximum was 1. The mean statistic was .42 and the standard deviation was .22. This suggests that the cohort experiences moderate to low levels of emotional contagion within their work teams.

Variable	Minimum	Maximum	Mean	Standard Deviation
Employee wellbeing	.2	1	.61	.16
Work stress	0	.75	.45	.17
Psychosocial safety climate	0	1	.65	.21
Leader-member exchange	0	1	.67	.20
Emotional contagion	0	1	.42	.22

Table 23. Normalised minimum, maximum, mean and standard deviation of the model variables on a scale of 0 to 1

Note. N = 237 participants

4.2. Sample demographics

Demographics are statistics that explain the characteristics of a population or sample. After data preparation, addressing missing data, and checking for outliers, the final group count was 237 employees within 41 work teams. The survey included three demographic questions on gender, year of birth, and tenure.

The frequency analysis of the demographic questions can be found in Table 24, which shows that 70% of responders (166) were female, 30% were male (71), and none of the participants selected other or "prefer not to disclose". Of the total sample, 42% (100) of respondents were born between 1990 and 1999, 34% (81) were born between 1980 and 1989, 12% (29) were born between 1970 and 1979, 7% (16) were born between 1960 and 1969, 3% (7) were born after 2000, and 2% (4) were born between 1950 and 1959. The largest proportion of participants (33%; 79), had been employed for one to two years by their current organisation, while 27% (64) had been employed by the organisation for three to five years and another 27% (63) for less than a year. Finally, 10% (24) had been employed by their current organisation for six to 10 years, 2% (4) for over 15 years, and 1% (3) for 11 to 15 years.

Demographic	Variable	Ν	%
Gender	Female	166	70
	Male	71	30
	Other or prefer not to disclose	0	0
Year of birth	1950-1959 (62-71 years old)*	4	2
	1960-1969 (52-61 years old)*	16	7
	1970-1979 (42-51 years old)*	29	12
	1980-1989 (32-41 years old)*	81	34
	1990-1999 (22-31 years old)	100	42
	2000-2009 (19-21 years old)*	7	3
Tenure	< 1 year	63	27
	1-2 years	79	33
	3-5 years	64	27
	6-10 years	24	10
	11-15 years	3	1
	> 15 years	4	2

Table 24. Frequency analysis of the demographic questions

Note. N = 237 participants; * at the data collection time

4.3. Results of measurement model

Before presenting the results of the measurement model, it is necessary to comment on two additional tests. This section begins with the results of the multivariate normality test. Then, the intraclass correlation of all measures is presented to justify the need of a two-level analysis. Finally, the measurement model is presented using two factor analysis as suggested by Costello and Osborne (2005). First, results of the multilevel exploratory factor analysis are presented to help uncover validity issues that could be ignored if this step was skipped. Second, the section presents the results of the complex multilevel confirmatory analysis. This last factor analysis is run on each variable, including the multilevel observed intraclass correlation coefficients for all items, the multilevel correlation between items, the scale model fit, its factor loadings, the reliability index using McDonald's omega, and its validity and collinearity statistical analysis.

4.3.1. Multivariate normality

Thode (2002) suggests testing for multivariate normality before running a multilevel structural equation model. In fact, it is also recommended with single level models. A normally distributed variable has a particular shape that can be plotted in a histogram and compared against the normal

curve. Moreover, the variable must also have a skewness and kurtosis between -2 and +2 (George & Mallery, 2010).

These tests were run using SPSS, and all continuous variables fell within the threshold mentioned. This test cannot be run on categorical variables such as gender or tenure. Table 25 presents the skewness and kurtosis statistics for all survey items.

	Skewness	Kurtosis
EWB_1	-0.772	0.341
EWB_2	-0.595	0.065
EWB_3	-0.673	0.123
EWB_4	-0.592	-0.072
WS_1	0.574	-0.452
WS_2	-0.018	-1.043
WS_3	0.413	-0.276
WS_4	-0.794	0.106
WS_5	0.550	-0.600
PSC_1	-0.720	0.061
PSC_2	-0.748	0.153
PSC_3	-0.282	-0.522
PSC_4	-0.284	-0.487
LMX_1	-0.557	-0.472
LMX_2	-0.421	-0.686
LMX_3	-0.540	-0.496
LMX_4	-0.792	0.126
LMX_5	-0.257	-0.650
LMX_6	-0.641	0.152
LMX_7	-0.409	0.160
EC_1	-0.555	0.454
EC_2	-0.623	0.278
EC_3	-0.010	-0.680
EC_4	0.157	-0.738
EC_5	-0.143	-0.894
EC_6	-0.025	-0.465

Table 25. Skewness and kurtosis of all survey items

Note. N=237; 41 clusters.

4.3.2. Intraclass correlation

To answer the research questions and test the hypotheses, this research project draws upon a twolevel structural equation model. The within level which is also known as the individual level has the employees. The between or cluster-level has the work teams which the employees are nested in. However, since data from nine different organisations was collected, some would suggest a three-level analysis. A three-level analysis is statistically impossible with the available statistical software packages and only nine organisations at the highest level of analysis. Additionally, some argue that to have sufficient reason to undertake a multilevel analysis, the intraclass correlation (ICC) at that particular level needs to be above .1 to reduce estimation problems (Lüdtke et al., 2008).

The ICC at the company between level was calculated for all the variables to verify its relevance and they all presented scores below .1. For employee wellbeing, the ICC at the company between level was of .004. For work stress, it was even lower, with a score of .003. For psychosocial safety climate, the ICC at the company between level has a score of .029. While for Leader-member exchange, the ICC was .088. Finally, emotional contagion presents a score of .081, still below the threshold of .1. The ICC at the team level was also calculated for all the variables and they all presented scores above .1. For example, the ICC at the team level for employee wellbeing was .373, while for work stress was .258. For psychosocial safety climate, it is .321; for Leadermember exchange, it is .335; and for emotional contagion, it is .181. A summary of these results can be found in the Table 26.

Variable	ICC company level	ICC team level
Employee wellbeing	.004	.373
Work stress	.004	.236
Psychosocial safety climate	.029	.321
Leader-member exchange	.088	.326
Emotional contagion	.081	.214

Table 26. Intraclass correlation at the company between level for the model variables

Note. N = 237 participants, 41 team-level clusters, 9 company-level clusters.

There is some controversy surrounding the threshold of the ICC. Some scholars have argued that in social sciences, it is sufficient to do a multilevel analysis with an ICC of .05 or even as low as .03 (Meuleman & Billiet, 2009). However, the literature does not present a case for such analysis and with only nine participating organisations, it was not possible to run a three-level analysis. Because of the low ICC at the company level, and because it aligned better with the research questions, this research project conducted a two-level analysis with employees at the individual or within level and work teams at the group or between level.

4.3.3. Multilevel exploratory factor analysis

An exploratory factor analysis (EFA) or theory building approach is a statistical technique that is used to analyse the correlation amongst all the measures under consideration. It is also used to explore the underlying theoretical structure of the construct (Finch, 2020). It is a statistical method that is not bound by theory. Instead, it is used to reveal correlations between items. According to Costello and Osborne (2005), EFA helps to uncover validity issues that would be ignored by a confirmatory factor analysis. Although this research project has used validated scales to measure the constructs included in the theorised model, multilevel EFA is used explore the collected data and determine if the multilevel theorised model makes sense from a statistical point of view (Dunn et al., 2015).

A multilevel EFA was conducted in Mplus v8.8 using the default rotation settings. The default rotation method in Mplus is GEOMIN, which has shown good performance in other studies (Preacher, 2017). The sample syntax of the multilevel EFA run with psychosocial safety climate and emotional contagion can be found on the Appendix 4. Due to the sample size and the Mplus specification, the multilevel EFA was run with two sets of variables at a time. Overall, ten multilevel EFAs were run on Mplus. As expected, in all tests, the artifactual second factor disappeared when the model was properly specified with two within factors and two between factors. There was only one item that did not load as expected. When conducting the multilevel EFA, it was found that an item (EC_2) of the emotional contagion scale did not load as expected. But, the item was retained, as this was only an exploratory analysis, and it would only have been removed if it also presented issues after the multilevel confirmatory factor analysis. The results of such are presented in the following section.

4.3.4. Multilevel confirmatory factor analysis

A confirmatory factor analysis (CFA) or theory confirming approach begins with a theory about what construct underlies a group of measure and then tests whether the correlations amongst these measures suggest that they are valid indicators of the underlying construct. In this project, the multilevel CFA was also conducted in Mplus v8.8. Due to the relatively small sample size, the decision was made to do an individual multilevel CFA and then, to compare the correlation amongst variables as per Zyphur's (2022) process. The following subsections present the observed intraclass correlation for each scale item, as well as the overall intraclass correlation for the scale. Then, the descriptive statistics are disclosed along with the model fit indicators that are prescribed by Hu and Bentler (1999). The factor loadings of each item are also included. Finally, the reliability of the scale is analysed using the McDonald's omega and each section finishes with an analysis of the correlation between variables.

4.3.4.1. Employee wellbeing

a. Observed intraclass correlation

The observed intraclass correlation of the employee wellbeing scale was 37.9%, indicating a moderate multilevel effect. The intraclass correlation of each item can be found in Table 27. All items present acceptable ICCs.

Table 27. Intraclass correlations of the employee wellbeing scale items

Item	ICC
EWB_1	.332
EWB_2	.262
EWB_3	.367
EWB_4	.347

Note. N=237; 41 clusters.

b. Model fit

The threshold criteria for model fit published by Hu and Bentler (1999) were considered when analysing the goodness of fit of the latent variable for employee wellbeing. This study reports on four different model fit indices for the latent variables. First, comparative fit index (CFI) and Tucker Lewis index (TLI) scored 1, which is considered perfect fit. Mplus offers standardised root mean square residuals (SRMR) for both levels of analysis: between and within. The SRMR at the within level was .013, and at the between level was .017. Both levels offer a good model fit according to SRMR as it is considered good between 0 and .08. Finally, the root mean square error of approximation (RMSEA) scored 0, which is also considered perfect fit. Overall, it can be concluded that the latent variable for employee wellbeing offered a good fit as all goodness of fit indices scored within their perfect, good or acceptable thresholds.

c. Factor loadings

Table 28 presents the standardised within and between factor loadings of the employee wellbeing scale. EWB_1 has a within standardised factor loading of .741 and a between standardised factor of .802. EWB_2's standardised factor loadings are .742 and .986, respectively. EWB_3 within standardised factor loading is .853 and the between standardised factor loading is .963. Finally, EWB 4's standardised factor loadings are .836 and .979, respectively.

Item	Within	Between
EWB_1	.741	.802
EWB_2	.742	.986
EWB_3	.853	.963

Table 28. Standardised within and between factor loadings of the employee wellbeing scale

EWB_4	.836	.979

Note. N=237; 41 clusters.

d. Reliability

The multilevel reliability of the employee wellbeing scale has been measured using the McDonald's omega (ω) as described in the section 3.4.5.8. A copy of the Mplus code used to calculate the McDonald's omega can be found in Appendix 5.

At the between level (team level), the McDonald's omega for employee wellbeing equals 0.968 with a correlation of 0.984 between the items. At the within level (employee level), the omega for EWB equals 0.866 with a correlation of 0.931 between the items at the individual level. Following the threshold levels suggested by Kline (2013) and Dunn et al. (2014), as stated in section 3.4.5.8, this scale presents a good reliability. A summary of all the calculated parameters to measure the McDonald's omega can be found in Table 29.

Table 29. Parameters to measure the McDonald's omega for the employee wellbeing scale

	Between level	Within level
$\sum (\lambda_i)^2$	5.785	8.452
$\sum (\varepsilon_i)$	0.190	1.309
ω	0.968	0.866
$\sqrt{\omega}$	0.984	0.931

4.3.4.2. Work stress

a. Observed intraclass correlation

The observed intraclass correlation of work stress was 24.4%, indicating a moderate to low multilevel effect. The intraclass correlation of each item of the work stress scale can be found on Table 30. All items present acceptable ICCs.

Table 30. Intraclass correlations of the work stress scale items

Item	ICC
WS_1	.201
WS_2	.206
WS_3	.225
WS_4	.127
W_5	.226

Note. N=237; 41 clusters.

b. Model fit

The threshold criteria for model fit published by Hu and Bentler (1999) were considered when analysing the goodness of fit of the latent variable for work stress. This study reports on four different model fit indices for the latent variables. First, comparative fit index (CFI) and Tucker Lewis index (TLI) scored 1, which is considered perfect fit. Mplus offers standardised root mean square residuals (SRMR) for both levels of analysis: between and within. The SRMR at the within level was .000, and at the between level was .003. Both levels offer a good model fit according to SRMR as it is considered good between 0 and .08. Finally, the root mean square error of approximation (RMSEA) scored 0, which is also considered perfect fit. Overall, it can be concluded that the latent variable for work stress offered a good fit as all goodness of fit indices scored within their perfect, good or acceptable thresholds.

c. Factor loadings

Table 31 presents the standardised within and between factor loadings of the work stress scale. WS_1 has a within standardised factor loading of .790 and a between standardised factor of .975. WS_2's standardised factor loadings are .731 and .988, respectively. WS_3 within standardised factor loading is .757 and the between standardised factor loading is .821. WS_4 within standardised factor loading is .698 and the between standardised factor loading is .814. Finally, WS_5's standardised factor loadings are .796 and .871, respectively.

Item	Within	Between
WS_1	.790	.975
WS_2	.731	.988
WS_3	.757	.821
WS_4	.698	.814
WS_5	.796	.871

Table 31. Standardised within and between factor loadings of the work stress scale

Note. N=237; 41 clusters.

d. Reliability

The multilevel reliability of the employee wellbeing scale has been measured using the McDonald's omega (ω) as described in section 3.4.5.8.

At the between level (team level), the McDonald's omega for WS equals 0.908 with a correlation of 0.953 between the items. At the within level (employee level), the omega equals 0.853 with a correlation of 0.923 between the items at the individual level. Following the threshold levels suggested by Kline (2013) and Dunn et al. (2014), as stated in the section 3.4.5.8, this scale

presents a good reliability. A summary of all the calculated parameters to measure the McDonald's omega can be found in the Table 32.

	Between level	Within level
$\sum (\lambda_i)^2$	1.603	3.314
$\sum(\varepsilon_i)$	0.162	0.571
ω	0.908	0.853
$\sqrt{\omega}$	0.953	0.923

Table 32. Parameters to measure the McDonald's omega for the work stress scale

4.3.4.3. Psychosocial safety climate

a. Observed intraclass correlation

The observed intraclass correlation of the psychosocial safety climate 4-item scale was 35%, indicating a moderate multilevel effect. The intraclass correlation of each item of the scale can be found on Table 33. All items present acceptable ICCs.

Table 33. Intraclass correlations of the psychosocial safety climate scale items

Item	ICC
PSC_1	.280
PSC_2	.292
PSC_3	.263
PSC_4	.315

Note. N=237; 41 clusters.

b. Model fit

The threshold criteria for model fit published by Hu and Bentler (1999) were considered when analysing the goodness of fit of the latent variable for psychosocial safety climate. This study reports on four different model fit indices for the latent variables. First, comparative fit index (CFI) scored .995 and Tucker Lewis index (TLI) scored 1, which are considered good and perfect fit, respectively. Mplus offers standardised root mean square residuals (SRMR) for both levels of analysis: between and within. The SRMR at the within level was .011, and at the between level was .015. Both levels offer a good model fit according to SRMR as it is considered good between 0 and .08. Finally, the root mean square error of approximation (RMSEA) scored 0, which is also considered perfect fit. Overall, it can be concluded that the latent variable for psychosocial safety climate offered a good fit as all goodness of fit indices scored within their perfect, good or acceptable thresholds.

c. Factor loadings

Table 34 presents the standardised within and between factor loadings of the psychosocial safety climate scale. PSC_1 has a within standardised factor loading of .769 and a between standardised factor of .912. PSC_2's standardised factor loadings are .756 and .881, respectively. PSC_3 within standardised factor loading is .765 and the between standardised factor loading is .901. Finally, PSC_4's standardised factor loadings are .753 and .881, respectively.

Table 34. Standardised within and between factor loadings of the psychosocial safety climate scale

Item	Within	Between
PSC_1	.769	.912
PSC_2	.756	.881
PSC_3	.765	.901
PSC_4	.753	.881

Note. N=237; 41 clusters.

d. Reliability

The multilevel reliability of the employee wellbeing scale has been measured using the McDonald's omega (ω) as described in section 3.4.5.8.

At the between level (team level), the McDonald's omega for PSC equals 0.978 with a correlation of 0.989 between the items. At the within level (employee level), the omega equals 0.861 with a correlation of 0.928 between the items at the individual level. Following the threshold levels suggested by Kline (2013) and Dunn et al. (2014), as stated in the section 3.4.5.8, this scale presents a good reliability. A summary of all the calculated parameters to measure the McDonald's omega can be found in the Table 35.

	Between level	Within level
$\sum (\lambda_i)^2$	4.629	7.685
$\sum (\varepsilon_i)$	0.106	1.245
ω	0.978	0.861
$\sqrt{\omega}$	0.989	0.928

Table 35. Parameters to measure the McDonald's omega for the psychosocial safety climate scale

4.3.4.4. Leader-member exchange

a. Observed intraclass correlation

The observed intraclass correlation of EWB was 44%, indicating a moderate multilevel effect. The intraclass correlation of each one of the seven items of the construct can be found in Table 36. All items present acceptable ICCs.

Item	ICC
LMX_1	.335
LMX_2	.315
LMX_3	.267
LMX_4	.434
LMX_5	.333
LMX_6	.393
LMX_7	.275

Table 36. Intraclass correlations of the employee wellbeing scale items

Note. N=237; 41 clusters.

b. Model fit

The threshold criteria for model fit published by Hu and Bentler (1999) were considered when analysing the goodness of fit of the latent variable for Leader-member exchange. This study reports on four different model fit indices for the latent variables. First, comparative fit index (CFI) scored .978 and Tucker Lewis index (TLI) scored .964, which are considered good above .95. Mplus offers standardised root mean square residuals (SRMR) for both levels of analysis: between and within. The SRMR at the within level was .036, and at the between level was .047. Both levels offer a good model fit according to SRMR as it is considered good between 0 and .08. Finally, the root mean square error of approximation (RMSEA) scored .056, which is considered acceptable below .08. Overall, it can be concluded that the latent variable for Leader-member exchange offered a good fit as all goodness of fit indices scored within their good or acceptable thresholds.

c. Factor loadings

Table 37 presents the standardised within and between factor loadings of the Leader-member exchange scale. LMX_1 has a within standardised factor loading of .793 and a between standardised factor of .970. LMX_2's standardised factor loadings are .748 and .963, respectively. LMX_3 within standardised factor loading is .801 and the between standardised factor loading is .968. LMX_4 within standardised factor loading is .701 and the between standardised factor loading is .984. LMX 5's standardised factor loadings are .795 and .820, respectively. LMX 6

within standardised factor loading is .790 and the between standardised factor loading is .989. Finally, LMX_7's standardised factor loadings are .730 and .774, respectively.

Item	Within	Between
LMX_1	.793	.970
LMX_2	.748	.963
LMX_3	.801	.968
LMX_4	.701	.984
LMX_5	.795	.820
LMX_6	.790	.989
LMX_7	.730	.885

Table 37. Standardised within and between factor loadings of the Leader-member exchange scale

Note. N=237; 41 clusters.

d. Reliability

The multilevel reliability of the employee wellbeing scale has been measured using the McDonald's omega (ω) as described in the section 3.4.5.8.

At the between level (team level), the McDonald's omega for LMX equals 0.984 with a correlation of 0.992 between the items. At the within level (employee level), the omega equals 0.869 with a correlation of 0.932 between the items at the individual level. Following the threshold levels suggested by Kline (2013) and Dunn et al. (2014), as stated in section 3.4.5.8, this scale presents a good reliability. A summary of all the calculated parameters to measure the McDonald's omega can be found in Table 38.

	Between level	Within level
$\sum (\lambda_i)^2$	14.185	16.070
$\sum (\varepsilon_i)$	0.234	2.413
ω	0.984	0.869
$\sqrt{\omega}$	0.992	0.932

Table 38. Parameters to measure the McDonald's omega for the Leader-member exchange scale

4.3.4.5. Emotional contagion

a. Observed intraclass correlation

The observed intraclass correlation of the emotional contagion scale was 37.9%, indicating a moderate multilevel effect. The intraclass correlation of each item can be found on Table 39. The only item that presents some initial problems is EC_2, as highlighted in the multilevel EFA.

Item	ICC
EC_1	.134
EC_2	.056
EC_3	.269
EC_4	.284
EC_5	.239
EC_6	.307

Table 39. Intraclass correlations of the emotional contagion scale items

Note. N=237; 41 clusters.

b. Model fit

Following Wieland et al.'s (2017) recommendations for scale purification, the item EC_2 was removed from the scale to enhance model fit. After rechecking the item ("At work, if colleagues appear to be worried about something, I generally remain calm"), it appeared that it was the only item that was reversely worded and not in accordance with the other items of the scale. Therefore, it appeared appropriate to remove the item from the scale.

The threshold criteria for model fit published by Hu and Bentler (1999) were considered when analysing the goodness of fit of the latent variable for the purified emotional contagion. This study reports on four different model fit indices for the latent variables. First, both comparative fit index (CFI) and Tucker Lewis index (TLI) scored 1, which is considered perfect fit. Mplus offers standardised root mean square residuals (SRMR) for both levels of analysis: between and within. The SRMR at the within level was .000, which is also considered perfect fit. At the between level, SRMR was .001, which is considered good between 0 and .08. Finally, the root mean square error of approximation (RMSEA) scored .000, which is also considered perfect fit. Overall, it can be concluded that the latent variable for the purified emotional contagion offered a good fit as all goodness of fit indices scored within their perfect, good or acceptable thresholds.

c. Factor loadings

Table 40 presents the standardised within and between factor loadings of the purified emotional contagion scale. EC_1 has a within standardised factor loading of .736 and a between standardised factor of .827. EC_3's standardised factor loadings are .738 and .888, respectively. EC_4 within standardised factor loading is .768 and the between standardised factor loading is .939. EC_5 within standardised factor loading is .767 and the between standardised factor loading is .965. Finally, EC_6's standardised factor loadings are .606 and .918, respectively.

Item	Within	Between
EC_1	.736	.827
EC_3	.738	.888
EC_4	.768	.939
EC_5	.767	.965
EC_6	.606	.918

Table 40.Standardised within and between factor loadings of the purified emotional contagion scale

Note. N=237; 41 clusters.

d. Reliability

The multilevel reliability of the employee wellbeing scale was measured using the McDonald's omega (ω) as described in section 3.4.5.8.

At the between level (team level), the McDonald's omega for EC equals 0.918 with a correlation of 0.958 between the items. At the within level (employee level), the omega equals 0.800 with a correlation of 0.894 between the items at the individual level. Following the threshold levels suggested by Kline (2013) and Dunn et al. (2014), as stated in section 3.4.5.8, this scale presents a good reliability. A summary of all the calculated parameters to measure the McDonald's omega can be found in Table 41.

	Between level	Within level
$\sum (\lambda_i)^2$	2.061	4.344
$\sum (\varepsilon_i)$	0.184	1.087
ω	0.918	0.800
$\sqrt{\omega}$	0.958	0.894

Table 41. Parameters to measure the McDonald's omega for the emotional contagion scale

4.3.4.6. Discriminant validity and correlations

Discriminant validity is checked to ensure that each item measures only one construct (Simonoff et al., 2013). Different measures measuring different constructs cannot be strongly correlated. If they are, constructs may not be discriminating between each other, which can cause problems with the results of the model. According to Simonoff et al. (2013), correlation levels between variables above .7 present several complications. The study used the method developed by Gorsuch (1983) to assess the discriminant validity. To test for discriminant validity, Gorsuch (1983) suggests that the model fit of the freely estimated multilevel CFA against a model where all correlations are constrained to 1. If the model fit of the constraint model is significantly worse

than the freely estimated model, it can be concluded that the discriminant validation of the tested variables is supported. Since the dataset is small, this test for discriminant validity could not be run.

Due to the between-level sample size, this test was done with two variables at a time and in all cases the constrained model was statistically inferior. Therefore, it can be concluded that the discriminant validity of the tested variables was supported. Moreover, the correlation analysis yielded no significant associations between the tested variables at the between level or at the within level. At the within level, all correlations between items were below .4, whereas at the between group level, PSC_2 and LMX_6 had a correlation of .819, and EWB_2 and PSC_3 had a correlation of .819. While these two are high correlation levels, van de Schoot and Hox (2013) would argue that multilevel structural equation modelling could still be performed.

4.4. Results of the multilevel structural model

The multilevel structural equation model was built in Mplus adapting the complex syntax and methodology developed by Preacher et al. (2010; 2016) to test multilevel mediation and moderation using multilevel structural equation modelling. This study used multilevel double mediation and moderation with random slopes. The script can be found in Appendix 6. The output requested Tech8 to request the optimization history in estimating the model. The model ran with 10,000 iterations and the potential scale reduction converged to 1.003. The estimator used was Bayes and the justification for this can be found in section 4.4.1. This chapter also presents the model fit results for the multilevel structural model and the results of the hypothesis testing.

4.4.1. Bayes estimator

With any statistical analysis, it is necessary to use an estimator. An estimator is a statistic technique to estimate some characteristic about the population using a sample. In other words, an estimator is the technique employed to create estimates. As with theoretical frameworks, there are many estimators that can be utilised to estimate a fact about the population. This research project used the Bayes estimator. In particular, the Mplus default programming for the Bayes estimator was used. Although the Bayesian approach is not the default estimator used in the majority of structural equation modelling analyses, some argue that it is the best estimator (Lee, 2007).

Bayes estimator presents some important benefits in contrast to other estimators such as the default maximum likelihood estimator (Muthén, 2010). First, frequentists use estimators like the maximum likelihood estimator and view parameters as constants. While on the other hand, Bayesian analysts see parameters as variables and in order to compose posterior distributions for parameter estimates, they combine prior distributions for parameters with the data likelihood.

Second, maximum likelihood estimator assumes that the distribution of the parameter estimates is non-parametric based on asymptotic large-sample theory. As such, to use maximum likelihood, a large sample size in needed. As this is a two-level model with a small number of clusters, this assumption is false. In contrast, Bayes estimator does not depend on the large-sample theory and does not assume that the parameter estimates distribution is normal. This would allow parameter estimates distributions that are strongly skewed or that present tailedness different from the normal distribution. Third, Muthén (2010) suggests that Bayes is an optimal estimator in instances where maximum likelihood estimator would not be able to compute the parameter estimates. Fourth, Bayes estimator provides a natural approach that cannot be established using other estimators with models with a very large number of parameters.

In summary, the Bayes estimator is the best estimator to test this project's hypothesis. Bayes makes fewer assumptions with small samples and therefore, it reduces the potential error. Moreover, as this project uses multilevel moderation with random slopes, it is important to consider how Mplus treats them. Hoffman and Walters (2022) provide an in-depth analysis on current best practices for the specification of multilevel models. They suggest that when modelling multilevel moderation with random slopes, it is recommended to use Bayes as the model estimator. The primary reason is that with the maximum likelihood estimator, Mplus cannot take the latent mean centring and the within and between decomposition. This problem means that the between regression coefficient cannot predict the between effect and instead, it becomes a contextual effect. For this, Mplus software is ideally suited for this project because, with its implementation of the Bayes estimator, it gets the multilevel decomposition that is appropriate to use with random slopes (Asparouhov & Muthén, 2019; Hoffman, 2019; Hoffman & Walters, 2022). González-Romá & Hernández (2017) point out the significance of the Bayes estimator with small number of clusters. In their review of multilevel SEM, they state that with a small number of teams, Bayes estimator helps predict better estimates with random slopes. Finally, this project takes the Asparouhov and Muthén (2010) approach on Bayes estimator for multilevel moderation models with random slopes and two-level random interactions. Asparouhov and Muthén (2010) provide a technical discussion on how Bayes estimator has been implemented in Mplus, the statistical software package used to test the hypotheses.

4.4.2. Model fit

With Bayes estimator, it is not possible to obtain the traditional model fit indices using the methodology developed by Anderson and Gerbing (1988). However, as reporting on traditional fit indices is best practice when doing structural equation modelling, some Bayes statisticians suggest that it is recommended to remove the random slope and use maximum likelihood estimator to report on model fit (Levy, 2011). Moreover, it could be argued that when adding the

random slope, the model fit can only improve the model, which justifies removing it to report on traditional model fit indices. Therefore, the reported fit indices are for the model without random slopes.

The cut-off criteria for model fit published by Hu and Bentler (1999) were considered when analysing the goodness of fit. This study reports on five different model fit indices: Chi-square over degrees of freedom (CMIN/DF), comparative fit index (CFI), Tucker Lewis Index (TLI), standardised root mean square residual (SRMR), and the root mean square error of approximation (RMSEA).

CMIN/DF had a score of 2.88 which is considered a good model fit as it is below 3. The CFI score was .994, which is considered good fit as it is above .95. The Tucker Lewis Index (TLI) score was .961 and it is considered good above .95. Mplus offers SRMR for both levels of analysis: between and within. The SRMR at the within level was .03, and at the between level was .005. Both levels offer a good model fit according to SRMR as it is considered good between 0 and .08. Finally, the RMSEA test offers a result of .06, which is considered acceptable below .08.

Overall, it can be concluded that the multilevel structural model offered a good fit as all goodness of fit indices scored within their good or acceptable thresholds.

4.4.3. Results of hypothesis testing

The results of the multilevel structural equation modelling indicated that when the variables were modelled as per the hypothesised model, significant relationships were apparent. These relationships are presented in this section which has been structured by hypothesis. As the estimator used was Bayes, there is no p-value as there would be with other estimators. Instead, Mplus gives the 95% credible interval for the estimate. If this interval does not contain the null value, the estimate is found to be significant. Both the non-standardised and standardised model results are presented in Table 42 and Table 43, respectively. The table contains the within and between-level estimates along with the 95% credible interval of the estimate.

		95% Credible Interval		
	Estimate	Lower 2.5%	Upper 2.5%	Significance
Between-level (work team) effects				
$EC \rightarrow Slope$	-0.748	-1.476	-0.011	*
$WS \rightarrow EWB$	-0.367	-0.557	-0.161	*

Table 42. Between-level, and within-level effects between employee wellbeing, work stress, Leadermember exchange, psychosocial safety climate, and emotional contagion

$PSC \rightarrow EWB$	0.809	0.189	1.938	*
$LMX \rightarrow EWB$	-0.109	-0.984	0.402	
$PSC \rightarrow WS$	-1.557	-2.740	-0.667	*
$LMX \rightarrow WS$	0.838	0.072	1.992	*
$PSC \rightarrow LMX$	0.871	0.654	1.105	*
Within-level (em	ployee) effects			
$WS \rightarrow EWB$	-	-	-	
$PSC \rightarrow EWB$	0.124	0.069	0.181	*
$LMX \rightarrow EWB$	0.058	-0.011	0.125	
$PSC \rightarrow WS$	0.007	-0.058	0.073	
$LMX \rightarrow WS$	-0.076	-0.143	-0.009	*
$PSC \rightarrow LMX$	0.514	0.397	0.629	*

Note. N=237; 41 clusters. EC = emotional contagion. EWB = employee wellbeing. WS = work stress. PSC = psychosocial safety climate. LMX = Leader-member exchange.

Table 43. Between-level, and within-level standardised effects between employee wellbeing, work stress, Leader-member exchange, psychosocial safety climate, and emotional contagion

		95% Credible Interval		
	Estimate	Lower 2.5%	Upper 2.5%	Significance
Between-level (work team) effects				
$EC \rightarrow Slope$	-0.652	-0.958	-0.019	*
$WS \rightarrow EWB$	-0.411	-0.625	-0.176	*
$PSC \rightarrow EWB$	0.620	0.403	0.817	*
$LMX \rightarrow EWB$	0.097	-0.884	0.356	
$PSC \rightarrow WS$	-1.148	-1.852	-0.508	*
$LMX \rightarrow WS$	0.567	0.079	1.326	*
$PSC \rightarrow LMX$	0.871	0.721	0.946	*
Within-level (employee) effects				
$WS \rightarrow EWB$	-0.061	200	060	*
$PSC \rightarrow EWB$	0.300	0.175	0.419	*
$LMX \rightarrow EWB$	0.131	-0.018	0.285	
$PSC \rightarrow WS$	-0.017	-0.141	0.177	
$LMX \rightarrow WS$	-0.187	-0.343	-0.025	*
$PSC \rightarrow LMX$	0.529	0.411	0.622	*

Note. N=237; 41 clusters. EC = emotional contagion. EWB = employee wellbeing. WS = work stress. PSC = psychosocial safety climate. LMX = Leader-member exchange.

The hypothesised model with the standardised results can be found on Figure 10.



Figure 10. Hypothesised model with standardised parameter estimates

Note. N=237; 41 clusters. Multilevel research model with the between and within effects as well as the observed variables. The model contains the standardised estimators. * Significant path at the 0.05 level.

4.4.3.1. Hypothesis 1. Emotional contagion moderates the relationship between work stress and employee wellbeing

The first hypothesis of this research project aimed at testing whether there is a cross-level moderation of emotional contagion on the relationship between work stress and employee wellbeing. To test this moderation effect, we utilised random slopes in the regression between work stress and employee wellbeing. Emotional contagion is understood as a team-level variable, and it is hypothesised that this has an impact on the employee-level individual slopes.

At the between level, a significant effect was found linking emotional contagion with the slope of the relationship between work stress and employee wellbeing. The between-level non-standardised estimate of the direct effect of emotional contagion on the within-level slope between work stress and employee wellbeing was -.748, with a credible interval range from -1.476 and -.011. As the credible interval does not contain 0, the estimate is found to be significant. The

between-level standardised estimate was -.652, with a credible interval range from -.958 and -.019. This parameter estimate is also significant.

The significant estimate for the relationship between emotional contagion and the slope of the relationship between work stress and employee wellbeing, accepts the hypothesis that emotional contagion moderates the relationship between work stress and employee wellbeing. The moderation effect was plotted (see Figure 11). The results indicate than when emotional contagion is high, the negative relationship between work stress and employee wellbeing is strengthened. In effect, this means that employees who work in teams with higher emotional contagion, and low levels of work stress are likely to experience higher levels of employee wellbeing. In contrast, employees who work in teams with higher levels of employee wellbeing.



Figure 11. Moderation effect

Note. Emotional contagion strengthens the negative relationship between work stress and employee wellbeing.

4.4.3.2. Hypothesis 2. Work stress is negatively associated with employee wellbeing

The second hypothesis of this study proposes that there is a negative association between work stress and employee wellbeing. This hypothesis affects both the within level and the between level and suggests that when employees experience high levels of work stress, are likely to experience lower levels of employee wellbeing.

This hypothesis was accepted at both levels of analysis. At the between level, a significant effect was found with a non-standardised estimate of the negative relationship between work stress and employee wellbeing of -.367, with a credible interval range from -.557 and -.161. As the credible

interval does not contain 0, the estimate is found to be significant. The between-level standardised estimate was -.411, with a credible interval range from -.625 and -.176, which is also significant with a 95% confidence level. At the within level, the non-standardised significant effect cannot be reported since that relationship was randomised to report on the moderation effect. However, the within-level standardised estimates can be reported. The standardised parameter estimate was -.061, with a credible interval range from -.2 and -.06, which is also significant with a 95% confidence level.

4.4.3.3. Hypothesis 3. Psychosocial safety climate is positively associated with employee wellbeing

The third hypothesis of this study proposes that there is a positive association between psychosocial safety climate and employee wellbeing. This hypothesis affects both the within level and the between level and assumes that when employees perceive that their organisation and mangers prioritise their wellbeing above business outputs, employees will have a higher level of wellbeing.

This hypothesis was accepted at both levels of analysis. At the between level, a significant effect was found with a non-standardised estimate of the positive relationship between psychosocial safety climate and employee wellbeing of .809, with a credible interval range from .189 and 1.938. As the credible interval does not contain 0, the estimate is found to be significant. The between-level standardised estimate was .62, with a credible interval range from .403 and .817, which is also significant with a 95% confidence level. At the within level, a significant effect was found with a non-standardised estimate of the positive relationship between psychosocial safety climate and employee wellbeing of .124, with a credible interval range from .069 and .181. As the credible interval does not contain 0, the estimate is found to be significant. The within-level standardised estimate was .3, with a credible interval range from .175 and .419, which is also significant with a 95% confidence level.

4.4.3.4. Hypothesis 4. Psychosocial safety climate is negatively associated with work stress

The fourth hypothesis of this study proposes that there is a negative association between psychosocial safety climate and work stress. The hypothesis related to both the within level and the between level and assumes that when employees perceive that their organisation and mangers prioritise their wellbeing above business outputs, employees will have lower levels of work stress.

Hypothesis 4 was partially accepted. At the between level, a significant effect was found with a non-standardised estimate of the negative relationship between psychosocial safety climate work stress of -1.557, with a credible interval range from -2.74 and -.667. As the credible interval does

not contain 0, the estimate is found to be significant. The between-level standardised estimate was -1.148, with a credible interval range from -1.852 and -.508, which is also significant with a 95% confidence level. At the within level, this relationship was found non-significant at a 95% credible interval.

4.4.3.5. Hypothesis 5. Psychosocial safety climate is positively associated with Leadermember exchange

The fifth hypothesis of this study proposes that there is a positive association between psychosocial safety climate and Leader-member exchange. This hypothesis affects both the within level and the between level and assumes that when employees perceive that their organisation and mangers prioritise their wellbeing above business outputs, employees are more likely to perceive that their line manager is supporting them.

This hypothesis was accepted at both levels of analysis. At the between level, a significant effect was found with a non-standardised estimate of the positive relationship between psychosocial safety climate and Leader-member exchange of .871, with a credible interval range from .654 and 1.105. As the credible interval does not contain 0, the estimate is found to be significant. The between-level standardised estimate was .871, with a credible interval range from .721 and .946, which is also significant with a 95% confidence level. At the within level, a significant effect was found with a non-standardised estimate of the positive relationship between psychosocial safety climate and Leader-member exchange of .514, with a credible interval range from .397 and .629. As the credible interval does not contain 0, the estimate is found to be significant. The within-level standardised estimate was .529, with a credible interval range from .411 and .622, which is also significant with a 95% confidence level.

4.4.3.6. Hypothesis 6. Leader-member exchange is positively associated with employee wellbeing

The sixth hypothesis of this study proposes that there is a positive association between Leadermember exchange and employee wellbeing. The hypothesis related to both the within level and the between level and suggests that when employees perceive that their manager support them, employees are likely to experience higher levels of employee wellbeing

Hypothesis 6 was not accepted at any of the two levels with a 95% confidence level.

4.4.3.7. Hypothesis 7. Leader-member exchange is negatively associated with work stress

The seventh hypothesis of this study proposes that there is a negative association between Leadermember exchange and work stress. This hypothesis affects both the within level and the between level and assumes that when employees perceive their managers supports them, employees experience lower levels of work stress.

This hypothesis was accepted at both levels of analysis. At the between level, a significant effect was found with a non-standardised estimate of the negative relationship between Leader-member exchange and work stress of -.838, with a credible interval range from -1.992 and -.072. As the credible interval does not contain 0, the estimate is found to be significant. The between-level standardised estimate was -.567, with a credible interval range from -1.326 and -.079, which is also significant with a 95% confidence level. At the within level, a significant effect was found with a non-standardised estimate of the negative relationship between Leader-member exchange and work stress of -.076, with a credible interval range from -.143 and -.009. As the credible interval does not contain 0, the estimate is found to be significant. The within-level standardised estimate was -.187, with a credible interval range from -.343 and -.025, which is also significant with a 95% confidence level.

4.4.3.8. Hypothesis 8. Negative mediation between psychosocial safety climate and work stress through Leader-member exchange

The model also reflected a full mediation in the relationship between Leader-member exchange and employee wellbeing through work stress at the single level of analysis (within). This means that at the individual level, employees' perceptions of how their organisation prioritises their wellbeing is fully mediated through their perception on how their manager supports them.

4.4.3.9. Hypothesis 9. Positive mediation between Leader-member exchange and employee wellbeing through work stress

The model also reflected a full mediation in the relationship between Leader-member exchange and employee wellbeing through work stress at both levels of analysis (between and within). Specifically, the results indicated a full mediation effect, indicating that the impact of employees' perceptions of how their manager supports their wellbeing on their wellbeing is entirely mediated through their experience of work stress. This mediation effect was observed at both the betweenlevel (team-level) and within-level (individual-level) analyses. To provide further clarification, this means that when employees perceive a high level of support from their manager in terms of their wellbeing (as captured by the Leader-member exchange measure), it leads to lower levels of work stress. In turn, reduced work stress is associated with higher levels of employee wellbeing. In other words, employees' perceptions of managerial support play a significant role in shaping their experience of work stress. When employees feel supported by their manager, it contributes to a lower level of stress in their work environment. This reduced work stress, in turn, has a positive impact on their overall wellbeing.

4.4.3.10. Hypothesis 10. Positive mediation between managerial priority and employee wellbeing through Leader-member exchange and work stress

The model also reflected a double mediation in the relationship between psychosocial safety climate and employee wellbeing. First, the relationship between psychosocial safety climate and work stress is fully mediated by Leader-member exchange at the within level. This means that employees' perceptions of how their organisation prioritises their wellbeing is fully mediated through employees' perceptions of managerial support. At the between level, this is only partially mediated since the direct relationship between psychosocial safety climate and work stress is significant. Second, the relationship between Leader-member exchange and employee wellbeing is fully mediated by work stress at both levels of analysis (between and within). This means that employees' perceptions of managerial support are fully mediated through their self-reported work stress level.

4.5. Conclusion of the chapter

This chapter has presented the results of the statistical analysis of the collected data. The chapter commenced with a presentation of several descriptive statistics to understand the sample dataset. The demographic statistics are minimum, maximum, mean and standard deviation for the variables included in the hypothesised model: employee wellbeing, work stress, psychosocial safety climate, Leader-member exchange, and emotional contagion. Then, the chapter presented the sample demographics including gender, tenure, and year of birth. The section following presented the results of the measurement model, including multivariate normality test, intraclass correlations, results from the multilevel exploratory factor analysis, and multilevel confirmatory factor analysis. The multilevel confirmatory factor analysis included the observed intraclass correlation, the latent variable model fit, its factor loadings, and the McDonald's omega reliability index for each variable along with the discriminant validity and correlation analysis amongst all variables.

The chapter concluded with the results of the multilevel structural equation model. This section began with a presentation of the estimator used: Bayes, and then the model fit of the structural model was presented. This chapter finished with the results of the hypothesis testing and a detailed presentation of each significant relationship.

Chapter 5. General Discussion and Implications

Chapter 4 has presented the results from the multilevel quantitative analysis in response to the research questions: *RQ1*. *Does emotional contagion among team members influence the levels of individual work stress and employee wellbeing?* and *RQ2*. *To what extend do organisational resources impact the level of employee wellbeing?*

The results from the statistical analysis evidence the impact of organisational resources and work teams on employee wellbeing. This chapter reflects on the results presented in Chapter 4 and discusses how these results contribute to the current body of scholarly knowledge, policy and to business practice. The first section presents an overview of the research questions and the hypotheses that were used to answer the research questions. The second section explores the theoretical implications of the results to understand how employees influence each other's wellbeing when working together and how the organisational resources impact the level of employee wellbeing. The third and final section develops the practical and policy implications of this thesis and how it contributes to practice.

5.1. Overview of the research questions and hypotheses

The main aim of this research project is to identify the influence of co-workers, and organisational resources, on employee wellbeing. This goal has the objective of making wellbeing more accessible for those organisations that try to support the wellbeing of their employees. Additionally, at least in Australia, the need for organisations to support employee wellbeing and reduce psychosocial risk is laid out in the framework published by Safe Work Australia in July 2022. The Work Health and Safety Act makes it mandatory for organisations to eliminate or, when this is not possible, minimise all hazards that pose a psychosocial risk (Safe Work Australia, 2022). Therefore, it is necessary to know how organisations can effectively address employee wellbeing and eliminate or reduce psychosocial hazards.

Until now, wellbeing, and more specifically, employee wellbeing, have been mainly studied from the individual perspective and considering employees as isolated units within a team. This individual analysis of employee wellbeing only offers a partial perspective and neglects the team influence - as if those teams had no impact on the individual. However, the academic literature provides extensive evidence that as humans are social animals, the relationships and interactions between humans have an impact on the way we operate, think, behave, and express (Robinson et al., 2019). By analysing employee wellbeing through an individual lens, organisations and policy makers have been implementing interventions targeting the individual employee. As such, to date there has been insufficient analysis of the impact of those employees who are not captured by those individual interventions on other employees. There is no research on the emotional contagion among team members on employee wellbeing and work stress and the impact of a distressed employee on others. Those distressed employees could be reducing, or even eliminating the positive impact of those individual interventions and dragging their colleagues' wellbeing down. Moreover, many organisations do not even know how to provide resources that support employees' wellbeing or if they do, they do not feel that these resources fully support wellbeing (Volini et al., 2021). Within this context, this research project has the main aim of contributing to the current knowledge on employee wellbeing and studying a collective conceptualisation of employee wellbeing and its "contagiousness" between team members.

To operationalise the aim of this thesis, two research questions have been posited (see Table 44). The first research question sought to analyse the emotional contagion among team members on work stress and employee wellbeing. The second research question studies the association between organisational resources and their impact on employee wellbeing. In combination, the two research questions build on previous research to offer an individual, collective, and organisational conceptualisation of employee wellbeing. The following section provides a discussion of each of these questions and their corresponding hypothesis.

Table 44. Research questions

RQ1	Does emotional contagion among team members influence the levels of individual work
	stress and employee wellbeing?
RQ2	To what extent do organisational resources impact the level of employee wellbeing?

5.1.1. Does emotional contagion among team members influence the levels of individual work stress and employee wellbeing?

The main aim of this study was to make employee wellbeing more accessible for policy makers and business practitioners. Since most research on employee wellbeing has examined it from the individual perspective, this research project sought to break with tradition to conceptualise a collective influence on employee wellbeing. Using a multilevel statistical lens, the first research question sought to understand the emotional contagion among co-workers on employee wellbeing and work stress. This influence was studied using two important relationships. First, the relationship between work stress and employee wellbeing was studied. The impact of emotional contagion on this relationship was then examined. This research question had two corresponding hypotheses, one for each relationship that was under examination. These hypotheses and the discussion of the findings associated with them are explored in the below two sections.

5.1.1.1. Work stress and employee wellbeing

The literature suggested a negative relationship between work stress and employee wellbeing at both levels of analysis. At the individual level, employees experiencing high levels of work stress

are likely to experience lower levels of employee wellbeing. In contrast, employees whose levels of work stress are under control are likely to experience higher levels of wellbeing. This relationship is also assumed at the team level, where teams with high levels of stress are likely to experience lower levels of collective wellbeing, and vice versa. As such, interventions aiming to reduce work stress would also support employee wellbeing. Csikszentmihalyi (1991) suggests that work stress is caused when employees do not have the necessary skills for the work challenge that they are facing. This dichotomy could be addressed from two different fronts. On one hand, interventions that look at improving the skill competencies of employees would be likely to reduce the levels of work stress and, therefore, improve their wellbeing levels (Brassington & Lomas, 2021). On the other hand, from a task management perspective, tasks assigned to an employee, or a team should fit their skill level so that they can achieve the flow level described by Csikszentmihalyi (1991). A flow level with a healthy and appropriate level of work stress is achieved when the challenge level is high, but it also matches the high level of skills for that employee or team.

The negative relationship between work stress and employee wellbeing was expressed in Hypothesis 2, where it was hypothesised that work stress is negatively associated with employee wellbeing. The results from the statistical analysis confirmed the expected negative association between work stress and employee wellbeing at both levels of analysis (see Chapter 4, section 4.4.3.2). As such, it is argued that employees who are exposed to high levels of work stress are likely to experience lower levels of wellbeing. In particular, organisations should focus on reducing or eliminating the levels of structural work stress that are not due to a particular moment in time or a specific project but intrinsic to an individual and their role. Those structural levels of work stress, in the medium term, have a very negative impact on employee wellbeing (Davies, 2021). Therefore, because organisations have a responsibility to reduce (to the maximum possible extent) the impact of psychosocial hazards, it is the organisation's responsibility to ensure that there are strategies in place to minimise the levels of structural work stress. Managing psychosocial hazards, according to Parker and Jorritsma (2021), can be achieved by fitting the tasks with the appropriate skill level of the worker and designing work to be mentally healthy. When this alignment exists, employees will be able to draw from organisational resources to protect their level of work stress and consequently, their wellbeing.

At the team level, there are particular positions or business areas that are more likely to experience higher levels of work stress (Davies, 2021). Such teams are likely to experience lower levels of wellbeing and therefore require higher levels of support from organisations and policy makers. This thesis found that the negative relationship between work stress and employee wellbeing is also significant at the team level. This has implications for practitioners – in particular, for

professional services, as those teams could be exposed to higher psychosocial risks and, therefore, require prompt action to reduce their levels of stress and mitigate the impact of stress on wellbeing.

According to conservation of resources theory, employees will strive to maintain, protect or acquire resources that will reduce their stress levels (Halbesleben et al., 2014). In particular, when an employee faces a resource draining event where they are likely to experience high levels of work stress, they are likely to use their current resources to protect themselves and experience lower levels of work stress. A resource draining event could be any circumstance where they are required to work above their skill level or their abilities. As such, minimising resource draining events is essential to guarantee healthy skill levels at both the individual and the team level.

5.1.1.2. The team influence on employee wellbeing

The statistical analysis found a negative relationship between work stress and employee wellbeing at both levels of analysis. Moreover, it is also theorised that the work team (between level) has an influence in such a relationship at the employee level (within level). In other words, emotional contagion among co-workers has an influence on the individual relationship between work stress and employee wellbeing. This influence is covered in Hypothesis 1, where it is hypothesised that emotional contagion moderates the relationship between work stress and employee wellbeing. This hypothesis aims to address the first research question of this thesis by testing if the emotional contagion between co-workers influences the individual relationship between work stress and employee wellbeing. For example, employees within teams with high levels of emotional contagion and with high levels of work stress are likely to have even lower levels of individual wellbeing due to the contagious effect of those co-workers who are also experiencing lower levels of work stress. In contrast, employees nested within teams with high levels of emotional contagion and with low levels of work stress may experience even higher levels of individual wellbeing as they would 'catch' the wellbeing of those around them.

The results from the statistical analysis confirmed the expected cross-level moderation of collective emotional contagion in the individual relationship between work stress and employee wellbeing (see Chapter 4, section 4.4.3.1). The significance of this moderation influence of the team's emotions on the relationship between work stress and employee wellbeing offers important theoretical and practical implications. Specifically, these results highlight the importance of the emotional contagion among team members on wellbeing and the influence that co-workers have on work stress and employee wellbeing. While the previous section mentioned the need for workplace interventions that aim to reduce the levels of work stress to improve wellbeing levels,

it is essential to consider the team influence on such relationship to ensure that those interventions achieve their full potential.

According to social exchange theory, there is a reciprocal dynamic between two or more employees (Eisenberger et al., 1987). When one employee gives, their co-workers would normally receive and give back. As such, from a positive action, the receiver is likely to also respond in a positive way, which could trigger a loop of positive actions. On the other hand, if there is a negative action or the receiver perceives it as negative, colleagues are likely to also respond in a negative way which could become a loop of negative actions within a work team. This dynamic explains the influence of the team on workplace behaviours. However, it misses an important factor. A practical contribution of this thesis is that employees not only respond to others' actions and behaviours, according to emotional contagion theory, employees with high levels of emotional contagion tend to 'catch' colleagues' emotions and synchronise with those. As such, ensuring that there is a positive environment within a work team is essential for its success.

Person-team fit theory outlines key factors that are necessary for team selection to maximise team performance (Ellis et al., 2022). Intrinsically, human resources and hiring managers have considered the team culture and ensure that new members "fit" or "add" to the team. The concept of cultural add and fit aligns with the new trends in diversity and inclusion (Liera, 2020) and tries to promote the idea of supporting the team or at least not dragging it down. If employees are likely to synchronise emotions and moods with those around them, it is important to guarantee a good environment for each individual member of a team. The need to find employees that add to the culture of the team has important limitations. With the current skill shortage in the Australian labour market, it appears that a number of organisations across all industries and sectors are even offering positions on the first interview if the interviewee meets the legal requirements for the position (Crouch, 2022). Having compliance requirements does not guarantee that the candidate will add to the team, which makes it even more urgent to address the current skill shortage and "the Great Resignation" that is currently affecting the Australian economy (Serenko, 2022). However, the findings from this research go beyond this theory. The results highlight that the solution does not uniquely fit within the hiring process, but instead it is a day-to-day maintenance of team wellbeing.

The results point out that it is the responsibility of managers to strive for consensus in team stress and wellbeing – to date, frameworks (such as employee engagement, and Leader-member exchange) take it for granted that there will be an "in-group" and "out-group" (Mejia et al., 2021). This study questions this assumption. With a significant cross-level moderation of emotional contagion on the individual relationship between work stress and employee wellbeing, if there is an "out-group", the whole team's wellbeing is at risk. This also questions the way modern management searches for the most amount of happiness for the most amount of people. A group of employees that work together form a team, and if that team is well-maintained, it is stronger than the sum of individuals in the team. But, if something, i.e., stress, perforates the alloy, the degradation will spread across the team, reducing their individual and collective levels of wellbeing.

The results from this thesis prompt actions for Governments, and regulators and a stakeholder with a vested interest in safe and productive workplaces. First, it would be necessary to work closely with organisations and sectors affected by the skill shortage to understand their talent needs. Second, there is a role for regulators in supporting public, private and non-for-profit sector organisations to build work teams which are more conductive to collective wellbeing. The Government should apply the talent needs to educational institutions at secondary, vocational, and higher education levels so that the curriculum meets the talent requirements. Regulators can also generate resources, guides and promote further research to support organisations in implementing the day-to-day maintenance of team wellbeing.

5.1.2. To what extent do organisational resources impact the level of employee wellbeing?

The second research question sought to understand how organisational resources impact the employee wellbeing level. This influence was studied using two organisational resources. First, managerial priority was captured by psychosocial safety climate. Second, managerial support was conceptualised using Leader-member exchange. These two organisational resources are theorised as influencers on employee wellbeing through five different hypotheses. The first hypothesis (H3) tested the relationship between managerial priority and employee wellbeing. The second hypothesis (H4) tested the relationship between managerial priority and work stress. This relationship also proposed a mediated influence of managerial priority on employee wellbeing through work stress. The third hypothesis (H5) tested the relationship between managerial priority and support. The fourth hypothesis (H6) tested the relationship between leader-member exchange and employee wellbeing, which also implies a mediated relationship between managerial priority and employee wellbeing though leader-member exchange . The fifth and last hypothesis (H7)tested the relationship between leader-member exchange and work stress. This relationship also proposed a mediated relationship between leader-member exchange and employee wellbeing through work stress. Overall, there was a double or chain mediation between managerial priority and employee wellbeing though leader-member exchange and work stress. These hypotheses and the discussion of their findings are explored below.

5.1.2.1. Managerial priority and employee wellbeing

In addressing Hypothesis 3, the present data indicated a positive relationship between psychosocial safety climate and employee wellbeing at both levels of analysis. Managerial priority was conceptualised by psychosocial safety climate, and it serves as a high-level organisational resource. At the employee level, when employees perceive that their managers prioritise their wellbeing, they are likely to experience higher levels of employee wellbeing. This positive impact is due to managerial priority being an essential resource for employees, according to conservation of resources theory. In contrast, employees who perceive lower levels of managerial priority within their organisation are likely to experience low levels of employee wellbeing. At the team level, when co-workers perceive that their organisation is prioritising the wellbeing of the team above productivity and other outputs, there are likely to also experience higher levels of collective wellbeing. Similarly, when the team perceive low levels of psychosocial safety climate, they are also likely to experience lower levels of wellbeing. The results from the statistical analysis confirmed the expected positive relationship between psychosocial safety climate and employee wellbeing at both levels of analysis (see Chapter 4, section 4.4.3.3).

This finding is particularly important as it suggests a new approach that organisations can adopt to manage wellbeing – through the provision of the resources of managerial priority and support. According to conservation of resources theory, managerial priority is an organisational resource used by employees to reduce their levels of work stress and boost their wellbeing (Dollard & Bakker, 2010). Managerial priority and support can minimise certain psychosocial hazards and act as resources to balance the demands created by others.

5.1.2.2. Managerial priority and work stress

Hypothesis 4 tested the negative relationship between psychosocial safety climate and work stress at both levels of analysis. As psychosocial safety climate serves as a high-level organisational resource, it is expected that at the individual level, employees perceiving that their organisation prioritises their wellbeing, are likely to experience lower levels of work stress. In contrast, employees who perceive low levels of managerial priority would be likely to experience higher levels of work stress. However, this hypothesised relationship was not supported (see Chapter 4, section, 4.4.3.4). According to conservation of resources theory, employees are likely to do anything to maintain, preserve or acquire resources to protect their levels of wellbeing (Hobfoll, 2001). However, Salanova et al. (2010) argue that high-level organisational resources require some means of transport from the organisation to the employee. Without such transport, those organisational resources may not be able to reach their final destination, the employee. On the other hand, it is also theorised that at the team level, work teams who perceive that their organisation is prioritising their wellbeing above productivity are likely to experience lower levels of work stress. In contrast, if the team perceives low levels of prioritisation by their organisation, the team members are likely to experience higher levels of work stress. The analysis supported this negative relationship at the team level (see Chapter 4, section, 4.4.3.4). According to social exchange theory, teams with high levels of psychosocial safety climate are likely to respond in a more positive way towards their organisation. As such, several research projects have linked higher levels of psychosocial safety climate with the optimal level of work stress to perform at the highest level (Farr-Wharton et al., 2022a). In contrast, teams who perceive that their organisation is not prioritising them above profitability are likely to experience higher levels of work stress, which leads to lower levels of employee wellbeing as seen in previous sections.

While, according to social exchange theory, both individuals and teams with high levels of psychosocial safety climate are likely to respond in a more positive way towards their organisation, this statement was only supported at the team level in this thesis. Clearly, this is a story about the key role of the direct manager's support as the vector through with managerial priority or psychosocial safety climate acts. This has been demonstrated through the Leader-member exchange mediation between psychosocial safety climate and work stress through Leader-member exchange and which is elaborated later in this chapter.

5.1.2.3. Managerial priority and managerial support

Conservation of resources theory suggests that managers are the social mechanism by which employees receive organisational resources as support (Salanova et al., 2010). This key role of managers as the vector between organisational resources and the employee is a key contribution of this thesis. As such, it is theorised that there is a positive relationship between psychosocial safety climate and Leader-member exchange at both levels of analysis (Hypothesis 5). The results from the statistical analysis confirmed the expected relationship between Leader-member exchange and work stress at both levels of analysis (see Chapter 4, section 4.4.3.5).

At the individual level, employees who perceive that their manager is supporting their wellbeing, are likely to also perceive that their organisation is prioritising their health by using the manager as a driver of organisational resources. Some even argue that regardless of how much the organisation is prioritising the wellbeing of employees, unless their manager supports their mental and physical health, they are not likely to perceive that their organisation is prioritising the (Salanova et al., 2010). At the team level, teams are also likely to perceive that their organisation is prioritising their wellbeing if their manager is supporting their collective wellbeing. The findings on both levels of analysis are a key contribution of this thesis as, no matter how much
effort the human resources department or senior management put in developing policies and practices to support their employees' wellbeing, employees will not feel supported unless they have a supporting line manager.

5.1.2.4. Managerial support and employee wellbeing

This project tested the relationship between leader-member exchange and employee wellbeing. It was theorised that there is a positive relationship between the perception of employees who believe that their manager supports their role and their wellbeing, and employee wellbeing. Additionally, such a relationship was also theorised at the team level, where teams that perceive high levels of leader-member exchange would be likely to experience higher levels of employee wellbeing. However, in addressing Hypothesis 6, the present data indicates no significant relationship between leader-member exchange and employee wellbeing at any of the two levels of analysis (see section 4.4.3.6). Instead, the influence of leader-member exchange on employee wellbeing operates indirectly through work stress as presented in the following section.

According to conservation of resources theory, employees and work teams are likely to do anything to protect their resources to protect themselves from high stress levels (Xerri et al., 2022). And as showed earlier, work stress is negatively related to employee wellbeing. Therefore, as previously mentioned, this thesis argues that leader-member exchange acts as an organisational agent to take organisational resources to employees and work teams and reduce their stress levels. Employees and teams get support from their line manager to draw upon organisational resources and reduce their levels of work stress, which ultimately, positively impacts their level of employee wellbeing. Therefore, the role of line managers in supporting the wellbeing of their employees must be a key consideration for organisations in eliminating psychosocial hazards and having a mentally healthy job.

5.1.2.5. Managerial support and work stress

In addressing Hypothesis 7, the present data indicates a negative relationship between Leadermember exchange and work stress at both levels of analysis. Managerial support is conceptualised by Leader-member exchange, and it serves as a vector of organisational resources from the organisation to the employee. At the employee level, when employees perceive that their manager supports their wellbeing, they are likely to experience lower levels of work stress. This impact is due to leader-member exchange being an essential resource for employees to draw upon organisational resources, according to conservation of resources theory. In contrast, employees who perceive lower levels of leader-member exchange are likely to experience high levels of work stress. At the team level, when co-workers perceive that their manager is supporting the wellbeing of the team, they are likely to also experience lower levels of collective stress. Similarly, when the team perceive low levels of leader-member exchange, they are not able to use organisational resources and are likely to experience higher levels of stress. The results from the statistical analysis confirmed the expected relationship between Leader-member exchange and work stress at both levels of analysis (see Chapter 4, section 4.4.3.7).

5.2. Contributions and implications for theory and research

This thesis moves knowledge in the field forward by providing a more nuanced perspective on how organisational resources and co-workers' emotional contagion contribute to employee wellbeing. Within this advancement of the scholarly knowledge, this thesis provides three key points. (1) By theoretically blending social exchange, emotional contagion, and conservation of resources theories, this study offers a unique approach to understanding the dynamics behind employee wellbeing. The complexity of this research project has presented the need to use theoretical blending (Cornelissen & Durand, 2012) between three theoretical lenses to understand the dynamics behind employee wellbeing. To this end, this multilevel statistical analysis has been able to account for the team influence on employee wellbeing and the team dynamics of wellbeing at work. This implies that the paradigm typically associated with the influencers of employee wellbeing and its individual responsibility may have been challenged and associated with the environment in which the employee wellbeing is also a collective obligation as the team has an important influence on the individual levels of wellbeing. In this way, supportive organisational resources and a positive team are important contributors to wellbeing at work.

(2) This research also challenges the paradigm typically associated with the influencers of employee wellbeing and its individual responsibility, demonstrating that employee wellbeing is also a collective obligation as the team has an important influence on the individual levels of wellbeing. This implies that supportive organisational resources and a positive team are important contributors to wellbeing at work. (3) Furthermore, this study provides a methodologically sound process for demonstrating how a team that cares and supports, with high contagion, is a team with higher employee wellbeing and lower work stress. This dynamic challenges the relationship between employee wellbeing and organisational resources, as they are presented within Hobfoll's conservation of resources theory. From a theoretical standpoint, your results support the notion that team emotional contagion moderates the individual-level relationship between work stress and employee wellbeing. Therefore, this research project contributes to the field of employee wellbeing by highlighting the importance of accounting for the team impact on organisational practices and expanding emotional contagion theory.

The findings contribute to improving and extending the adopted theories in several ways. Firstly, the study blends social exchange, emotional contagion, and conservation of resources theories to

provide a more nuanced perspective on how organisational resources and co-workers contribute to employee wellbeing. This theoretical blending enables an understanding of how team contagion for wellbeing and the key role of the line manager as the transporter of organisational resources can impact employee wellbeing. Secondly, the study challenges the paradigm typically associated with the influencers of employee wellbeing and its individual responsibility, demonstrating that employee wellbeing is also a collective obligation as the team has an important influence on the individual levels of wellbeing. This implies that supportive organisational resources and a positive team are important contributors to wellbeing at work. Thirdly, the study provides a methodologically sound process for demonstrating how a team that cares and supports, with high contagion, is a team with higher employee wellbeing and lower work stress. Fourthly, the study supports the notion that team emotional contagion moderates the individual-level relationship between work stress and employee wellbeing. Finally, the study highlights the importance of accounting for the team impact on organisational practices and expanding emotional contagion theory.

5.3. Contributions and implications for practice and policy

This thesis contains several findings around the key role of the line manager as the vector of organisational resources and the collective contagion of wellbeing. As such, several contributions and implications for professional service practitioners and policy-makers are presented in this section. First, this thesis highlights the important role of line managers as transporters of organisational resources from the organisation to the employees. These line managers are also key in maintaining the day-to-day wellbeing of the team and not assuming that there will always be an "in-group" and an "out-group". Second, this thesis suggests that both practitioners and policy makers should add to their individual interventions collective or team-designed projects to further support their employees' levels of wellbeing. This section further develops these two contributions and points out their implications from a practical standpoint.

This thesis highlights the important role organisational support as a resource for employees to influence their level of wellbeing. The significant team influence on strengthening the relationship between work stress and employee wellbeing poses important implications for both business practitioners and policy makers. Organisations need to ensure that their practices to reduce work stress and support employee wellbeing are not only based on the unique individual, but also on the team, while maintaining wellbeing consensus. It has been found that for those teams with high levels of emotional contagion, the influence of the individual work stress is very high. Therefore, distressed employees could reduce the impact of any wellbeing interventions, causing important losses for the organisation. These implications are not only financial losses, but they also carry a huge social cost and decline in mental health. As seen, the economic cost of poor mental health

in Australia has been estimated at between AUD\$200 and AUD\$220 billion a year according to the Australian Productivity Commission (King, 2021). This thesis suggests that both professional service practitioners and policy makers should move from individual interventions to collective or team-designed projects to further support their employees' levels of wellbeing.

A lot can be said about primary interventions for wellbeing versus other non-primary interventions. Primary interventions include those that address the psychosocial hazard directly, usually at the organisational level using psychosocial safety climate and appropriate work design. On the other hand, secondary or tertiary interventions act after the employee has been exposed to a psychosocial risk and aim at reducing their level of stress and negative impact on their wellbeing. This thesis points to the fact that the collective wellbeing has a role and may indeed be very important. Future research is necessary to establish how important it is and how beneficial could be for organisations to implement collective interventions to support the team's wellbeing.

These organisational resources require the support from line managers who act as transporters of organisational resources from the organisation to the employees. The role of line managers is particularly important in the role of organisations in eliminating psychosocial hazards at work. Yet, most line managers are not human resources professionals and may have not received the appropriate training or may not have the necessary soft skills to be a supportive manager. Without an appropriate level of managerial support, employees may not perceive that their organisation is prioritising them, which could have important negative implications for the wellbeing of the employees. Following the narrative about the crucial role of managers, this thesis suggests that there is a need to change the management style from a majority to a general consensus. By doing so, managers could be able to avoid the negative impact of the "out-group" in terms of stress or wellbeing, on the team's wellbeing. For all these reasons, this study highlights the importance of ensuring that managers are also trained and evaluated on non-productivity skills. Additionally, organisations should pay particular attention to those teams that are exposed to inevitable psychosocial hazards as those require prompt action to reduce their levels of stress, because, as this thesis has pointed out, team emotional contagion impacts the relationship between work stress and employee wellbeing. Finally, team interventions may be a mechanism that allow organisations to support the wellbeing of their employees and account for the contagious effect of employee wellbeing.

For policy makers, these findings also have significant implications. First, workplace regulations that seek to protect the wellbeing of employees should not solely focus on the individual but instead on the team. Most legislation recommends or even requires primary interventions as a priority to eliminate psychosocial hazards before they impact on work stress. However,

organisations find it easier and more cost-effective to apply individual and non-primary approaches. Yet, this study shows that organisational resources are fundamental and that the interventions should take place at the team level due to the contagious nature of employee wellbeing. The team has important implications for employee wellbeing and as such, regulations on team interventions should be a mechanism to support it even further. Second, the need to hire employees who add to the work team poses an important pressure due to the current skill shortage and the "War for Talent" (Goldstone et al., 2021; Serenko, 2022). As such, policy makers should prioritise policies that regulate the current skill shortage not only in the short term but also in the medium and long term. For instance, policies should focus on moving to a model of training leadership which maintains consensus wellbeing, and not majority wellbeing. This thesis can inform wellbeing interventions as well as wellbeing interventions research by incorporating a more nuanced conceptualisation of collective wellbeing and its antecedents.

Finally, all these implications pose important consequences for the future of work. While this may be stepping a little beyond the scope of this project, it is important to note that our work environments keep evolving and changing constantly. As such, continuous research is needed to understand the impact of new psychosocial hazards (e.g., the relationship between employees and artificial intelligence, remote teams, and so on) and how organisations can address them.

5.4. Conclusion of the chapter

This chapter has discussed the findings of the study along with its theoretical, methodological, practical and policy contributions and implications. To sum up, this thesis has made four important contributions. First, there is the theoretical blending of social exchange theory, emotional contagion theory and conservation of resources theory which has enabled the understanding of how organisational resources and co-workers contribute to employee wellbeing. Second, this thesis has used a new methodological approach to study the antecedents of employee wellbeing and its collectiveness by using multilevel structural equation modelling. Third, this thesis highlights the important role of line managers as transporters of organisational resources from the organisation to the employees. The role of line managers is particularly important considering that in general, they are not human resources professionals, and they get promoted on their technical skills but once promoted, they are required to have strong interpersonal skills to be able to support the wellbeing of their employees. And fourth, this thesis suggests that both practitioners and policy makers should add to the individual interventions, collective or teamdesigned projects to further support their employees' levels of wellbeing. In doing so, organisations should leave the idea that there will always be an "out-group" and focus on moving to a model of training leadership which maintains consensus wellbeing, and not majority wellbeing

Chapter 6. Conclusions

This thesis has studied the role of organisational resources and teams on employee wellbeing through a multilevel lens. The aim of this final chapter is to conclude the research project by offering a brief recapitulation the main points of this study. First, it offers a summary of the main findings, and it continues with the limitations of this research project and recommendations for future research.

6.1. Important findings

This thesis contributes to the body of scholarly knowledge associated with employee wellbeing. Specifically, this research project studied the influence of organisational resources and coworkers on employee wellbeing using a multilevel lens. The study aimed to develop a more nuanced understanding concerning the collective drivers of wellbeing. This thesis sought to move the current discourse beyond its almost exclusive focus on individual mental disposition. Moreover, the thesis also aimed to assist in developing workplace interventions that focus on the maintenance and acquisition of better levels of collective wellbeing. An improved understanding of employee wellbeing and its antecedents contributes to the development of robust workplace interventions that aim at reducing the current rise of work-related stress, anxiety, and depression.

This project has been carried out during the COVID-19 pandemic, which has caused many workplace changes and an increased level of work stress and psychological distress. All data was collected in the professional services industry as employees within this industry kept working in teams during the Australian lockdowns. Moreover, the current "War for Talent", with the increasing competition between recruiters and the prominent skill shortage that the Australian economy is currently facing, adds extra pressures for managers and organisations (Goldstone et al., 2021; Serenko, 2022). Given these pressures, some organisations try to create a competitive advantage and be at the forefront of the employee value proposition. This thesis has proposed several practical contributions that can assist supporting employee wellbeing and reducing work stress. On the theoretical side, the project offers a nuanced theoretical insight on employee wellbeing research. The study uses the theoretical blending of three different and complementary theories (social exchange theory, emotional contagion theory, and conservation of resources theory) to integrate plausible reasoning and provide new insights while avoiding silos of knowledge (Cornelissen & Durand, 2012). Moreover, it proposes an advanced quantitative methodological approach to employee wellbeing research by using multilevel structural equaition modelling.

Psychosocial safety climate as managerial priority and Leader-member exchange as managerial support served as organisational resources to understand how these contribute to reducing work stress and to improving employees' wellbeing. Moreover, the vital importance of the line managers has been proven. The literature argued that there is an important disconnection between the technical and interpersonal skills of line managers. These managers serve as organisational agents who transport the organisation's resources from the organisation to the employees. Therefore, when the link between the line manager and the employee and/or the team is broken, those organisational resources do not fully arrive at their destination (the employee), and do not serve their purpose (to reduce the employee level of stress and support their level of wellbeing). Since line managers do not necessarily have human resources training (as documented in the literature), it is important for the human resources to serve as organisation to ensure that they have the necessary skills and resources to serve as organisational agents.

Emotional contagion between co-workers served as a moderator in the negative relationship between work stress and employee wellbeing. The significant moderation of emotional contagion between co-workers on the relationship between work stress and employee wellbeing served to answer the first research question. This result asserts the contagious effect between co-workers and the collective conceptualisation of employee wellbeing. This implies that there is an urgent need for practices and policies that incorporate the idea of collective wellbeing instead of exclusively considering it an individual and non-transferable concept. The Australian regulation makes it mandatory for organisations to eliminate or, if not possible, reduce workplace hazards that present a psychosocial risk for employees and other stakeholders. If organisations and policy makers wish to comply with this regulation, it is necessary to design interventions and roles/positions that consider wellbeing as a collective variable which is impacted by the team in which the employee works.

Finally, this thesis also highlights the need to shift the current management design from a majority approach to a consensus approach. Until now, managers are trained to maintain a majority approach in aspects such as employee wellbeing. If the majority presents good or acceptable levels of wellbeing, it is a sufficient effort from a managerial perspective. However, due to the significant contagious effect that this thesis highlights, these managerial practices should shift to a consensus approach where nobody is left as an "out-group". Current practices take for granted that there will always be an "out-group" and an "in-group". Yet, as this thesis has found, there is a significant moderation impact of team emotional contagion on the individual level relationship between work stress and employee wellbeing. If managers assume and accepts that there will always be an "out-group", those individuals within that group could drag their co-workers' levels

of wellbeing. Further research is necessary to understand the degree of the impact of these employees with negative work experiences on their co-workers.

6.2. Limitations and recommendations for future research

As in all research projects, this study is not free from limitations. An important limitation of this study is the number of teams that participated in the project. As discussed on Chapter 3, while there are no concerns with the validity of the results of this study, it is worth noting that future research could benefit from using a larger sample. In particular, a larger sample of work teams would have been useful for the two-level data analysis given the current limitations of statistical packages that test for multilevel analysis using cross-level moderation and use Bayes estimator. For this particular project, getting a larger sample was impossible due to several factors. First, the engagement strategy and data collection processes were done with no funds. Second, the data had to be collected during several COVID-19 outbreaks that impacted Australian society, causing many participants to drop out of the study. As the concepts measured rely on employee perceptions and these are subject to the current circumstances, the employees' mood at the time of their participation could have an impact on their responses. COVID-19 has had an important impact on mental health, and therefore, collecting data at another time would be interesting and serve as a comparison with this study. Third, there was a limited and specific amount of time to collect data. To combat this limitation, subsequent studies could look at introducing several waves of data collection. A time-lagged approach would further validate the results of this thesis and offer a more dynamic and representative picture of the influence of teams and organisational resources on employee wellbeing.

Further research should explore other potential levels of analysis such as organisational influence, a comparison between different industries or even a cross-cultural analysis. Moreover, further research should drill down deeper, possibly through qualitative approaches, to understand more about how wellbeing contagion works within work teams. Other studies could look at how line managers serve as transporters of organisational resources to enhance the impacts of psychosocial safety climate in employee and collective wellbeing. Future quantitative research might usefully look at some of those things, and perhaps expand the nested multilevel analysis to include organisational, team and individual level in the same model. And last but not least, other studies should look at new psychosocial hazards that appear as the work environment evolves over time. These include but are not limited to the impact of virtual or flexible work groups or the relationship between intelligent robots and employees.

Multilevel statistical analysis is an emerging analysis paradigm, and it presents important limitations in the transferability while building highly complex models which offer nuance.

Kozlowski and Klein (2000) considered multilevel organisational theory within the pragmatic research philosophy. They argue that while it is a quantitative research methodology, it was born within a reality that is in constant debate and interpreted as new situations arise, which falls under the pragmatism ontology. Moreover, according to Kozlowski and Klein (2000), when building a multilevel model from a literature review, values play a large role as researcher adopts both objective and subjective points of view. Further research is needed in this area to keep developing the multilevel organisational theory and multilevel statistical methods.

6.3. Conclusion of the chapter

This study has investigated the impact of organisational resources and co-workers on employee wellbeing through a multilevel lens. The findings conclude that team emotional contagion influences the individual level relationship between work stress and employee wellbeing. In this sense, managers play a crucial role in incorporating consensus wellbeing practices and avoid assuming that there will always be an "in-group" and an "out-group". In addition, organisational resources play a key role for employees to draw upon to support their levels of wellbeing. These organisational resources only get to the employees that they target though line managers. As such, the role of line managers is essential for organisations to support the wellbeing of their employees. This chapter has offered a review of the important findings of this thesis as well as the limitations and recommendations for further research.

References

- Abraham, S. (2011). Acquisition and allocation of human, financial, and physical resources in the health care system. *The Health Care Manager*, *30*(1), 38-44. https://doi.org/10.1097/HCM.0b013e3182078b1d
- Adamou, M., Goddard, A., Kyriakidou, N., Mooney, A., O'Donoghue, D., Pattani, S., & Roycroft, M. (2020). The wellbeing thermometer: A novel framework for measuring wellbeing. *Psychology*, *11*(10), 1471-1480. https://doi.org/10.4236/psych.2020.1110093
- Aguinis, H., Gottfredson, R. K., & Joo, H. (2013). Best-practice recommendations for defining, identifying, and handling outliers. *Organizational Research Methods*, 16(2), 270-301. https://doi.org/10.1177/1094428112470848
- Akhtar, S., Luqman, R., Raza, F., Riaz, H., Tufail, H. S., & Shahid, J. (2017). The impact of workplace incivility on the psychological wellbeing of employees through emotional exhaustion. *European Online Journal of Natural and Social Sciences*, 6(3), 492-507.
- Akinci, C., & Saunders, M. N. K. (2015). Using questionnaire surveys for within-organisation HRD research. In M. N. K. Saunders & P. Tosey (Eds.), *Handbook of research methods on human resource development* (pp. 217-230). Edward Elgar Publishing. https://doi.org/10.4337/9781781009246.00024
- Akram, Z., Khan, A. G., Akram, U., Ahmad, S., & Song, L. J. (2022). A contemporary view of interpersonal aggression and cyberbullying through ICT: Multilevel insights from LMX differentiation. *Internet Research*, 32(5), 1700-1724. https://doi.org/10.1108/INTR-11-2020-0659
- Al-Daihani, S. M., Al-Qallaf, J. S., & AlSaheeb, S. A. (2018). Use of social media by social science academics for scholarly communication. *Global Knowledge, Memory and Communication*, 67(6/7), 412-424. https://doi.org/10.1108/GKMC-11-2017-0091
- Alharahsheh, H. H., & Pius, A. (2020). A review of key paradigms: Positivism vs interpretivism. Global Academic Journal of Humanities and social Sciences, 2(3), pp. 39-43. https://www.doi.org/10.36348/gajhss.2020.v02i03.001
- Alin, A. (2010). Multicollinearity. Wiley Interdisciplinary Reviews. Computational Statistics, 2(3), 370-374. https://doi.org/10.1002/wics.84
- AlMazrouei, H., Zacca, R., & Mustafa, G. (2022). Mediating-moderating effect of employee creativity and team potency on expatriate innovative work behaviour. *International Journal of Organizational Analysis (2005)*. https://doi.org/10.1108/IJOA-05-2021-2752
- Alotaibi, T. S. (2022). Flexibility and Saudi employees' perceptions of job satisfaction: A multisector study. *Journal of East-West Business*, 1-27. https://doi.org/10.1080/10669868.2022.2118926
- Alshamsi, A. I., Santos, A., & Thomson, L. (2022). Psychosocial safety climate moderates the effect of demands of hospital accreditation on healthcare professionals: A longitudinal study. *Frontiers in Health Services, 2*. https://doi.org/10.3389/frhs.2022.824619
- Alvesson, M., & Skoldberg, K. (2009). Positivism, social constructionism, critical realism: Three reference points in the philosophy of science. *Reflexive Methodology: New Vistas for Qualitative Research*, 15-52.

- Anand, P. (2016). *Happiness explained: What human flourishing is and what we can do to promote it* (1st Ed.). Oxford University Press.
- Anderson, J. C., & Gerbing, D. W. (1988). Structural equation modeling in practice: A review and recommended two-step approach. *Psychological Bulletin*, 103(3), 411-423. https://doi.org/10.1037/0033-2909.103.3.411
- Annas, J. (2002). Democritus and eudaimonism. In V. Caston & D. Graham (Eds.), Presocratic philosophy: Essays in honour of Alexander Mourelatos (pp. 169-182). Ashgate.
- Annas, J., & Wang, H. L. (1989). Aristotle on virtue and happiness. *Philosophy and Culture*, 35(4), 157-170.
- Anscombe, G. E. M. (1958). Modern moral philosophy. Philosophy, 33(124), 1-19.
- Argyle, M. (1999). Causes and correlates of happiness. In D. Kahneman, E. Diener, & N. Schwarz (Eds.), Well-being: The foundations of hedonic psychology (pp. 353-373). Russell Sage Foundation.
- Argyle, M. (2001). The psychology of happiness (2nd ed.). Routledge.
- Argyle, M., Martin, M., & Crossland, J. (1989). Happiness as a function of personality and social encounters. In J. P. Forgas & J. M. Innes (Eds.), *Recent advances in social psychology: An international perspective* (pp. 189-203). Elsevier.
- Asare, B. Y. A., Makate, M., Powell, D., Kwasnicka, D., & Robinson, S. (2022). Cost of health-related work productivity loss among fly-in fly-out mining workers in Australia. *International Journal of Environmental Research and Public Health*, 19(16), 10056. https://doi.org/10.3390/ijerph191610056
- Asparouhov, T. & Muthén, B. O. (2010). *Bayesian analysis using Mplus. Technical appendix*. Muthén & Muthén. https://www.statmodel.com
- Asparouhov, T., & Muthén, B. (2019). Latent variable centering of predictors and mediators in multilevel and time-series models. *Structural Equation Modeling*, 26(1), 119-142. https://doi.org/10.1080/10705511.2018.1511375
- Asparouhov, T., & Muthén, B. O. (2006). Multilevel mixture models. *Journal of Educational Psychology*, *3*, 1-28.
- Asparouhov, T., & Muthén, B. O. (2008). Multilevel mixture models. *Advances in Latent Variables Models*, 27-51.
- Aspinwall, L. G., & Brunhart, S. M. (1996). Distinguishing optimism from denial: Optimistic beliefs predict attention to health threats. *Personality and Social Psychology Bulletin*, 22(10), 993-1003. https://doi.org/10.1177/01461672962210002
- Austin, J. T., & Vancouver, J. B. (1996). Goal constructs in psychology: Structure, process, and content. *Psychological Bulletin*, 122(3), 338-375.
- Australian Bureau of Statistics (ABS). (2022, May 27). Australian industry. Annual estimate of key economic and financial performance of industries in Australia, including income, expenses, profit and capital expenditure. https://www.abs.gov.au/statistics/industry/industry-overview/australian-industry/latest-release
- Australian Institute of Health and Welfare. (2022). *Mental health services in Australia.* https://www.aihw.gov.au/reports/mental-health-services/mental-health-services-in-australia

- Awan, S. E., Bennamoun, M., Sohel, F., Sanfilippo, F., & Dwivedi, G. (2021). Imputation of missing data with class imbalance using conditional generative adversarial networks. *Neurocomputing*, 453, 164-171. https://doi.org/10.1016/j.neucom.2021.04.010
- Ayoko, O. B., Tan, P. P., & Li, Y. (2022). Leader-follower interpersonal behaviors, emotional regulation and LMX quality. *Journal of Management & Organization*, 1-18. https://doi.org/10.1017/jmo.2022.26
- Babič, K., Černe, M., Connelly, C. E., Dysvik, A., & Škerlavaj, M. (2019). Are we in this together? Knowledge hiding in teams, collective prosocial motivation and Leader-member exchange. *Journal of Knowledge Management*, 23(8), 1502-1522. https://doi.org/10.1108/JKM-12-2018-0734
- Baig, F. A., Han, X., Hasnain, Z., & Rogger, D. (2021). Introducing the worldwide bureaucracy indicators: A new global dataset on public sector employment and compensation. *Public Administration Review*, 81(3), 564-571. https://doi.org/10.1111/puar.13355
- Bailey, T. S., Dollard, M. F., McLinton, S. S., & Richards, P. A. M. (2015). Psychosocial safety climate, psychosocial and physical factors in the aetiology of musculoskeletal disorder symptoms and workplace injury compensation claims. *Work and Stress, 29*(2), 190-211. https://doi.org/10.1080/02678373.2015.1031855
- Bakker, A. B. (2015). Towards a multilevel approach of employee well-being. *European Journal of Work* and Organizational Psychology, 24(6), 839-843. https://doi.org/10.1080/1359432X.2015.1071423
- Bakker, A. B., & Demerouti, E. (2014). Job demands-resources theory. In P. Y. Chen & C. L. Cooper (Eds.), Work and wellbeing (pp. 37-64). Wiley Blackwell. https://doi.org/10.1002/9781118539415.wbwell019
- Balch, C. V. (2010). Internet survey methodology. Cambridge Scholars Publishing.
- Bamberger, S. G., Larsen, A., Vinding, A. L., Nielsen, P., Fonager, K., Nielsen, R. N., Ryom, P., & Omland, Ø. (2015). Assessment of work intensification by managers and psychological distressed and non-distressed employees: A multilevel comparison. *Industrial Health*, 53(4), 322-331. https://doi.org/10.2486/indhealth.2014-0176
- Baptiste, N. R. (2008). Tightening the link between employee wellbeing at work and performance: A new dimension for HRM, *Management Decision*, 46(2), 284-309. https://doi.org/10.1108/00251740810854168
- Barnett, V. (1994). Outliers in sample surveys. Journal of Applied Statistics, 21(5), 373-381. https://doi.org/10.1080/757584014
- Barry, M. S., Robert, I. S., & Lisa, H. P. (1994). Employee positive emotion and favorable outcomes at the workplace. *Organization Science*, 5(1), 51-71. https://doi.org/10.1287/orsc.5.1.51
- Barsade, S. G. (2002). The ripple effect: Emotional contagion and its influence on group behavior. *Administrative Science Quarterly*, 47(4), 644-675. https://doi.org/10.2307/3094912
- Bartel, C. A., & Saavedra, R. (2000). The collective construction of work group moods. *Administrative Science Quarterly*, 45(2), 197-231. https://doi.org/10.2307/2667070
- Baselmans, B. M. L., & Bartels, M. (2018). A genetic perspective on the relationship between eudaimonic -and hedonic well-being. *Scientific Reports*, 8(1), 1-10. https://doi.org/10.1038/s41598-018-32638-1

- Bauer, J., McAdams, D., & Pals, J. (2008). Narrative identity and eudaimonic well-being. *Journal of Happiness Studies*, 9(1), 81-104. https://doi.org/10.1007/s10902-006-9021-6
- Bayhan Karapinar, P., Metin Camgoz, S., & Tayfur Ekmekci, O. (2020). Employee wellbeing, workaholism, work-family conflict and instrumental spousal support: A moderated mediation model. *Journal of Happiness Studies*, 21(7), 2451-2471. https://doi.org/10.1007/s10902-019-00191x
- Beaumont, J. F., & Rivest, L. P. (2009). Dealing with outliers in survey data. In C. R. Rao (Ed.), Handbook of statistics: Sample surveys: Design, methods and applications (pp. 247-279). Elsevier. https://doi.org/10.1016/S0169-7161(08)00011-4
- Becher, H., & Dollard, M. F. (2016). Psychosocial safety climate and better productivity in Australian workplaces. Costs, productivity, presenteeism, absenteeism. Asia Pacific Centre for Work Health and Safety, WHO Collaborating Centre in Occupational Health. University of South Australia.
- Benight, C. C., Ironson, G., Klebe, K., Carver, C. S., Wynings, C., Burnett, K., Greenwood, D., Baum, A., & Schneiderman, N. (1999). Conservation of resources and coping self-efficacy predicting distress following a natural disaster: A causal model analysis where the environment meets the mind. *Anxiety, Stress, & Coping*, 12(2), 107-126. https://doi.org/10.1080/10615809908248325
- Benitez, M., Peccei, R., & Medina, F. J. (2019). Employee well-being profiles and service quality: A unitlevel analysis using a multilevel latent profile approach. *European Journal of Work and Organizational Psychology*, 28(6), 859-872. https://doi.org/10.1080/1359432X.2019.1678587
- Benny, C., Yamamoto, S., McDonald, S., Chari, R., & Pabayo, R. (2022). Modelling maternal depression: An agent-based model to examine the complex relationship between relative income and depression. *International Journal of Environmental Research and Public Health*, 19(7), 4208. https://doi.org/10.3390/ijerph19074208
- Berezan, O., Krishen, A. S., Agarwal, S., & Kachroo, P. (2020). Exploring loneliness and social networking: Recipes for hedonic well-being on Facebook. *Journal of Business Research*, 115, 258-265. https://doi.org/10.1016/j.jbusres.2019.11.009
- Bergsma, A., Poot, G., & Liefbroer, A. (2008). Happiness in the Garden of Epicurus. Journal of Happiness Studies, 9(3), 397-423. https://doi.org/10.1007/s10902-006-9036-z
- Berryman, S. (2016). Democritus. In J. Cooper (Ed.), *The Standford encyclopedia of philosophy*. Stanford, CA.
- Beyond Blue. (2022). About us. https://www.beyondblue.org.au/about-us/who-we-are-and-what-we-do
- Birdi, K., Clegg, C., Patterson, M., Robinson, A., Stride, C. B., Wall, T. D., & Wood, S. J. (2008). The impact of human resource and operational management practices on company productivity: A longitudinal study. *Personnel Psychology*, *61*(3), 467-501. https://doi.org/10.1111/j.1744-6570.2008.00136.x
- Blanca, M. J., Arnau, J., López-Montiel, D., Bono, R., & Bendayan, R. (2013). Skewness and kurtosis in real data samples. *Methodology*, 9(2), 78-84. https://doi.org/10.1027/1614-2241/a000057
- Blau, P. M. (1964). Exchange and power in social life. Wiley.

- Boateng, G. O., Neilands, T. B., Frongillo, E. A., Melgar-Quiñonez, H. R., & Young, S. L. (2018). Best practices for developing and validating scales for health, social, and behavioral research: A primer. *Frontiers in Public Health*, 6, 149. https://doi.org/10.3389/fpubh.2018.00149
- Bocean, C. G., Nicolescu, M. M., Cazacu, M., & Dumitriu, S. (2022). The role of social responsibility and ethics in employees' wellbeing. *International Journal of Environmental Research and Public Health*, 19(14), 8838. https://doi.org/10.3390/ijerph19148838
- Bognar, G. (2010). Authentic happiness. *Utilitas*, *22*(3), 272-284. https://doi.org/10.1017/S0953820810000191
- Bond, S., Tuckey, M., & Dollard, M. F. (2010). Psychosocial safety climate, workplace bullying, and symptoms of posttraumatic stress. *Organization Development Journal*, *28*(1), 37-56.
- Bookwala, J., & Fekete, E. (2009). The role of psychological resources in the affective well-being of never-married adults. *Journal of Social and Personal Relationships*, 26(4), 411-428. https://doi.org/10.1177/0265407509339995
- Bosmans, K., Hardonk, S., De Cuyper, N., & Vanroelen, C. (2015). Explaining the relation between precarious employment and mental well-being. A qualitative study among temporary agency workers. *Work*, 53(2), 249. https://doi.org/10.3233/WOR-152136
- Bradburn, N. (1969). The structure of psychological well-being. Adline.
- Bradford, T. W. (2015). Beyond fungible: Transforming money into moral and social resources. *Journal of Marketing*, *79*(2), 79-97. https://doi.org/10.1509/jm.13.0437
- Brassington, K., & Lomas, T. (2021). Can resilience training improve well-being for people in high-risk occupations? A systematic review through a multidimensional lens. *The Journal of Positive Psychology*, 16(5), 573-592. https://doi.org/10.1080/17439760.2020.1752783
- Braungart, J. M., Plomin, R., Defries, J. C., & Fulker, D. W. (1992). Genetic influence on tester-rated infant temperament as assessed by Bayley's infant behavior record: Nonadoptive and adoptive siblings and twins. *Developmental Psychology*, 28(1), 40-47. https://doi.org/10.1037/0012-1649.28.1.40
- Brink, D. O. (2000). Making a necessity of virtue: Aristotle and Kant on virtue. *Philosophical Review*, 109(3), 428. https://doi.org/10.1215/00318108-109-3-428
- Brough, P., Timms, C., O'Driscoll, M. P., Kalliath, T., Siu, O. L., Sit, C., & Lo, D. (2014). Work-life balance: A longitudinal evaluation of a new measure across Australia and New Zealand workers. *International Journal of Human Resource Management, 25*(19), 2724-2744. https://doi.org/10.1080/09585192.2014.899262
- Brunetto, Y., Dick, T., Xerri, M., & Cully, A. (2020). Building capacity in the healthcare sector: A strengths-based approach for increasing employees' well-being and organisational resilience. *Journal of Management & Organization*, 26(3), 309-323. https://doi.org/10.1017/jmo.2019.53
- Brunetto, Y., Farr-Wharton, B., Wankhade, P., Saccon, C., & Xerri, M. (2022a). Managing emotional labour: The importance of organisational support for managing police officers in England and Italy. *International Journal of Human Resource Management*, 1-23. Advance online publication. https://doi.org/10.1080/09585192.2022.2047755

- Brunetto, Y., Farr-Wharton, R., & Shacklock, K. (2011). Supervisor-subordinate communication relationships, role ambiguity, autonomy and affective commitment for nurses. *Contemporary Nurse: A Journal for the Australian Nursing Profession*, 39(2), 227-239. https://doi.org/10.5172/conu.2011.227
- Brunetto, Y., Saheli, N., Dick, T., & Nelson, S. (2022b). Psychosocial safety climate, psychological capital, healthcare SLBS' wellbeing and innovative behaviour during the COVID 19 pandemic. *Public Performance & Management Review*, 45(4), 751-772. https://doi.org/10.1080/15309576.2021.1918189
- Brunetto, Y., Shacklock, K., Teo, S., & Farr-Wharton, R. (2014). The impact of management on the engagement and well-being of high emotional labour employees. *International Journal of Human Resource Management*, 25(17), 2345-2363. https://doi.org/10.1080/09585192.2013.877056
- Brunetto, Y., Xerri, M., & Farr-Wharton, B. (2021). The link between organizational support, wellbeing and engagement for emergency service employees: A comparative analysis. *Public Money & Management*, 1-8. Advance online publication. https://doi.org/10.1080/09540962.2021.1987733
- Brunetto, Y., Xerri, M., Trinchero, E., Farr-Wharton, R., Shacklock, K., & Borgonovi, E. (2016). Publicprivate sector comparisons of nurses' work harassment using set: Italy and Australia. *Public Management Review*, 18(10), 1479-1503. https://doi.org/10.1080/14719037.2015.1114136

Buckingham, A., & Saunders, P. (2004). The survey methods workbook: From design to analysis. Polity.

Burgoon, J. K., Steen, L. A., & Dillman, L. (1995). Interpersonal adaptation: Dyadic interaction patterns. Cambridge University Press.

Burns, J. H. (2005). Happiness and utility: Jeremy Bentham's equation. Utilitas, 17(1), 46-61.

Burton, A. L., Lero Jonson, C., Miller, W. T., Petrich, D. M., & Burton, V. S. (2022). Understanding who is hired to work in U.S. prisons and why it matters: A call for research. *Corrections: Policy, Practice and Research*, 1-22. Advance online publication. https://doi.org/10.1080/23774657.2022.2101161

- Butler, J., & Kern, M. L. (2016). The PERMA-Profiler: A brief multidimensional measure of flourishing. *International Journal of Wellbeing*, 6(3) 1-48. https://doi.org/10.5502/ijw.v6i3.526
- Cacioppo, J. T., & Gardner, W. L. (1999). Emotion. Annual Review of Psychology, 191.
- Caporael, L. R. (1997). The evolution of truly social cognition: The core configurations model. *Personality and Social Psychology Review*, 1(4), 276-298. https://doi.org/10.1207/s15327957pspr0104_1
- Carter, R. L. (2006). Solutions for missing data in structural equation modeling. *Research & Practice in Assessment*, *1*, 4.
- Cassell, C., Cunliffe, A. L., & Grandy, G. (2018). *The SAGE handbook of qualitative business and management research methods: Methods and challenges*. SAGE Publications.
- Cassiano, M. S., Bennett, B. A., Andres, E., & Ricciardelli, R. (2022). What it takes to be a "good" correctional officer: Occupational fitness and co-worker expectations from the perspective of correctional officer recruits in Canada. *Criminology & Criminal Justice*, 174889582210874. Advance online publication. https://doi.org/10.1177/17488958221087488

- Cawcutt, K. A., Erdahl, L. M., Englander, M. J., Radford, D. M., Oxentenko, A. S., Girgis, L., Migliore, L. L., Poorman, J. A., & Silver, J. K. (2019). Use of a coordinated social media strategy to improve dissemination of research and collect solutions related to workforce gender equity. *Journal of Women's Health*, 28(6), 849-862. https://doi.org/10.1089/jwh.2018.7515
- Chang, C. M., Liu, L. W., Huang, H. C., & Hsieh, H. H. (2019). The influence of workplace incivility on employees' emotional exhaustion in recreational sport/fitness clubs: A cross-level analysis of the links between psychological capital and perceived service climate. *Healthcare*, 7(4), 159. https://doi.org/10.3390/healthcare7040159
- Chang, S., van Witteloostuijn, A., & Eden, L. (2010). From the editors: Common method variance in international business research. *Journal of International Business Studies*, 41(2), 178-184. https://doi.org/10.1057/jibs.2009.88
- Chen, H., Wang, H., Yuan, M., & Xu, S. (2021). Daily challenge/hindrance demands and cognitive wellbeing: A multilevel moderated mediation model. *Frontiers in Psychology*, 12, 616002. https://doi.org/10.3389/fpsyg.2021.616002
- Chen, Q., Kong, Y., Niu, J., Gao, W., Li, J., & Li, M. (2019). How leaders' psychological capital influences their followers' psychological capital: Social exchange or emotional contagion. *Frontiers* in Psychology, 10(July), 1578-1578. https://doi.org/10.3389/fpsyg.2019.01578
- Cheng, H., Fan, Y., & Lau, H. (2022). An integrative review on job burnout among teachers in China: Implications for human resource management. *International Journal of Human Resource Management*, 1-33. Advance online publication. https://doi.org/10.1080/09585192.2022.2078991
- Christakis, N. A., & Fowler, J. H. (2009). *Connected: The surprising power of our social networks and how they shape our lives* (1st Ed.). Little, Brown and Company.
- Clark, K., Duckham, M., Guillemin, M., Hunter, A., McVernon, J., O'Keefe, C., Pitkin, C., Prawer, S., Sinnott, R., Warr, D., & Waycott, J. (2018). Advancing the ethical use of digital data in human research: Challenges and strategies to promote ethical practice. *Ethics and Information Technology*, 21(1), 59-73. https://doi.org/10.1007/s10676-018-9490-4
- Clarke, Friede, T., Putz, R., Ashdown, J., Martin, S., Blake, A., Adi, Y., Parkinson, J., Flynn, P., Platt, S., & Stewart-Brown, S. (2011). Warwick-Edinburgh mental well-being scale (WEMWBS): Validated for teenage school students in England and Scotland. A mixed methods assessment. *BMC Public Health*, 11(1), 487-487. https://doi.org/10.1186/1471-2458-11-487
- Cobb, C. W., & Daly, H. (1989). The index of sustainable economic welfare. In H. E. Daly, J. B. Cobb & Cobb, C. W. (Eds.). For the common good. Redirecting the economy toward community, the environment, and a sustainable future. Beacon Press.
- Collier, J. E. (2020). *Applied structural equation modeling using AMOS: Basic to advanced techniques*. Routledge.
- Collis, J., & Hussey, R. (2013). Business research: A practical guide for undergraduate and postgraduate students (4th ed.). Palgrave Macmillan.
- Cook, T. D., & Campbell, D. T. (1976). The design and conduct of quasi-experiments and true experiments in fields settings. In M. Dunnette (Ed.), *Handbook of industrial and organizational psychology* (pp. 223-326). Rand McNally.

Cooper, D. R., & Schindler, P. S. (2014). Business research methods (12th ed.). McGraw-Hill/Irwin.

- Cornelissen, J., & Durand, R. (2012). More than just novelty: Conceptual blending and causality. *The Academy of Management Review*, *37*(1), 152-154. https://doi.org/10.5465/amr.2011.0129
- Corrales, D. C., Corrales, J. C., & Ledezma, A. (2018). How to address the data quality issues in regression models: A guided process for data cleaning. *Symmetry*, 10(4), 99. https://doi.org/10.3390/sym10040099
- Costa, A. C. (2003). Work team trust and effectiveness. *Personnel Review*, 32(5), 605-622. https://doi.org/10.1108/00483480310488360
- Costello, A. B., & Osborne, J. (2005). Best practices in exploratory factor analysis: Four recommendations for getting the most from your analysis. *Practical Assessment, Research & Evaluation*, 10, 7.
- Craighead, C. W., Ketchen, D. J., Dunn, K. S., & Hult, G. G. (2011). Addressing common method variance: Guidelines for survey research on information technology, operations, and supply chain management. *IEEE Transactions on Engineering Management*, 58(3), 578-588. https://doi.org/10.1109/TEM.2011.2136437
- Creswell, J. W. (2014). A concise introduction to mixed methods research. SAGE Publications.
- Creswell, J. W., & Creswell, J. D. (2018). *Research design: Qualitative, quantitative, and mixed methods approaches* (5th ed.). SAGE Publications.
- Croasmun, J. T., & Ostrom, L. (2011). Using likert-type scales in the social sciences. *Journal of Adult Education*, 40(1), 19.
- Cropanzano, R., & Mitchell, M. S. (2005). Social exchange theory: An interdisciplinary review. Journal of Management, 31(6), 874-900. https://doi.org/10.1177/0149206305279602
- Cropanzano, R., Anthony, E. L., Daniels, S. R., & Hall, A. V. (2016). Social exchange theory: A critical review with theoretical remedies. *Academy of Management Annals*, 11(1), 1-38. https://doi.org/10.5465/annals.2015.0099
- Crouch, W. (2022, September 5). Dan Murphy's offers on-the-spot interviews for 2,200 jobs. *Yahoo! Finance*. https://au.finance.yahoo.com/
- Csikszentmihalyi, M. (1991). Flow: The psychology of optimal experience (1st. ed.). Harper Collins.
- Cullen, F. T., Link, B. G., Wolfe, N. T., & Frank, J. (1985). The social dimensions of correctional officer stress. Justice Quarterly, 2(4), 505-533. https://doi.org/10.1080/07418828500088711
- D'Haenens, E., Van Damme, J., & Onghena, P. (2010). Multilevel exploratory factor analysis: Illustrating its surplus value in educational effectiveness research. *School Effectiveness and School Improvement*, 21(2), 209-235. https://doi.org/10.1080/09243450903581218
- Danna, K., & Griffin, R. W. (1999). Health and well-being in the workplace: A review and synthesis of the literature. *Journal of Management*, 25(3), 357-384. https://doi.org/10.1177/014920639902500305
- Danner, D. D., Snowdon, D. A., & Friesen, W. V. (2001). Positive emotions in early life and longevity: Findings from the nun study. *Journal of Personality and Social Psychology*, 80(5), 804-813. https://doi.org/10.1037/0022-3514.80.5.804

- Dansereau, F., Cashman, J., & Graen, G. (1973). Instrumentality theory and equity theory as complementary approaches in predicting the relationship of leadership and turnover among managers. *Organizational Behavior & Human Performance*, 10(2), 184-200. https://doi.org/10.1016/0030-5073(73)90012-3
- Dansereau, F., Yammarino, F. J., & Kohles, J. C. (1999). Multiple levels of analysis from a longitudinal perspective: Some implications for theory building. (Special topic forum on multilevel theory building). Academy of Management Review, 24(2), 346. https://doi.org/10.2307/259086
- Davies, A. C. L. (2021). Stress at work: Individuals or structures? *Industrial Law Journal*. https://doi.org/10.1093/indlaw/dwab006
- de Cieri, H., Shea, T., Cooper, B., & Oldenburg, B. (2019). Effects of work-related stressors and mindfulness on mental and physical health among Australian nurses and healthcare workers. *Journal of Nursing Scholarship*, 51(5), 580-589. https://doi.org/10.1111/jnu.12502
- de Jonge, J., & Schaufeli, W. B. (1998). Job characteristics and employee wellbeing: A test of Warr's Vitamin Model in health care workers using structural equation modelling, *Journal of Organizational Behavior*, 19(4),387-407. https://doi.org/10.1002/(SICI)1099-1379(199807)19:4<387::AID-JOB851>3.0.CO;2-9
- de Koning, R., Egiz, A., Kotecha, J., Ciuculete, A. C., Ooi, S. Z. Y., Bankole, N. D. A., Erhabor, J., Higginbotham, G., Khan, M., Dalle, D. U., Sichimba, D., Bandyopadhyay, S., & Kanmounye, U. S. (2021). Survey fatigue during the COVID-19 pandemic: An analysis of neurosurgery survey response rates. *Frontiers in Surgery*, *8*, 690680. https://doi.org/10.3389/fsurg.2021.690680
- de Vaus, D. A. (2014). Surveys in social research (6th ed.). Routledge.
- de Vente, W., Olff, M., van Amsterdam, J. G. C., Kamphuis, J. H., & Emmelkamp, P. M. G. (2003). Physiological differences between burnout patients and healthy controls: Blood pressure, heart rate, and cortisol responses. *Occupational and Environmental Medicine (London, England)*, 60(suppl 1), i54-i61. https://doi.org/10.1136/oem.60.suppl 1.i54
- Deci, E., & Ryan, R. (2008). Hedonia, eudaimonia, and well-being: An introduction. Journal of Happiness Studies, 9(1), 1-11. https://doi.org/10.1007/s10902-006-9018-1
- Deitz, G., Hansen, J. D., DeCarlo, T., & Babakus, E. (2021). How store managers' employee climate perceptions affect frontline employee, customer and store performance outcomes: An examination in the small-store setting. *European Journal of Marketing*, 55(10), 2620-2648. https://doi.org/10.1108/EJM-02-2020-0107
- Deloitte. (2017). The Wellbeing@Work Index. https://www2.deloitte.com/tl/en/pages/aboutdeloitte/solutions/wellbeing-work-index-2017.html
- Demerouti, E., Bakker, A. B., & Bulters, A. J. (2004). The loss spiral of work pressure, work-home interference and exhaustion: Reciprocal relations in a three-wave study. *Journal of Vocational Behavior*, 64(1), 131-149. https://doi.org/10.1016/S0001-8791(03)00030-7
- Denzin, N. K. (2012). Triangulation 2.0. *Journal of Mixed Methods Research*, 6(2), 80-88. https://doi.org/10.1177/1558689812437186

- Department of Health. (2022, March). *Prioritising mental health and suicide prevention workforce and governance (pillar 5)*. https://www.health.gov.au/sites/default/files/documents/2022/03/budget-2022-23-mental-health-workforce-and-governance-pillar-5.pdf
- DeSimone, J. A., & Harms, P. D. (2017). Dirty data: The effects of screening respondents who provide low-quality data in survey research. *Journal of Business and Psychology*, 33(5), 559-577. https://doi.org/10.1007/s10869-017-9514-9
- Deusdedit, B., & Alazeezi, A. S. (2022). Transformative leadership and organizational citizenship behavior in UAE. International Journal of Business and Technology Management, 4(3), 412-422. https://doi.org/10.55057/ijbtm.2022.4.3.35
- Diener, E. (1984). Subjective well-being. *Psychological Bulletin*, *95*(3), 542-575. https://doi.org/10.1037/0033-2909.95.3.542
- Diener, E. (2009). Assessing well-being the collected works of Ed Diener (1st ed.). Springer Netherlands. https://doi.org/10.1007/978-90-481-2354-4
- Diener, E., Suh, E. M., Lucas, R. E., & Smith, H. L. (1999). Subjective well-being: Three decades of progress. *Psychological Bulletin*, 125(2), 276-302. https://doi.org/10.1037/0033-2909.125.2.276
- Diener, E., Thapa, S., & Tay, L. (2020). Positive emotions at work. Annual Review of Organizational Psychology and Organizational Behavior, 7(1), 451-477. https://doi.org/10.1146/annurev-orgpsych-012119-044908
- Dienesch, R. M., & Liden, R. C. (1986). Leader-member exchange model of leadership: A critique and further development. *The Academy of Management Review*, 11(3), 618-634. https://doi.org/10.2307/258314
- Dodge, R., Daly, A. P., Huyton, J., & Sanders, L. D. (2012). The challenge of defining wellbeing. *International Journal of Wellbeing*, 2(3), 222-235. https://doi.org/10.5502/ijw.v2.i3.4
- Dollard, M. F. (2019). The PSC-4: A short PSC tool. In M. F. Dollard (Ed.), *Psychosocial safety climate*. Springer.
- Dollard, M. F., & Bakker, A. B. (2010). Psychosocial safety climate as a precursor to conducive work environments, psychological health problems, and employee engagement. *Journal of Occupational* and Organizational Psychology, 83(3), 579-599. https://doi.org/10.1348/096317909X470690
- Dollard, M. F., & Kang, S. (2007). Psychosocial safety climate measure. Work & Stress Research Group.
- Dollard, M. F., Bailey, T., Owen, M. (2022). A minute in time saves minds: Psychosocial safety climate (PSC- 4) cut-points for early detection of mental distress. *Research Square*. Advance online publication. https://doi.org/10.21203/rs.3.rs-1379315/v1
- Dollard, M. F., Opie, T., Lenthall, S., Wakerman, J., Knight, S., Dunn, S., Rickard, G., & MacLeod, M. (2012). Psychosocial safety climate as an antecedent of work characteristics and psychological strain: A multilevel model. *Work & Stress*, 26(4), 385-404. https://doi.org/10.1080/02678373.2012.734154
- Donaldson, S. I., & Grant-Vallone, E. J. (2002). Understanding self-report bias in organizational behavior research. *Journal of Business and Psychology*, 17(2), 245-260. https://doi.org/10.1023/A:1019637632584

- Doran, C. M., & Kinchin, I. (2017). A review of the economic impact of mental illness. Australian Health Review, 43(1), 43-48. https://doi.org/10.1071/AH16115
- Dubey, A., & Rasool, A. (2021). Efficient technique of microarray missing data imputation using clustering and weighted nearest neighbour. *Scientific Reports*, 11(1), 24297. https://doi.org/10.1038/s41598-021-03438-x
- Dulebohn, J. H., Bommer, W. H., Liden, R. C., Brouer, R. L., & Ferris, G. R. (2012). A meta-analysis of antecedents and consequences of Leader-member exchange: Integrating the past with an eye toward the future. *Journal of Management*, 38(6), 1715-1759. https://doi.org/10.1177/0149206311415280
- Dunn, E. C., Masyn, K. E., Johnston, W. R., & Subramanian, S. V. (2015). Modeling contextual effects using individual-level data and without aggregation: An illustration of multilevel factor analysis (MLFA) with collective efficacy. *Population Health Metrics*, 13(1), 12. https://doi.org/10.1186/s12963-015-0045-1
- Dunn, T. J., Baguley, T., & Brunsden, V. (2014). From alpha to omega: A practical solution to the pervasive problem of internal consistency estimation. *The British Journal of Psychology*, 105(3), 399-412. https://doi.org/10.1111/bjop.12046
- Dyer, N. G., Hanges, P. J., & Hall, R. J. (2005). Applying multilevel confirmatory factor analysis techniques to the study of leadership. *The Leadership Quarterly*, 16(1), 149-167. https://doi.org/10.1016/j.leaqua.2004.09.009
- Easton, S. D., Safadi, N. S., & Santo, L. D. (2022). Mental health outcomes among public social workers in the occupied Palestinian territories. *Journal of Loss & Trauma*, 1-19. Advance online publication. https://doi.org/10.1080/15325024.2022.2105483
- Edgar, F., Geare, A., Halhjem, M., Reese, K., & Thoresen, C. (2015). Well-being and performance: Measurement issues for HRM research. *The International Journal of Human Resource Management*, 26(15), 1983-1994.
- Edmondson, A. (1999). Psychological safety and learning behavior in work teams. *Administrative Science Quarterly*, 44(2), 350-383.
- Eisenberger, R., Cotterell, N., & Marvel, J. (1987). Reciprocation ideology. *Journal of Personality and Social Psychology*, 53(4), 743-750. https://doi.org/10.1037/0022-3514.53.4.743
- Ekman, P., Levenson, R. W., & Friesen, W. V. (1983). Autonomic nervous system activity distinguishes among emotions. *Science (American Association for the Advancement of Science)*, 221(4616), 1208-1210. https://doi.org/10.1126/science.6612338
- El-Den, S., Schneider, C., Mirzaei, A., & Carter, S. (2020). How to measure a latent construct: Psychometric principles for the development and validation of measurement instruments. *The International Journal of Pharmacy Practice*, 28(4), 326-336. https://doi.org/10.1111/ijpp.12600
- Elders, L. J. (2019). *The ethics of St. Thomas Aquinas. Happiness, natural law and the virtues*. The Catholic University of America Press.
- Ellis, F. Y. A., Amos-Abanyie, S., Kwofie, T. E., Amponsah-Kwatiah, K., Afranie, I., & Aigbavboa, C. O. (2022). Contribution of person-team fit parameters to teamwork effectiveness in construction project teams. *International Journal of Managing Projects in Business*, 15(6), 983-1002. https://doi.org/10.1108/IJMPB-04-2021-0114

- Ellison, J. M., & Jaegers, L. A. (2022). Suffering in silence: Violence exposure and post-traumatic stress disorder among jail correctional officers. *Journal of Occupational and Environmental Medicine*, 64(1), e28-e35. https://doi.org/10.1097/JOM.00000000002432
- Ellison, J. M., Cain, C. M., & Jaegers, L. A. (2022). Just another day's work: The nexus between workplace experiences and Post-Traumatic Stress Disorder (PTSD) in jail settings. *Journal of Criminal Justice*, 81, 101903. Advance online publication. https://doi.org/10.1016/j.jcrimjus.2022.101903
- Elorza, U., Harris, C., Aritzeta, A., & Balluerka, N. (2016). The effect of management and employee perspectives of high-performance work systems on employees' discretionary behaviour. *Personnel Review*, 45(1), 121-141. https://doi.org/10.1108/PR-07-2014-0167
- Emerson, R. M. (1976). Social exchange theory. *Annual Review of Sociology*, 2(1), 335-362. https://doi.org/10.1146/annurev.so.02.080176.002003
- Emmons, R. A., & McCullough, M. E. (2003). Counting blessings versus burdens: An experimental investigation of gratitude and subjective well-being in daily life. *Journal of Personality and Social Psychology*, 84(2), 377-389. https://doi.org/10.1037/0022-3514.84.2.377
- Emond, B. (2006). *The philosopher's happiness in Plato's "Republic*" [Doctoral dissertation, University of Ottawa]. http://dx.doi.org/10.20381/ruor-18552
- Engels, M., Boß, L., Engels, J., Kuhlmann, R., Kuske, J., Lepper, S., Lesener, L., Pavlista, V., Diebig, M., Lunau, T., Ruhle, S. A., Zapkau, F. B., Angerer, P., Hoewner, J., Lehr, D., Schwens, C., Süß, S., Wulf, I. C., & Gano, N. (2022). Facilitating stress prevention in micro and small-sized enterprises: Protocol for a mixed method study to evaluate the effectiveness and implementation process of targeted web-based interventions. *BMC Public Health*, *22*(1), 1-13. https://doi.org/10.1186/s12889-022-12921-7
- Engstrom, S., & Whiting, J. (1993). Aristotle, Kant, and the Stoics. Cambridge University Press.
- Fair Work Ombudsman. (2022). *Dictionary of the Fair Work Ombudsman*. Australian Government. https://www.fairwork.gov.au/dictionary/e
- Fareed, M. Z., Qin, S., Almutairi, M., Munir, K., & Fareed, M. M. S. (2022). Transformational leadership and project success: The mediating role of trust and job satisfaction. *Frontiers in Psychology*, 13. Advance online publication. https://doi.org/10.3389/fpsyg.2022.954052
- Farr-Wharton, B., Brunetto, Y., & Xerri, M. (2021). Austerity, staff inadequacy, and contracting-out social services: How many government inquiries does it take to improve social policy outcomes in aged care? *Australian Journal of Public Administration*, 80(4), 790-808. https://doi.org/10.1111/1467-8500.12463
- Farr-Wharton, B., Brunetto, Y., Hernandez Grande, A., Brown, K., Teo, S. (2022a). Emergency service workers: The role of policy and management in (re)shaping wellbeing for emergency service workers. *Review of Public Personnel Administration*. Advance online publication. https://doi.org/10.1177/0734371X221130977
- Farr-Wharton, B., Brunetto, Y., Xerri, M., Shriberg, A., Newman, S., & Dienger, J. (2022b). Work harassment in the UK and US nursing context. *Journal of Management & Organization*, 28(2), 348-362. https://doi.org/10.1017/jmo.2019.16

- Farr-Wharton, B., Brunetto, Y., Xerri, M., Shriberg, A., Newman, S., & Dienger, J. (2019). Work harassment in the UK and US nursing context. *Journal of Management & Organization*, 1-15. https://doi.org/10.1017/jmo.2019.16
- Feilzer, M. Y. (2010). Doing mixed methods research pragmatically: Implications for the rediscovery of pragmatism as a research paradigm. *Journal of Mixed Methods Research*, 4(1), 6-16. https://doi.org/10.1177/1558689809349691
- Ferrari, J., Luhrs, T., & Lyman, V. (2007). Eldercare volunteers and employees: Predicting caregiver experiences from service motives and sense of community. *Journal of Primary Prevention*, 28(5), 467-479. https://doi.org/10.1007/s10935-007-0108-6
- Finch, W. H. (2020). Exploratory factor analysis. SAGE Publications.
- Finkel, E. J., Simpson, J. A., & Eastwick, P. W. (2017). The psychology of close relationships: Fourteen core principles. *Annual Review of Psychology*, 68(1), 383-411. https://doi.org/10.1146/annurevpsych-010416-044038
- Fordyce, M. (1988). A review of research on the happiness measures: A sixty second index of happiness and mental health. *Social Indicators Research*, 20(4), 355-381. https://doi.org/10.1007/BF00302333
- Fowers, B. J. (2016). Aristotle on eudaimonia: On the virtue of returning to the source. In J. Vittersø (Ed.), Handbook of eudaimonic well-being. International handbooks of quality-of-life (1st ed.). Springer International Publishing. https://doi.org/10.1007/978-3-319-42445-3
- Fraenkel, J. R., Wallen, N. E., & Hyun, H. H. (2014). *How to design and evaluate research in education*. (9th ed.). McGraw Hill.
- França, C. M., Couto, R. S., & Velloso, P. B. (2021). Missing Data Imputation in Internet of Things Gateways. *Information*, 12(10), 425. https://doi.org/10.3390/info12100425
- Franken, E., Bentley, T., Shafaei, A., Farr-Wharton, B., Onnis, L., & Omari, M. (2021). Forced flexibility and remote working: Opportunities and challenges in the new normal. *Journal of Management & Organization*, 27(6), 1131-1149. https://doi.org/10.1017/jmo.2021.40
- Frankl, V. E., & Lasch, I. (2006). Man's search for meaning. Beacon Press.
- Frederickson, B. L. (1998). What good are positive emotions. *Review of General Psychology*, 2(3), 377-389.
- Frederickson, B. L. (2001). The role of positive emotions in positive psychology. American Psychologist, 56(3), 218-226. https://doi.org/10.1037/0003-066X.56.3.218
- Frederickson, B. L. (2003). The value of positive emotions. *American Scientist*, 91(4), 330-335. https://doi.org/10.1511/2003.26.865
- Frederickson, B. L., & Losada, M. F. (2005). Positive affect and the complex dynamics of human flourishing. *American Psychologist*, 60(7), 678-686. https://doi.org/10.1037/0003-066X.60.7.678
- Friedman, P. H. (1992). Friedman well-being scale and professional manual. Foundation for Well-being.
- Fritz, C., & Sonnentag, S. (2006). Recovery, well-being, and performance-related outcomes: The role of workload and vacation experiences. *Journal of Applied Psychology*, 91(4), 936-945. https://doi.org/10.1037/0021-9010.91.4.936

- Fuller, C. M., Simmering, M. J., Atinc, G., Atinc, Y., & Babin, B. J. (2016). Common methods variance detection in business research. *Journal of Business Research*, 69(8), 3192-3198. https://doi.org/10.1016/j.jbusres.2015.12.008
- Gable, S. L., Reis, H. T., Impett, E. A., & Asher, E. R. (2004). What do you do when things go right? The intrapersonal and interpersonal benefits of sharing positive events. *Journal of Personality and Social Psychology*, 87(2), 228-245. https://doi.org/10.1037/0022-3514.87.2.228
- Gamero, N., González-Romá, V., & Peiró, J. M. (2008). The influence of intra-team conflict on work teams' affective climate: A longitudinal study. *Journal of Occupational and Organizational Psychology*, 81(1), 47–69. https://doi.org/10.1348/096317907X180441
- Garrihy, J. (2022). "That doesn't leave you": Psychological dirt and taint in prison officers' Occupational cultures and identities. *British Journal of Criminology*, 62(4), 982-999. https://doi.org/10.1093/bjc/azab074
- George, D. & Mallery, M. (2010). SPSS for Windows step by step: A simple suide and reference, 17.0 update (10th ed.). Pearson.
- Gerstner, C. R., & Day, D. V. (1997). Meta-analytic review of Leader-member exchange theory: Correlates and construct issues. *Journal of Applied Psychology*, 82(6), 827-844. https://doi.org/10.1037/0021-9010.82.6.827
- Ghauri, P., Grønhaug, K., & Strange, R. (2020). *Research methods in business studies* (5th ed.). Cambridge University Press. https://doi.org/10.1017/9781108762427
- Giromini, L., Young, G., & Sellbom, M. (2022). Assessing negative response bias using self-report measures: New articles, new issues. *Psychological Injury and Law*, 15(1), 1-21. https://doi.org/10.1007/s12207-022-09444-2
- Gliner, J. A., Morgan, G. A., & Leech, N. L. (2017). *Research methods in applied settings. An integrated approach to design and analysis* (3rd ed.). Routledge.
- Goldstone, L., Hamer, B., Grogan, E., & Ponniah, S. (2021). The future of work. What workers want: Winning the war for talent. PricewaterhouseCoopers. https://www.pwc.com.au/importantproblems/future-of-work/what-workers-want-report.pdf
- González-Romá, V., & Hernández, A. (2014). Climate uniformity: its influence on team communication quality, task conflict, and team performance. *The Journal of applied psychology*, 99(6), 1042–1058. https://doi.org/10.1037/a0037868
- González-Romá, V., & Hernández, A. (2017). Multilevel modeling: Research-based lessons for substantive researchers. *Annual Review of Organizational Psychology and Organizational Behavior*, 4, 183–210. https://doi.org/10.1146/annurev-orgpsych-041015-062407
- González-Romá, V., Peiró, J. M., & Tordera, N. (2002). An examination of the antecedents and moderator influences of climate strength. *Journal of Applied Psychology*, 87(3), 465–473. https://doi.org/10.1037/0021-9010.87.3.465
- Gorsuch, R. (1983). Factor analysis (2nd ed.). Lawrence.
- Gouldner, A. W. (1960). The norm of reciprocity: A preliminary statement. American Sociological Review, 25(2), 161-178. https://doi.org/10.2307/2092623

- Goulette, N., Denney, A. S., Crow, M. S., & Ferdik, F. V. (2022). "Anything can happen at any time": Perceived causes of correctional officer injuries. *Criminal Justice Review (Atlanta, Ga.)*, 47(1), 17-33. https://doi.org/10.1177/0734016820952521
- Grad, F. P. (2002). The preamble of the constitution of the World Health Organization. *Bulletin of the World Health Organization*, 80(12), 981-982.
- Graen, G. B., & Uhl-Bien, M. (1995). Relationship-based approach to leadership: Development of Leader-member exchange (LMX) theory of leadership over 25 years: Applying a multi-level multidomain perspective. *The Leadership Quarterly*, 6(2), 219-247. https://doi.org/10.1016/1048-9843(95)90036-5
- Graen, G. B., Novak, M. A., & Sommerkamp, P. (1982). The effects of leader-member exchange and job design on productivity and satisfaction: Testing a dual attachment model. *Organizational Behavior* and Human Performance, 30(1), 109-131. https://doi.org/10.1016/0030-5073(82)90236-7
- Grant, A. M. (2022). Working within a research paradigm and choosing a topic. In A. Grant (Ed.), *Doing your research project with documents. A step-by-step guide to take you from start to finish* (pp. 11-22). Bristol University Press. https://doi.org/10.51952/9781447363941.ch002
- Grant, A. M., Christianson, M. K., & Price, R. H. (2007). Happiness, health, or relationships? Managerial practices and employee well-being tradeoffs. *Academy of Management Perspectives*, *21*(3), 51-63.
- Grant, P., & McGhee, P. (2020). Hedonic versus (true) eudaimonic well-being in organizations. In S. Dhiman (Ed.), *The Palgrave handbook of workplace well-being* (pp. 1-19). Palgrave Macmillan. https://doi.org/10.1007/978-3-030-02470-3 37-1pp
- Green, R. G., Murphy, K. D., & Snyder, S. M. (2000). Should demographics be placed at the end or at the beginning of mailed questionnaires? An empirical answer to a persistent methodological question. *Social Work Research*, 24(4), 237-241.
- Gulzar, S., Ghauri, S., Abbas, Z., Hussain, K., & Bashiru Jibril, A. (2020). Antecedents of employee wellbeing in the banking sector: The moderating role of working environment. *Problems and Perspectives in Management*, 18(4), 448-460. https://doi.org/10.21511/ppm.18(4).2020.36
- Gunasekara, A. N., Wheeler, M. A., & Bardoel, A. (2022). The impact of working from home during COVID-19 on time allocation across competing demands. *Sustainability*, 14(15), 9126. https://doi.org/10.3390/su14159126
- Gupta, A., & Singh, D. (2017). Happiness: An overview and the ways of enhancing it. *Indian Journal of Positive Psychology*, 8(4), 583-586.
- Hackman, J. R. (1992). Group influences on individuals in organizations. In M. D. Dunnette & L. M.
 Hough (Eds.), *Handbook of Industrial and Organizational Psychology* (2nd ed., pp. 199-267).
 Consulting Psychologists Press.
- Hagino, C. (2002). A brief overview of the development process for written, self-report, health-related surveys. *Journal of the Canadian Chiropractic Association*, *46*(1), 11-21.
- Hagqvist, E., Ekberg, K., Lidwall, U., Nyberg, A., Landstad, B. J., Wilczek, A., Bååthe, F., & Sjöström,
 M. (2022). The Swedish HealthPhys study: Study description and prevalence of clinical burnout and major depression among physicians. *Chronic Stress* 6, 24705470221083866. https://doi.org/10.1177/24705470221083866

- Haj Salem, N., & Ishtiaq Ishaq, M. (2022). Employee engagement, innovative work behaviour, and employee wellbeing: Do workplace spirituality and individual spirituality matter? *Business Ethics,* the Environment & Responsibility. Advance online publication. https://doi.org/10.1111/beer.12463
- Hakanen, J. J., Peeters, M. C. W., & Perhoniemi, R. (2011). Enrichment processes and gain spirals at work and at home: A 3-year cross-lagged panel study (Report). *Journal of Occupational and Organizational Psychology*, 84(1), 8. https://doi.org/10.1111/j.2044-8325.2010.02014.x
- Hakanen, J. J., Peeters, M. C. W., & Schaufeli, W. B. (2018). Different types of employee well-being across time and their relationships with job crafting. *Journal of Occupational Health Psychology*, 23(2), 289-301. https://doi.org/10.1037/ocp0000081
- Halbesleben, J. R. B. (2010). The role of exhaustion and workarounds in predicting occupational injuries: A cross-lagged panel study of health care professionals. *Journal of Occupational Health Psychology*, 15(1), 1-16. https://doi.org/10.1037/a0017634
- Halbesleben, J. R. B., & Bowler, W. M. (2007). Emotional exhaustion and job performance: The mediating role of motivation. *Journal of Applied Psychology*, 92(1), 93-106. https://doi.org/10.1037/0021-9010.92.1.93
- Halbesleben, J. R. B., & Buckley, M. R. (2004). Burnout in organizational life. Journal of Management, 30(6), 859-879. https://doi.org/10.1016/j.jm.2004.06.004
- Halbesleben, J. R. B., & Wheeler, A. R. (2008). The relative roles of engagement and embeddedness in predicting job performance and intention to leave. *Work & Stress: Engagement at work: An Emerging Concept*, 22(3), 242-256. https://doi.org/10.1080/02678370802383962
- Halbesleben, J. R. B., & Wheeler, A. R. (2011). I owe you one: Coworker reciprocity as a moderator of the day-level exhaustion-performance relationship. *Journal of Organizational Behavior*, 32(4), 608. https://doi.org/10.1002/job.748
- Halbesleben, J. R. B., & Wheeler, A. R. (2015). To invest or not? The role of coworker support and trust in daily reciprocal gain spirals of helping behavior. *Journal of Management*, 41(6), 1628-1650. https://doi.org/10.1177/0149206312455246
- Halbesleben, J. R. B., Harvey, J., & Bolino, M. C. (2009). Too engaged? A conservation of resources view of the relationship between work engagement and work interference with family. *Journal of Applied Psychology*, 94(6), 1452-1465. https://doi.org/10.1037/a0017595
- Halbesleben, J. R. B., Neveu, J.-P., Paustian-Underdahl, S. C., & Westman, M. (2014). Getting to the "COR": Understanding the role of resources in Conservation of Resources Theory. *Journal of Management*, 40(5), 1334-1364. https://doi.org/10.1177/0149206314527130
- Hall, G. B., Dollard, M. F., & Coward, J. (2010). Psychosocial safety climate: Development of the PSC-12. International Journal of Stress Management, 17(4), 353-383. https://doi.org/10.1037/a0021320
- Hall, L. H., Johnson, J., Watt, I., Tsipa, A., & O'Connor, D. B. (2016). Healthcare staff wellbeing, burnout, and patient safety: A systematic review. *PLoS ONE*, 11(7). https://doi.org/10.1371/journal.pone.0159015
- Hallquist, M. (2017, Dec 7). *Multilevel SEM*. https://psu-psychology.github.io/psy-597-SEM/15_multilevel/multilevel_sem.html

- Hameed, I., Ijaz, M. U., & Sabharwal, M. (2022). The impact of human resources environment and organizational identification on employees' psychological well-being. *Public Personnel Management*, 51(1), 71-96. https://doi.org/10.1177/00910260211001397
- Harker, L., & Keltner, D. (2001). Expressions of positive emotion in women's college yearbook pictures and their relationship to personality and life outcomes across adulthood. *Journal of Personality and Social Psychology*, 80(1), 112-124. https://doi.org/10.1037/0022-3514.80.1.112
- Harradine, A., Batanero, C., & Rossman, A. (2011). Students and teachers' knowledge of sampling and inference. In C. Batanero, G. Burrill, & C. Reading (Eds.), *Teaching statistics in school mathematics-challenges for teaching and teacher education* (pp. 235-246). Springer Netherlands. https://doi.org/10.1007/978-94-007-1131-0_24
- Harunavamwe, M., & Ward, C. (2022). The influence of technostress, work-family conflict, and perceived organisational support on workplace flourishing amidst COVID-19. *Frontiers in Psychology*, 13, 1-17. https://doi.org/10.3389/fpsyg.2022.921211
- Hatfield, E., Cacioppo, J. T., & Rapson, R. L. (1993). Emotional contagion. Current Directions in Psychological Science, 2(3), 96-100. https://doi.org/10.1111/1467-8721.ep10770953
- Hayes, A. F., & Coutts, J. J. (2020). Use omega rather than cronbach's alpha for estimating reliability. *Communication Methods and Measures*, 14(1), 1-24. https://doi.org/10.1080/19312458.2020.1718629
- Heijmans, N., Lieshout, J. van, & Wensing, M. J. (2015). Improving participation rates by providing choice of participation mode: Two randomized controlled trials. *BMC Medical Research Methodology*, 15(1), 29-29. https://doi.org/10.1186/s12874-015-0021-2
- Herrando, C., & Constantinides, E. (2021). Emotional contagion: A brief overview and future directions. *Frontiers in Psychology*, 12, 712606-712606. https://doi.org/10.3389/fpsyg.2021.712606
- Herzog-Evans, M., & Sturgeon, J. (2022). "Lockdown probation leaders" and fundamental human needs. *European Journal of Probation*, *14*(2), 148-168. https://doi.org/10.1177/20662203221113822
- Higgins, E. M., & Swartz, K. (2022). Pursuing a correctional career: The motivations and reasons for staying. *Corrections: Policy, Practice and Research*, 7(4), 275-295. https://doi.org/10.1080/23774657.2020.1793429
- Higgins, E. M., Smith, J., & Swartz, K. (2022a). "We keep the nightmares in their cages": Correctional culture, identity, and the warped badge of honor. *Criminology*, 60(3), 429-454. https://doi.org/10.1111/1745-9125.12306
- Higgins, E. M., Swartz, K., & Roberts, A. (2022b). How conflict "bleeds over" for correctional staff:
 Exploring work-family conflict through correctional subculture. *Journal of Crime & Justice*, 45(1), 87-102. https://doi.org/10.1080/0735648X.2020.1870526
- Hoang, G., Luu, T. T., Nguyen, T. T., Du, T., & Le, L. P. (2022). Examining the effect of entrepreneurial leadership on employees' innovative behavior in SME hotels: A mediated moderation model. *International Journal of Hospitality Management*, 102, 103142. Advance online publication. https://doi.org/10.1016/j.ijhm.2022.103142
- Hobfoll, S. E. (1989). Conservation of resources. A new attempt at conceptualizing stress. *The American Psychologist*, *44*(3), 513-524. https://doi.org/10.1037/0003-066X.44.3.513

- Hobfoll, S. E. (2001). The influence of culture, community, and the nested-self in the stress process: Advancing conservation of resources theory. *Applied Psychology*, 50(3), 337-421. https://doi.org/10.1111/1464-0597.00062
- Hobfoll, S. E. (2002). Social and psychological resources and adaptation. *Review of General Psychology*, 6(4), 307-324. https://doi.org/10.1037/1089-2680.6.4.307
- Hochschild, A. R. (2012). *The managed heart commercialization of human feeling* (3rd ed.). University of California Press.
- Hochschild, A. R. (2013). Afterword: Welfare state reform, recognition and emotional labor. Social Policy and Society: A Journal of the Social Policy Association, 12(3), 487-489. https://doi.org/10.1017/S1474746413000122
- Hochwarter, W. A., Laird, M. D., & Brouer, R. L. (2008). Board up the windows: The interactive effects of hurricane-induced job stress and perceived resources on work outcomes. *Journal of Management*, 34(2), 263-289. https://doi.org/10.1177/0149206307309264
- Hoffman, M. L. (2002). How automatic and representational is empathy, and why. *The Behavioral and Brain Sciences*, *25*(1), 38-39. https://doi.org/10.1017/S0140525X02410011
- Hoffman, M. L. (2019). On the interpretation of parameters in multivariate multilevel models across different combinations of model specification and estimation. *Advances in Methods and Practices in Psychological Science*, 2(3), 288-311. https://doi.org/10.1177/2515245919842770
- Hoffman, M. L., & Walters, R. W. (2022). Catching up on multilevel modeling. Annual Review of Psychology, 73(1), 659-689. https://doi.org/10.1146/annurev-psych-020821-103525
- Holman, D. (2002). Employee wellbeing in call centres, *Human Resource Management Journal*, 12(4), 35-50. https://doi.org/10.1111/j.1748-8583.2002.tb00076.x
- Hoque, A. S. M. M., Gwadabe, U. M., & Rahman, M. A. (2017). Corporate entrepreneurship upshot on innovation performance: The mediation of employee engagement. *Journal of Humanities, Language, Culture and Business, 1*(6), 54-67.
- Hoque, Z., & Rana, T. (2018). Dealing with human ethical issues in research: Some advice. Methodological Issues in Accounting Research: Theories & Methods, 29, 602-612
- House, R., Rousseau, D. M., & Thomas-Hunt, M. (1995). The meso paradigm: A framework for integration of micro and macro organizational (pp. 71-114). In L. L. Cummings & B. Staw (Eds.), *Research in organizational behaviour*. JAI Press.
- Hsu, F. S., Liu, Y., & Tsaur, S.-H. (2019). The impact of workplace bullying on hotel employees' wellbeing: Do organizational justice and friendship matter? *International Journal of Contemporary Hospitality Management*, 31(4), 1702-1719. https://doi.org/10.1108/IJCHM-04-2018-0330
- Hu, L. T., & Bentler, P. M. (1999). Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. *Structural Equation Modeling*, 6(1), 1-55. https://doi.org/10.1080/10705519909540118
- Huang, F. L., & Cornell, D. G. (2016). Using multilevel factor analysis with clustered data: Investigating the factor structure of the positive values scale. *Journal of Psychoeducational Assessment*, 34(1), 3-14. https://doi.org/10.1177/0734282915570278

- Huo, C., Arslan Safdar, M., Waheed Akhtar, M., & Misbah A. (2022). Linking responsible leadership and green innovation: The role of knowledge sharing and Leader-member exchange. *Frontiers in Environmental Science*, 10. Advance online publication. https://doi.org/10.3389/fenvs.2022.945817
- Huo, M., & Boxall, P. (2022). Do workers respond differently to learning from supervisors and colleagues? A study of job resources, learning sources and employee wellbeing in China. *International Journal of Human Resource Management*, 33(4), 742-762. https://doi.org/10.1080/09585192.2020.1744029
- Huo, M., Boxall, P., & Cheung, G. W. (2022). Lean production, work intensification and employee wellbeing: Can line-manager support make a difference? *Economic and Industrial Democracy*, 43(1), 198-220. https://doi.org/10.1177/0143831X19890678
- Huppert, F., & So, T. (2013). Flourishing across Europe: Application of a new conceptual framework for defining well-being. *Social Indicators Research*, 110(3), 837-861. https://doi.org/10.1007/s11205-011-9966-7
- Hussain, K., Abbas, Z., Gulzar, S., Jibril, A. B., & Hussain, A. (2020). Examining the impact of abusive supervision on employees' psychological wellbeing and turnover intention: The mediating role of intrinsic motivation. *Cogent Business & Management*, 7(1), 1818998. https://doi.org/10.1080/23311975.2020.1818998
- Idris, M. A., Dollard, M. F., Coward, J., & Dormann, C. (2012). Psychosocial safety climate: Conceptual distinctiveness and effect on job demands and worker psychological health. *Safety Science*, 50(1), 19-28. https://doi.org/10.1016/j.ssci.2011.06.005
- Ilies, R., Schwind, K. M., & Heller, D. (2007). Employee well-being: A multilevel model linking work and nonwork domains. *European Journal of Work and Organizational Psychology*, 16(3), 326-341. https://doi.org/10.1080/13594320701363712
- Irani, L. (2019). *Chasing innovation making entrepreneurial citizens in modern India*. University of California.
- Ito, J. K., & Brotheridge, C. M. (2003). Resources, coping strategies, and emotional exhaustion: A conservation of resources perspective. *Journal of Vocational Behavior*, 63(3), 490-509. https://doi.org/10.1016/S0001-8791(02)00033-7
- Jaiswala, A., Dyaramb, L., & Khatric, N. (2022). Interplay of diversity, inclusion, and politics: Impact on employee well-being. *IIMB Management Review*. Advance online publication. *https://doi.org/10.1016/j.iimb.2022.08.002*
- James, L. R., & Jones, A. P. (1976). Organizational structure: A review of structural dimensions and their conceptual relationships with individual attitudes and behavior. Organizational Behavior and Human Performance, 16(1), 74-113. https://doi.org/10.1016/0030-5073(76)90008-8
- Jankelova, N., & Joniakova, Z. (2021). Communication skills and transformational leadership style of first-line nurse managers in relation to job satisfaction of nurses and moderators of this relationship. *Healthcare*, 9(3), 346. https://doi.org/10.3390/healthcare9030346
- Jian, G. (2022). From empathic leader to empathic leadership practice: An extension to relational leadership theory. *Human Relations (New York)*, 75(5), 931-955. https://doi.org/10.1177/0018726721998450

- Jiang, Z., & Qu, Y. (2022). Leader favorable feedback and withdrawal behavior: A moderated mediation model of gratitude and Leader-member exchange. *Personnel Review*. https://doi.org/10.1108/PR-04-2021-0228
- Johnson, A., Dey, S., Nguyen, H., Groth, M., Joyce, S., Tan, L., Glozier, N., & Harvey, S. B. (2020). A review and agenda for examining how technology-driven changes at work will impact workplace mental health and employee well-being. *Australian Journal of Management*, 45(3), 402-424. https://doi.org/10.1177/0312896220922292
- Jong, J., & Ford, M. T. (2016). The lagged effects of job demands and resources on organizational commitment in federal government agencies: A multi-level analysis. *Journal of Public Administration Research and Theory*, 26(3), 475-492. https://doi.org/10.5465/ambpp.2015.13949abstract
- Jordan, P. J., & Troth, A. C. (2020). Common method bias in applied settings: The dilemma of researching in organizations. *Australian Journal of Management*, 45(1), 3-14. https://doi.org/10.1177/0312896219871976
- Jung, H. S., & Yoon, H. H. (2019). Emotional contagion and collective commitment among leaders and team members in deluxe hotel. *Service Business*, 13(4), 737-754. https://doi.org/10.1007/s11628-019-00403-7
- Juniper, B. (2011). Defining employee wellbeing. Occupational Health, 63(10), 25.
- Kahneman, D., Diener, E., & Schwarz, N. (1999). *Well-being: The foundations of hedonic psychology*. Russell Sage Foundation.
- Kahneman, D., Slovic, P., & Tversky, A. (1982). Judgment under uncertainty: Heuristics and biases. Cambridge University Press.
- Kalshoven, K., & Boon, C. T. (2012). Ethical leadership, employee well-being, and helping: The moderating role of human resource management. *Journal of Personnel Psychology*, 11(1), 60-68. https://doi.org/10.1027/1866-5888/a000056
- Kammerlander, N., Kipfelsberger, P., & Herhausen, D. (2017). The impact of family management on employee well-being: A multilevel study. *Academy of Management Proceedings*, 2017(1), 11313. https://doi.org/10.5465/ambpp.2017.109
- Kankam, P. K. (2019). The use of paradigms in information research. *Library & Information Science Research*, 41(2), 85-92. https://doi.org/10.1016/j.lisr.2019.04.003
- Kapoor, K. K., Tamilmani, K., Rana, N. P., Patil, P., Dwivedi, Y. K., & Nerur, S. (2017). Advances in social media research: Past, present and future. *Information Systems Frontiers*, 20(3), 531-558. https://doi.org/10.1007/s10796-017-9810-y
- Karanja, E., Zaveri, J., & Ahmed, A. (2013). How do MIS researchers handle missing data in surveybased research: A content analysis approach. *International Journal of Information Management*, 33(5), 734-751. https://doi.org/10.1016/j.ijinfomgt.2013.05.002
- Kashdan, T. B., Biswas-Diener, R., & King, L. A. (2008). Reconsidering happiness: The costs of distinguishing between hedonics and eudaimonia. *The Journal of Positive Psychology*, 3(4), 219-233. https://doi.org/10.1080/17439760802303044

- Kaushik, V., & Walsh, C. A. (2019). Pragmatism as a research paradigm and its implications for social work research. *Social Sciences*, 8(9), 255. https://doi.org/10.3390/socsci8090255
- Keena, L. D., Lambert, E. G., Haynes, S. H., May, D., & Doty, P. A. (2022). Testing the job demandsresources model with organizational trust among prison staff. *Criminal Justice Review*, 47(2), 148-166. https://doi.org/10.1177/07340168221076789
- Keltner, D., & Bonanno, G. A. (1997). A study of laughter and dissociation: Distinct correlates of laughter and smiling during bereavement. *Journal of Personality and Social Psychology*, 73(4), 687-702. https://doi.org/10.1037/0022-3514.73.4.687
- Keltner, D., & Cowen, A. (2021). A taxonomy of positive emotions. *Current Opinion in Behavioral Sciences*, 39, 216-221. https://doi.org/10.1016/j.cobeha.2021.04.013
- Kern, M. L., Waters, L. E., Adler, A., & White, M. A. (2015). A multidimensional approach to measuring well-being in students: Application of the PERMA framework. *The Journal of Positive Psychology*, 10(3), 262-271. https://doi.org/10.1080/17439760.2014.936962
- Kessler, R. C., Green, J. G., Gruber, M. J., Sampson, N. A., Bromet, E., Cuitan, M., Furukawa, T. A., Gureje, O., Hinkov, H., Hu, C. Y., Lara, C., Lee, S., Mneimneh, Z., Myer, L., Oakley-Browne, M., Posada-Villa, J., Sagar, R., Viana, M. C., & Zaslavsky, A. M. (2011). Screening for serious mental illness in the general population with the K6 screening scale: Results from the WHO World Mental Health (WMH) survey initiative. *International Journal of Methods in Psychiatric Research, 20*(1), 62. https://doi.org/10.1002/mpr.333
- Keyes, C. L. M. (2002). The mental health continuum: From languishing to flourishing in life. Journal of Health and Social Behavior, 43(2), 207-222. https://doi.org/10.2307/3090197
- Khan, A. J., & Iqbal, J. (2020). Training and employee commitment: The social exchange perspective. *Journal of Management Sciences*, 7(1), 88-100. http://dx.doi.org/10.20547/jms.2014.2007106
- Kim, M. H., & Kim, J. (2021). Corporate social responsibility, employee engagement, well-being and the task performance of frontline employees. *Management Decision*, 59(8), 2040-2056. https://doi.org/10.1108/MD-03-2020-0268
- Kim, M. H., & Yi, Y. J. (2019). Impact of leader-member-exchange and team-member-exchange on nurses' job satisfaction and turnover intention. *International Nursing Review*, 66(2), 242-249. https://doi.org/10.1111/inr.12491
- King, L. A. (2001). The health benefits of writing about life goals. *Personality and Social Psychology Bulletin*, 27(7), 798-807. https://doi.org/10.1177/0146167201277003
- King, S. (2021, May 6). A brief overview of the mental health inquiry report. Speech to the mental health coordinating council. Productivity Commission. https://www.pc.gov.au/mediaspeeches/speeches/mental-health/mental-health-brief-overview.pdf
- Kiran, S. (2021). Occupational health could be the new normal challenge in the trade and health cycle: Keywords analysis between 1990 and 2020. Safety and Health at Work, 12(2), 272-276. https://doi.org/10.1016/j.shaw.2020.11.003
- Klein, K. J., Dansereau, F., & Hall, R. J. (1994). Levels issues in theory development, data collection, and analysis. Academy of Management Review, 19(2), 195. https://doi.org/10.2307/258703
- Kline, P. (2013). Handbook of psychological testing (2nd ed.). Routledge.

- Kock, N. (2015). Common method bias in PLS-SEM: A full collinearity assessment approach. *International Journal of e-Collaboration*, 11(4), 1-10. https://doi.org/10.4018/ijec.2015100101
- Kollo, T. (2008). Multivariate skewness and kurtosis measures with an application in ICA. *Journal of Multivariate Analysis*, 99, 2328-2338.
- Kozlowski, S. W. J., & Klein, K. J. (2000). A multilevel approach to theory and research in organizations: Contextual, temporal, and emergent processes. In K. J. Klein & S. W. J. Kozlowski (Eds.), *Multilevel theory, research, and methods in organizations: Foundations, extensions, and new directions* (pp. 3-90). Jossey-Bass.
- Krekel, C., Ward, G., & De Neve, J. E. (2019). Employee wellbeing, productivity, and firm performance. Saïd Business School WP 2019-04. http://dx.doi.org/10.2139/ssrn.3356581
- Kuhn, T. S. (1970). The structure of scientific revolutions (2nd ed.). University of Chicago Press.
- Kun, A., Balogh, P., & Krasz, K. G. (2016). Development of the work-related well-being questionnaire based on Seligman's PERMA model. *Periodica Polytechnica. Social and Management Sciences*, 25(1), 56. https://doi.org/10.3311/PPso.9326
- Kuoppala, J., Lamminpää, A., Liira, J., & Vainio, H. (2008). Leadership, job well-being, and health effects--a systematic review and a meta-analysis. *Journal of Occupational and Environmental Medicine*, 50(8), 904–915. https://doi.org/10.1097/JOM.0b013e31817e918d
- Kwak, S. K., & Kim, J. H. (2017). Statistical data preparation: Management of missing values and outliers. *Korean Journal of Anesthesiology*, 70(4), 407-411. https://doi.org/10.4097/kjae.2017.70.4.407
- Kyriazos, T. A. (2018). Applied psychometrics: Sample size and sample power considerations in factor analysis (EFA, CFA) and SEM in general. *Psychology*, 9(8), 2207-2230. https://doi.org/10.4236/psych.2018.98126
- Lai, M. H. C. (2021). Composite reliability of multilevel data: It's about observed scores and construct meanings. *Psychological Methods*, 26(1), 90-102. https://doi.org/10.1037/met0000287
- Lambert, E. G., Elechi, O. O., & Otu, S. (2022a). Testing the job demands-resources model in explaining life satisfaction of Nigerian correctional staff. *Psychology, Crime & Law*, 28(5), 435-453. https://doi.org/10.1080/1068316X.2021.1909020
- Lambert, E. G., Qureshi, H., Holbrook, M. A., Frank, J., & Hines, C. (2022b). Testing the job demandsresources model for organizational commitment among Indian correctional officers. *Criminal Justice Studies*, 1-20. Advance online publication. https://doi.org/10.1080/1478601X.2022.2115487
- Law, R., Dollard, M. F., Tuckey, M. R., & Dormann, C. (2011). Psychosocial safety climate as a lead indicator of workplace bullying and harassment, job resources, psychological health and employee engagement. *Accident Analysis and Prevention*, 43(5), 1782-1793. https://doi.org/10.1016/j.aap.2011.04.010
- Lawless, A., & Willocks, K. E. (2021). The wicked problem of employee wellbeing: Creating safe space within a change laboratory. *Action Learning*, 18(2), 121-135. https://doi.org/10.1080/14767333.2021.1931808
- Lazarus, R. S., & Folkman, S. (1984). Stress, appraisal, and coping. Springer.

- Le, A., Han, B. H., & Palamar, J. J. (2021). When national drug surveys "take too long": An examination of who is at risk for survey fatigue. *Drug and Alcohol Dependence*, 225, 108769. Advance online publication. https://doi.org/10.1016/j.drugalcdep.2021.108769
- Lee, K., & Pennings, J. M. (2002). Mimicry and the market: Adoption of a new organizational form. *Academy of Management Journal*, 45(1), 144-162. https://doi.org/10.2307/3069289
- Lee, R. T., & Ashforth, B. E. (1996). A meta-analytic examination of the correlates of the three dimensions of job burnout. *Journal of Applied Psychology*, 81(2), 123-133. https://doi.org/10.1037/0021-9010.81.2.123
- Lee, S. Y. (2007). Structural equation modeling a Bayesian approach. Wiley.
- Levy, R. (2011). Bayesian data-model fit assessment for structural equation modeling. *Structural Equation Modeling*, *18*(4), 663-685. https://doi.org/10.1080/10705511.2011.607723
- Lewin, K. (1951). Field theory in the social sciences. HarperCollins.
- Li, Y., Xu, M., Waters, L., Yang, X., Wu, C., & Wu, Z. (2022). Significant task activates trait gratitude for organizational citizenship behaviors: The mediating role of psychological availability. *European Journal of Work and Organizational Psychology*, 1-13. Advance online publication. https://doi.org/10.1080/1359432X.2022.2064745
- Liera, R. (2020). Moving beyond a culture of niceness in faculty hiring to advance racial equity. *American Educational Research Journal*, 57(5), 1954-1994. https://doi.org/10.3102/0002831219888624
- Likert, R. (1931). Minnesota Mechanical Ability Tests. Personnel Journal, 10(4), 278.
- Lin, N. (1983). Social resources and social actions: A progress report. Connections, 6, 10-16.
- Lin, P. T., Vu, T. T., Nguyen, V. P., & Wu, Q. (2022). Self-determination theory and accountant employees' psychological wellbeing: The roles of positive affectivity and psychological safety. *Frontiers in Psychology*, 13, 870771. Advance online publication. https://doi.org/10.3389/fpsyg.2022.870771
- Liu, Y., Wu, A. D., & Zumbo, B. D. (2010). The impact of outliers on cronbach's coefficient alpha estimate of reliability: Ordinal/rating scale item responses. *Educational and Psychological Measurement*, 70(1), 5-21. https://doi.org/10.1177/0013164409344548
- Loh, M. Y., Idris, M. A., Dollard, M. F., & Isahak, M. (2018). Psychosocial safety climate as a moderator of the moderators: Contextualizing JDR models and emotional demands effects. *Journal of Occupational and Organizational Psychology*, 91(3), 620-644.
- Loh, M. Y., Idris, M. A., Dormann, C., & Muhamad, H. (2019). Organisational climate and employee health outcomes: A systematic review. *Safety Science*, 118, 442-452. https://doi.org/10.1016/j.ssci.2019.05.052
- Lüdtke, O., Marsh, H. W., Robitzsch, A., Trautwein, U., Asparouhov, T., & Muthén, B. (2008). The multilevel latent covariate model: A new, more reliable approach to group-level effects in contextual studies. *Psychological Methods*, 13(3), 203-229. https://doi.org/10.1037/a0012869
- Luu, T. T. (2022). Family support and posttraumatic growth among tourism workers during the COVID-19 shutdown: The role of positive stress mindset. *Tourism Management*, 88, 104399. Advance online publication. https://doi.org/10.1016/j.tourman.2021.104399

- Lykken, D., & Tellegen, A. (1996). Happiness is a stochastic phenomenon. *Psychological Science*, 7(3), 186-189. https://doi.org/10.1111/j.1467-9280.1996.tb00355.x
- Lyons, T. F. (1971). Role clarity, need for clarity, satisfaction, tension, and withdrawal. *Organizational* behavior and human performance, 6(1), 99-110. https://doi.org/10.1016/0030-5073(71)90007-9
- Lyubomirsky, S., & Lepper, H. (1999). A measure of subjective happiness: Preliminary reliability and construct validation. *Social Indicators Research*, *46*(2), 137-155. https://doi.org/10.1023/A:1006824100041
- Lyubomirsky, S., Sheldon, K. M., & Schkade, D. (2005). Pursuing happiness: The architecture of sustainable change. *Review of General Psychology*, 9(2), 111-131. https://doi.org/10.1037/1089-2680.9.2.111
- Maarouf, H. (2019). Pragmatism as a supportive paradigm for the mixed research approach:
 Conceptualizing the ontological, epistemological, and axiological stances of pragmatism.
 International Business Research, 12(9). https://doi.org/10.5539/ibr.v12n9p1
- Mack, K., & Rhineberger-Dunn, G. (2022). What matters most? Comparing the impact of individual, job, and organizational factors on job stress and job satisfaction among juvenile justice personnel. *Criminal Justice Studies*, 35(1), 18-37. https://doi.org/10.1080/1478601X.2021.1929207
- Magen, Z., & Aharoni, R. (1991). Adolescents' contributing toward others: Relationship to positive experiences and transpersonal commitment. *Journal of Humanistic Psychology*, 31(2), 126-143. https://doi.org/10.1177/0022167891312015
- Majumdar, A., & Kumar, S. (2022). Organizational justice and employee well-being in India: Through a psychological lens. *Business Perspectives and Research*. Advance online publication. https://doi.org/10.1177/22785337221077396
- Mäkikangas, A., Bakker, A. B., Aunola, K., & Demerouti, E. (2010). Job resources and flow at work: Modelling the relationship via latent growth curve and mixture model methodology. *Journal of Occupational and Organizational Psychology*, 83(3), 795-814. https://doi.org/10.1348/096317909X476333
- Martinescu, E., Jansen, W., & Beersma, B. (2021). Negative gossip decreases targets' organizational citizenship behavior by decreasing social inclusion. A multi-method approach. *Group & Organization Management*, 46(3), 463-497. https://doi.org/10.1177/1059601120986876
- Martin-Howard, S., Reid, L., & Baglivio, M. T. (2022). COVID-19's Impact on black, female correctional officers and justice-involved individuals at Rikers Island jail. *Crime and Delinquency*, 68(8), 1247-1270. https://doi.org/10.1177/00111287211073676
- Maslach, C., & Jackson, S. (1981). Maslach burnout inventory: Research edition. Consulting Psychologist Press.
- Maslow, A. H. (1968). Toward a psychology of being (2nd ed.). Van Nostrand.
- Maslow, A. H. (1998). Toward a psychology of being (3rd ed.). J. Wiley & Sons.
- McLeod, J., Jenkin, A., Walters, G., & Irving, R. (2021). The role and performance of supporter directors: A social exchange theory perspective. *Sport Management Review*, 24(5), 862-885. https://doi.org/10.1080/14413523.2021.1880744

- Medina-Craven, M. N., & Ostermeier, K. (2020). Investigating justice and bullying among healthcare workers. *Employee Relations*, 43(1), 31-44. https://doi.org/10.1108/ER-04-2019-0195
- Megheirkouni, M. (2022). Psychological contract, leadership, and job satisfaction: An empirical investigation into the non-profit sports sector. *Annals of Leisure Research*, 25(2), 203-226. https://doi.org/10.1080/11745398.2020.1769488
- Mejia, C., Pittman, R., Beltramo, J. M. D., Horan, K., Grinley, A., & Shoss, M. K. (2021). Stigma & dirty work: In-group and out-group perceptions of essential service workers during COVID-19. *International Journal of Hospitality Management, 93*, 102772. https://doi.org/10.1016/j.ijhm.2020.102772
- Melamed, S., Shirom, A., Toker, S., Berliner, S., & Shapira, I. (2006). Burnout and risk of cardiovascular disease: Evidence, possible causal paths, and promising research directions. *Psychological Bulletin*, 132(3), 327-353. https://doi.org/10.1037/0033-2909.132.3.327
- Memon, U., Ali, A. J., & Nisa, Z. U. (2022). Teachers' war against terrorism: A mediated moderation model. *Journal of Public Affairs*, 22(1). https://doi.org/10.1002/pa.2406
- Mertler, C. A. (2020). Introduction to educational research (3rd ed.). SAGE Publications.
- Metcalfe, M. (2014). How concepts solve management problems. Edward Elgar.
- Meuleman, B., & Billiet, J. (2009). A Monte Carlo sample size study: How many countries are needed for accurate multilevel SEM?. Survey Research Methods, 3(1), 45-58. https://doi.org/10.18148/srm/2009.v3i1.666
- Miller, W. T., Burton, A. L., Jonson, C. L., & Burton, V. S. (2022). The creation of a correctional officer trainee self-efficacy index: An application of item response theory. *Journal of Forensic Psychology Research and Practice*, 22(3), 285-306. https://doi.org/10.1080/24732850.2021.1973235
- Milner, K., Greyling, M., Goetzel, R., Da Silva, R., Kolbe-Alexander, T., Patel, D., Nossel, C., & Beckowski, M. (2015). The relationship between leadership support, workplace health promotion and employee wellbeing in South Africa. *Health Promotion International*, 30(3), 514-522. https://doi.org/10.1093/heapro/dat064
- Mirzaei, A., Carter, S. R., Patanwala, A. E., & Schneider, C. R. (2022). Missing data in surveys: Key concepts, approaches, and applications. *Research in Social and Administrative Pharmacy*, 18(2), 2308-2316. https://doi.org/10.1016/j.sapharm.2021.03.009
- Mitchell, M. S., Cropanzano, R. S., & Quisenberry, D. M. (2012). Social exchange theory, exchange resources, and interpersonal relationships: A modest resolution of theoretical difficulties. In K. Törnblom & A. Kazemi (Eds.), *Handbook of Social Resource Theory* (pp. 99-118). Springer. https://doi.org/10.1007/978-1-4614-4175-5 6
- Mohd Salleh, E. S., Dato' Mansor, Z., Mohamed Zainal, S. R., & Md. Yasin, I. (2020). Multilevel analysis on employee wellbeing: The roles of authentic leadership, rewards, and meaningful work. Asian Academy of Management Journal, 25(1). https://doi.org/10.21315/aamj2020.25.1.7
- Moreno-Jiménez, B., Rodríguez-Muñoz, A., Garrosa Hernández, E., & Blanco, L. M. (2014). Development and validation of the Occupational Hardiness Questionnaire. *Psicothema*, 26(2), 207-214. https://doi.org/10.7334/psicothema2013.49

- Morgan, D. L. (2014). Pragmatism as a paradigm for social research. *Qualitative Inquiry*, 20(8), 1045-1053. https://doi.org/10.1177/1077800413513733
- Mueller, B. H., & Lee, J. (2002). Leader-member exchange and organizational communication satisfaction in multiple contexts. *The Journal of Business Communication*, 39(2), 220-244. https://doi.org/10.1177/002194360203900204
- Murphy, J. K., Noble, J., Chakraborty, P. A., Michlig, G. J., Greenshaw, A. J., Michalak, E. E., & Lam,
 R. W. (2023). Values and preferences related to workplace mental health programs and interventions: An International Survey. *medRxiv*. https://doi.org/10.1101/2023.03.02.23286684
- Muthén, B. O. (1994). Multilevel covariance structure analysis. *Sociological Methods & Research*, 22(3), 376-398. https://doi.org/10.1177/0049124194022003006
- Muthén, B. O. (2010). *Bayes analysis in Mplus: A brief introduction*. Muthén & Muthén. https://www.statmodel.com
- Muthén, L. K., & Muthén, B. O. (2009). Multilevel modelling with latent variables using Mplus: Crosssectional analysis [PowerPoint slides]. Starmodel. https://www.statmodel.com/download/Topic%207-v25.pdf
- Muthén, L. K., & Muthén, B. O. (2022, April 26). *Mplus version history*. Mplus. https://www.statmodel.com/verhistory.shtml
- Nasir, S., & Waheed, A. (2022). Mediating role of meaningful work between sustainable human resource management and employee engagement: A study on banking sector of Pakistan. *Al-Qantara*, 8(3), 186-198.
- Näswall, K., Wong, J., & Malinen, S. (2021). Measuring wellbeing in organizational contexts. In T. Wall, C. L. Cooper, & P. Brough (Eds.), *The SAGE handbook of organizational wellbeing* (pp. 463-477). https://doi.org/10.4135/9781529757187
- National Economic Foundation. (2014). *Well-being at work. A review of the literature*. https://b.3cdn.net/nefoundation/71c1bb59a2ce151df7_8am6bqr2q.pdf
- Neto, M., Carvalho, V. S., Chambel, M. J., Manuel, S., Miguel, J. P., & De Fatima Reis, M. (2016).
 Work-family conflict and employee well-being over time: The loss spiral effect. *Journal of Occupational and Environmental Medicine*, 58(5), 429-435.
 https://doi.org/10.1097/JOM.000000000000707
- Neveu, J. P. (2007). Jailed resources: Conservation of resources theory as applied to burnout among prison guards. *Journal of Organizational Behavior*, 28(1), 21-42. https://doi.org/10.1002/job.393
- Ng, T. W. H., & Feldman, D. C. (2012). Employee voice behavior: A meta-analytic test of the conservation of resources framework. *Journal of Organizational Behavior*, 33(2), 216-234. https://doi.org/10.1002/job.754
- Nishikawa, B. T., & Orsato, R. J. (2021). Professional services in the age of platforms: Towards an analytical framework. *Technological Forecasting & Social Change*, 173, 121131. Advance online publication. https://doi.org/10.1016/j.techfore.2021.121131
- Nunfam, V. F. (2021). Mixed methods study into social impacts of work-related heat stress on Ghanaian mining workers: A pragmatic research approach. *Heliyon*, 7(5), e06918. https://doi.org/10.1016/j.heliyon.2021.e06918
- Ocen, E., Francis, K., & Angundaru, G. (2017). The role of training in building employee commitment: The mediating effect of job satisfaction. *European Journal of Training and Development*, *41*(9), 742-757. https://doi.org/10.1108/EJTD-11-2016-0084
- Oktavio, A. (2020). To what extent is leader-member exchange and psychological safety able to influence counterproductive work behavior? Evidence from the hospitality industry in Surabaya, Indonesia. SHS Web of Conferences, 76. https://doi.org/10.1051/shsconf/20207601001
- Oliver, M. B. (2022). Social media use and eudaimonic well-being. *Current Opinion in Psychology*, 45, 101307. Advance online publication. https://doi.org/10.1016/j.copsyc.2022.101307
- Orsila, R., Luukkaala, T., Manka, M.-L., & Nygard, C.-H. (2011). A new approach to measuring workrelated well-being. *International Journal of Occupational Safety and Ergonomics*, 17(4), 341-359. https://doi.org/10.1080/10803548.2011.11076900
- Pallant, J. (2020). SPSS survival manual. A step by step guide to data analysis using IBM SPSSi (7th ed.). Routledge. https://doi.org/10.4324/9781003117452
- Panter, A. T., & Sterba, S. K. (2011). Handbook of ethics in quantitative methodology. Routledge.
- Parker, G. B., & Hyett, M. P. (2011). Measurement of well-being in the workplace: The development of the work well-being questionnaire. *Journal of Nervous & Mental Disease*, 199(6), 394-397.
- Parker, S. K., & Jorritsma, K. (2021). Good work design for all: Multiple pathways to making a difference. *European Journal of Work and Organizational Psychology*, 30(3), 456-468. https://doi.org/10.1080/1359432X.2020.1860121
- Parks, M., & McKay, L. (2022). Teacher emotions and emotional labour: The significance of staffroom relationships in an Australian high school. *Australian Educational Researcher*. Advance online publication. https://doi.org/10.1007/s13384-022-00529-0
- Peterson, C., Park, N., & Seligman, M. (2005). Orientations to happiness and life satisfaction: The full life versus the empty life. *Journal of Happiness Studies*, 6(1), 25-41. https://doi.org/10.1007/s10902-004-1278-z
- Petitta, L., Jiang, L., & Härtel, C. E. J. (2017). Emotional contagion and burnout among nurses and doctors: Do joy and anger from different sources of stakeholders matter?: Contagion and burnout among nurses and doctors. *Stress and Health*, 33(4), 358-369. https://doi.org/10.1002/smi.2724
- Petitta, L., Probst, T. M., Ghezzi, V., & Barbaranelli, C. (2019). Cognitive failures in response to emotional contagion: Their effects on workplace accidents. *Accident Analysis and Prevention*, 125, 165-173. https://doi.org/10.1016/j.aap.2019.01.018
- Pham, T. M., Pandis, N., & White, I. R. (2022). Missing data, part 2. Missing data mechanisms: Missing completely at random, missing at random, missing not at random, and why they matter. *American Journal of Orthodontics and Dentofacial Orthopedics*, 162(1), 138-139. https://doi.org/10.1016/j.ajodo.2022.04.001
- Platts, K., Breckon, J., & Marshall, E. (2022). Enforced home-working under lockdown and its impact on employee wellbeing: A cross-sectional study. *BMC Public Health*, 22(1), 199. https://doi.org/10.1186/s12889-022-12630-1

- Podsakoff, P. M., MacKenzie, S. B., Lee, J.-Y., & Podsakoff, N. P. (2003). Common method biases in behavioral research: A critical review of the literature and recommended remedies. *Journal of Applied Psychology*, 88(5), 879-903. https://doi.org/10.1037/0021-9010.88.5.879
- Podubinski, T., & Glenister, K. M. (2021). The pandemic is not occurring in a vacuum: The impact of COVID-19 and other disasters on workforce mental health in Australia. *Disaster Medicine and Public Health Preparedness*, 1-7. Advance online publication. https://doi.org/10.1017/dmp.2021.238
- Ponting, S. S. (2020). Organizational identity change: Impacts on hotel leadership and employee wellbeing. *The Service Industries Journal*, 40(1-2), 6-26. https://doi.org/10.1080/02642069.2019.1579799
- Porter, C. M. (2018). Long live social exchange theory. *Industrial and Organizational Psychology*, 11(3), 498-504. https://doi.org/10.1017/iop.2018.102
- Potter, R., Bailey, T., & Dollard, M. F. (2019). A qualitative investigation into high psychosocial safety climate university work groups (pp 251-270). In M. F. Dollard, C. Dormann, M. A. Idris (Eds.), *Psychosocial safety climate: A new work stress theory*. Springer. https://doi.org/10.1007/978-3-030-20319-1_10
- Preacher, K. J. (2017, July 24). *Multilevel structural equation modelling* (seminar). Statistical Horizons. https://statisticalhorizons.com/wp-content/uploads/MSEM-Sample-Material.pdf
- Preacher, K. J., Zhang, Z., & Zyphur, M. J. (2016). Multilevel structural equation models for assessing moderation within and across levels of analysis. *Psychological Methods*, 21(2), 189-205. https://doi.org/10.1037/met0000052
- Preacher, K. J., Zyphur, M. J., & Zhang, Z. (2010). A general multilevel SEM framework for assessing multilevel mediation. *Psychological Methods*, 15(3), 209-233. https://doi.org/10.1037/a0020141
- Prince, S. (2006). Socrates, Antisthenes, and the Cynics. In S. Ahbel-Rappe & R. Kamtekar (Eds.), *A companion to Socrates*. Blackwell Publishing.
- Prosek, E. A., & Gibson, D. M. (2021). Promoting rigorous research by examining lived experiences: A review of four qualitative traditions. *Journal of Counseling & Development*, 99(2), 167-177. https://doi.org/10.1002/jcad.12364
- Quinlan, C., Babin, B., Carr, J., Griffin, M., & Zikmund, W. G. (2019). Business research methods (2nd ed.). Cengage.
- Rachlis, C. (1978). Marcuse and the problem of happiness. Canadian Journal of Political and Social Theory, 2(1). 63-88.
- Ranalli, M. (2016). Recent Developments in Approximated-Likelihood Inference Methods for Latent Variable Models. *Econometrics: Econometric & Statistical Methods Special Topics eJournal*.
- Rehman, A. A., & Alharthi, K. (2016). An introduction to research paradigms. *International Journal of Educational Investigations*, 3(8), 51-59.
- Reid, M. (2008). Contemporary marketing in professional services. *The Journal of Services Marketing*, 22(5), 374-384. https://doi.org/10.1108/08876040810889148

- Rhineberger-Dunn, G., & Mack, K. Y. (2022). Workplace safety: Perceived dangerousness versus experienced fear among community corrections personnel. *Criminal Justice and Behavior*, 49(11), 1618-1636. https://doi.org/10.1177/00938548221105207
- Riggle, R. J., Edmondson, D. R., & Hansen, J. D. (2009). A meta-analysis of the relationship between perceived organizational support and job outcomes: 20 years of research. *Journal of Business Research*, 62(10), 1027-1030. https://doi.org/10.1016/j.jbusres.2008.05.003
- Rita Yi Man, L., Beiqi, T., & Kwong Wing, C. (2019). Sustainable construction safety knowledge sharing: A partial least square-structural equation modeling and a feedforward neural network approach. *Sustainability*, 11(20), 5831. https://doi.org/10.3390/su11205831
- Robinson, D. N. (1989). Aristotle's psychology. Columbia University Press.
- Robinson, K. J., Bosch, O. J., Levkowitz, G., Busch, K. E., Jarman, A. P., & Ludwig, M. (2019). Social creatures: Model animal systems for studying the neuroendocrine mechanisms of social behaviour. *Journal of Neuroendocrinology*, *31*(12), e12807. https://doi.org/10.1111/jne.12807
- Roche, M., Shang, S., Bentley, T., Catley, B., Blackwood, K., Teo, S., & Sutton, A. (2022). Mindfulness older workers and relational leadership. *Journal of Management & Organization*, 1-18. Advance online publication. https://doi.org/10.1017/jmo.2022.11
- Roesch, S. C., Aldridge, A. A., Stocking, S. N., Villodas, F., Leung, Q., Bartley, C. E., & Black, L. J. (2010). Multilevel factor analysis and structural equation modeling of daily diary coping data: Modeling trait and state variation. *Multivariate Behavioral Research*, 45(5), 767-789. https://doi.org/10.1080/00273171.2010.519276
- Rogers, C. R. (1961). On becoming a person: A therapist's view of psychotherapy. Constable.
- Roscigno, V. J., Zheng, H., & Crowley, M. (2022). Workplace age discrimination and socialpsychological well-being. *Society and Mental Health*, *12*(3), 195-214. https://doi.org/10.1177/21568693221116139
- Ryan, A. B. (2006). Post-positivist approaches to research. In A. B. Ryan (Ed.), *Researching and writing your thesis: A guide for postgraduate students* (pp. 12-26). MACE: Maynooth Adult and Community Education.
- Ryan, R. M., & Deci, E. L. (2001). On happiness and human potentials: A review of research on hedonic and eudaimonic well-being. *Annual Review of Psychology*, 52(1), 141-166. https://doi.org/10.1146/annurev.psych.52.1.141
- Ryff, C. D. (2019). Entrepreneurship and eudaimonic well-being: Five venues for new science. *Journal of Business Venturing*, 34(4), 646-663. https://doi.org/10.1016/j.jbusvent.2018.09.003
- Ryff, C. D., & Keyes, C. L. M. (1995). The structure of psychological well-being revisited. *Journal of Personality and Social Psychology*, 69(4), 719-727. https://doi.org/10.1037/0022-3514.69.4.719
- Ryff, C. D., Morozink Boylan, J., & Kirsch, J. A. (2021). Eudaimonic and hedonic well-being. An integrative perspective with linkages to sociodemographic factors and health. In M. T. Lee, L. D. Kubzansky, & T. J. VanderWeele (Eds.), *Measuring well-being: Interdisciplinary perspectives from the social sciences and the humanities* (pp. 92-135). Oxford University Press. https://doi.org/10.1093/oso/9780197512531.001.0001

- Sackett, P. R., & Larson, J. R., Jr. (1990). Research strategies and tactics in industrial and organizational psychology. In M. D. Dunnette & L. M. Hough (Eds.), *Handbook of industrial and organizational* psychology (pp. 419-489). Consulting Psychologists Press.
- Safe Work Australia. (2022, July). *Managing psychosocial hazards at work. Code of Conduct.* https://www.safeworkaustralia.gov.au/sites/default/files/2022-08/model_code_of_practice_-_managing_psychosocial_hazards_at_work_25082022_0.pdf
- Salanova, M., Agut, S., & Peiró, J. M. (2005). Linking organizational resources and work engagement to employee performance and customer loyalty: The mediation of service climate. *Journal of Applied Psychology*, 90(6), 1217-1227. https://doi.org/10.1037/0021-9010.90.6.1217
- Salanova, M., Schaudeli, W. B., Xanthopoulou, D., & Bakker, A. B. (2010). The gain spiral of resources and work engagement: Sustaining a positive worklife. In A. B. Bakker & M. P. Leiter (Eds.), *Work engagement: A handbook of essential theory and research* (pp. 118-131). Psychology Press. Taylor & Francis Group.
- Salem, N. H., Ishaq, M. I., Yaqoob, S., Raza, A., & Zia, H. (2022). Employee engagement, innovative work behaviour, and employee wellbeing: Do workplace spirituality and individual spirituality matter? *Business Ethics, the Environment & Responsibility*. Advance online publication. https://doi.org/10.1111/beer.12463
- Sales-Pardo, M., Guimerà, R., Moreira, A. A., & Amaral, L. A. N. (2007). Extracting the hierarchical organization of complex systems. *Proceedings of the National Academy of Sciences -PNAS*, 104(39), 15224-15229. https://doi.org/10.1073/pnas.0703740104
- Salmee, E., Mansor, Z., Mohamed Zainal, S., & Yasin, I. (2020). Multilevel analysis on employee wellbeing: The roles of authentic leadership, rewards, and meaningful work. *Asian Academy of Management Journal*, 25. http://dx.doi.org/10.21315/aamj2020.25.1.7
- Salmerón Gómez, R., García García, C., & García Pérez, J. (2020). Comment on "a note on collinearity diagnostics and centering" by Velilla (2018). *The American Statistician*, 74(1), 68-71. https://doi.org/10.1080/00031305.2019.1635527
- Saunders, M. N. K., Lewis, P., & Thornhill, A. (2019). *Research methods for business students* (8th ed.). Pearson.
- Schmitt, T. A., Sass, D. A., Chappelle, W., & Thompson, W. (2018). Selecting the "best" factor structure and moving measurement validation forward: An illustration. *Journal of Personality Assessment*, 100(4), 345-362. https://doi.org/10.1080/00223891.2018.1449116
- Schneider, B., Ehrhart, M. G., & MacEy, W. H. (2013). Organizational climate and culture. Annual Review of Psychology, 64(1), 361-388. https://doi.org/10.1146/annurev-psych-113011-143809
- Schoenewolf, G. (1990). Emotional contagion: Behavioral induction in individuals and groups. *Modern Psychoanalysis*, *15*, 49-61.
- Scotland, J. (2012). Exploring the philosophical underpinnings of research: Relating ontology and epistemology to the methodology and methods of the scientific, interpretive, and critical research paradigms. *English Language Teaching*, 5(9), 9-16.

- Selenko, E., Mäkikangas, A., Mauno, S., & Kinnunen, U. (2013). How does job insecurity relate to selfreported job performance? Analysing curvilinear associations in a longitudinal sample. *Journal of Occupational and Organizational Psychology*, 86(4), 522-542. https://doi.org/10.1111/joop.12020
- Seligman, M. E. P. (1998). Learned optimism. Pocket Books.
- Seligman, M. E. P. (2002). Authentic happiness. Using the new positive psychology to realize your potential for lasting fulfillment. The Free Press.
- Seligman, M. E. P. (2012). *Flourish. A visionary new understanding of happiness and well-being* (7th ed.). SAGE Publications.
- Seligman, M. E. P., Steen, T. A., Park, N., & Peterson, C. (2005). Positive psychology progress: Empirical validation of interventions. *The American Psychologist*, 60(5), 410. https://doi.org/10.1037/0003-066X.60.5.410
- Serenko, A. (2022). The Great Resignation: The great knowledge exodus or the onset of the Great Knowledge Revolution? *Journal of Knowledge Management*. Advance online publication. https://doi.org/10.1108/JKM-12-2021-0920
- Seron, C., Roberts, K. H., Hulin, C. L., & Rousseau, D. M. (1980). Developing an interdisciplinary science of organizations. *Contemporary Sociology*, 9(1), 113. https://doi.org/10.2307/2065621
- Sestino, A., & De Mauro, A. (2022). Leveraging artificial intelligence in business: Implications, applications and methods. *Technology Analysis & Strategic Management*, 34(1), 16-29. https://doi.org/10.1080/09537325.2021.1883583
- Shaheen, M. (2021). The concept of originality in academic research of engineering. *Education Research International, 2021*, 1-14. Advance online publication. https://doi.org/10.1155/2021/9462201
- Shahzad, M. A., Iqbal, T., Jan, N., & Zahid, M. (2022). The role of transformational leadership on firm performance: Mediating effect of corporate sustainability and moderating effect of knowledgesharing. *Frontiers in Psychology*, 13, 883224. Advance online publication. https://doi.org/10.3389/fpsyg.2022.883224
- Sharot, T., & Sunstein, C. R. (2020). How people decide what they want to know. *Nature Human Behaviour*, 4(1), 14-19. https://doi.org/10.1038/s41562-019-0793-1
- Sheldon, K. M., & Houser-Marko, L. (2001). Self-concordance, goal attainment, and the pursuit of happiness: Can there be an upward spiral? *Journal of Personality and Social Psychology*, 80(1), 152-165. https://doi.org/10.1037/0022-3514.80.1.152
- Shi, X., & Gordon, S. (2020). Organizational support versus supervisor support: The impact on hospitality managers' psychological contract and work engagement. *International Journal of Hospitality Management*, 87, 102374. https://doi.org/10.1016/j.ijhm.2019.102374
- Shieh, Y. Y., & Fouladi, R. T. (2003). The effect of multicollinearity on multilevel modeling parameter estimates and standard errors. *Educational and Psychological Measurement*, 63(6), 951-985. https://doi.org/10.1177/0013164403258402
- Shirom, A. (1989). Burnout in work organizations. In C. L. Cooper & I. Robertson (Eds.), International review of industrial and organizational psychology (p. 25-48), Wiley.

- Sideridis, G. D., Tsaousis, I., & Al-Sadaawi, A. (2018). Assessing construct validity in math achievement: An application of multilevel structural equation modeling (MSEM). *Frontiers in Psychology*, 9, 1451-1467.
- Siebert, D. C., Siebert, C. F., & Taylor-McLaughlin, A. (2007). Susceptibility to emotional contagion: Its measurement and importance to social work. *Journal of Social Service Research*, 33(3), 47-56. https://doi.org/10.1300/J079v33n03_05
- Simms, L. J., Zelazny, K., Williams, T. F., Bernstein, L., Ben-Porath, Y. S., & de Clercq, B. (2019). Does the number of response options matter? Psychometric perspectives using personality questionnaire data. *Psychological Assessment*, 31(4), 557-566. https://doi.org/10.1037/pas0000648
- Simon, H. A., & Laird, J. (2019). The sciences of the artificial (3rd ed.). The MIT Press.
- Simonoff, J. S., Scott, M. A., & Marx, B. P. (2013). The SAGE handbook of multilevel modeling. SAGE Publications.
- Sjöblom, K., Mäkiniemi, J. P., Mäkikangas, A. (2022). "I was given three marks and told to buy a Porsche" – Supervisors' experiences of leading psychosocial safety climate and team psychological safety in a remote academic setting. *International Journal of Environmental Research and Public Health*, 19(19), 12016. https://doi.org/10.3390/ijerph191912016
- Smith, H. P., Ferdik, F., Turner, A. L., & Radcliffe, S. (2022). An evaluation of a yoga program designed for correctional administrators and officers. *Journal of Offender Rehabilitation*, 61(1), 37-60. https://doi.org/10.1080/10509674.2021.2017385
- Song, M. K., Lin, F. C., Ward, S. E., & Fine, J. P. (2013). Composite variables: When and how. *Nursing research (New York)*, 62(1), 45-49. https://doi.org/10.1097/NNR.0b013e3182741948
- South Australia Government. (2015). Your workplace index. How can your business perform at its best? South Australia Government. https://docplayer.net/14715384-Your-workplace-index-how-can-your-business-perform-at-its-best.html
- Spell, C. S. & Arnold, T. J. (2007). A multi-level analysis of organizational justice climate, structure, and employee mental health. *Journal of Management*, 33(5), 724-751. https://doi.org/10.1177/0149206307305560
- Staines, R. (2010). Benefits of teamwork. Nursing Standard, 24(28), 62-63.

Steely Smith, M., Cooley, B., & Bensel, T. T. (2022). "We are all humans and deserve a decent way to go": Examining professional's experiences with providing end-of-life care in correctional institutions. *Criminal Justice Review*, 47(2), 225-242. https://doi.org/10.1177/07340168211020810

Stein, J. H., & Cropanzano, R. (2011). Death awareness and organizational behavior. Journal of Organizational Behavior, 32(8), 1189-1193. https://doi.org/10.1002/job.715

Steinert, S. (2020). Corona and value change: The role of social media and emotional contagion. *Ethics and Information Technology*, 23(Suppl 1), 59-68. https://doi.org/10.1007/s10676-020-09545-z

Stelman, S. (2016). *Employee engagement in the interpersonal context* [Master's thesis, Colorado State University]. ProQuest.

https://mountainscholar.org/bitstream/handle/10217/176659/Stelman_colostate_0053N_13692.pdf?s equence=1&isAllowed=y

Stevens, J. P. (2009). Applied multivariate statistics for the social sciences (5th ed.). Routledge.

- Stinglhamber, F., Nguyen, N., Josse, M., & Demoulin, S. (2022). The development of prison officers' job satisfaction and its impact on depersonalization of incarcerated persons: The role of organizational dehumanization. *Criminal Justice and Behavior*, 49(11), 9385482210871-9385482211617. https://doi.org/10.1177/00938548221087182
- Sturm, V. E., Datta, S., Roy, A. R. K., Sible, I. J., Kosik, E. L., Veziris, C. R., Chow, T. E., Morris, N. A., Neuhaus, J., Kramer, J. H., Miller, B. L., Holley, S. R., & Keltner, D. (2020). Big smile, small self: Awe walks promote prosocial positive emotions in older adults. *Emotion*, 22(5), 1044-1058. https://doi.org/10.1037/emo0000876
- Su, L., Tang, B., & Nawijn, J. (2020). Eudaimonic and hedonic well-being pattern changes: Intensity and activity. Annals of Tourism Research, 84, 103008. https://doi.org/10.1016/j.annals.2020.103008
- Subramony, M., Golubovskaya, M., Keating, B., Solnet, D., Field, J., & Witheriff, M. (2022). The influence of pandemic-related workplace safety practices on frontline service employee wellbeing outcomes. *Journal of Business Research*, 149, 363-374. https://doi.org/10.1016/j.jbusres.2022.05.040
- Suter, J., Kowalski, T., Anaya-Montes, M., Chalkley, M., Jacobs, R., & Rodriguez-Santana, I. (2020). The impact of moving to a 12h shift pattern on employee wellbeing: A qualitative study in an acute mental health setting. *International Journal of Nursing Studies*, *112*, 103699 https://doi.org/10.1016/j.ijnurstu.2020.103699
- Sy, T., Côté, S., & Saavedra, R. (2005). The contagious leader: Impact of the leader's mood on the mood of group members, group affective tone, and group processes. *Journal of Applied Psychology*, 90(2), 295-305. https://doi.org/10.1037/0021-9010.90.2.295
- Tang, Y., Lam, C. K., Ouyang, K., Huang, X., & Tse, H. H. M. (2022). Comparisons draw us close: The influence of leader-member exchange dyadic comparison on coworker exchange. *Personnel Psychology*, 75(1), 215-240. https://doi.org/10.1111/peps.12465
- Taylor, S. E. (1991). Asymmetrical effects of positive and negative events: The mobilizationminimization hypothesis. *Psychological Bulletin*, 110(1), 67-85. https://doi.org/10.1037/0033-2909.110.1.67
- Teclaw, R., Price, M. C., & Osatuke, K. (2012). Demographic question placement: Effect on item response rates and means of a veterans' health administration survey. *Journal of Business and Psychology*, 27(3), 281-290. https://doi.org/10.1007/s10869-011-9249-y
- Teddlie, C., & Tashakkori, A. (2009). Foundations of mixed methods research: Integrating quantitative and qualitative approaches in the social and behavioral sciences. SAGE Publications.
- Tee, E. Y. J. (2015). The emotional link: Leadership and the role of implicit and explicit emotional contagion processes across multiple organizational levels. *The Leadership Quarterly*, 26(4), 654-670. https://doi.org/10.1016/j.leaqua.2015.05.009
- Tellegen, A., Lykken, D. T., Bouchard, T. J., Wilcox, K. J., Segal, N. L., & Rich, S. (1988). Personality similarity in twins reared apart and together. *Journal of Personality and Social Psychology*, 54(6), 1031-1039. https://doi.org/10.1037/0022-3514.54.6.1031
- Terpstra-Tong, J., Ralston, D. A., Treviño, L. J., Naoumova, I., de la Garza Carranza, M. T., Furrer, O., Li, Y., & Darder, F. L. (2020). The quality of Leader-member exchange (LMX): A multilevel

analysis of individual-level, organizational-level and societal-level antecedents. *Journal of International Management*, *26*(3), 100760. https://doi.org/10.1016/j.intman.2020.100760

- Thien, L. M., Uthai, M., & Yeap, S. B. (2022). Does middle leaders' learning-centred leadership matter in promoting teacher professional learning? A partial least squares analysis. *Educational Management, Administration & Leadership*. Advance online publication. https://doi.org/10.1177/17411432221115522
- Thode, H. C. (2002). Testing for normality. Marcel Dekker. https://doi.org/10.1201/9780203910894
- Topp, C.W., Østergaard S.D., Søndergaard S., & Bech P. (2015). The WHO-5 Well-Being Index: A systematic review of the literature. *Psychotherapy and Psychosomatics*, 84, 167-176.
- Tortia, E. C., Sacchetti, S., & López-Arceiz, F. J. (2022). A human growth perspective on sustainable HRM practices, worker well-being and organizational performance. *Sustainability*, *14*(17), 11064. https://doi.org/10.3390/su141711064
- Totterdell, P., Kellett, S., Teuchmann, K., & Briner, R. B. (1998). Evidence of mood linkage in work groups. *Journal of Personality and Social Psychology*, 74(6), 1504-1515. https://doi.org/10.1037/0022-3514.74.6.1504
- Tsui, A. S., Pearce, J. L., Porter, L. W., & Tripoli, A. M. (1997). Alternative approaches to the employeeorganizational relationship: Does investment in employees pay off? *The Academy of Management Journal*, 40(5), 1089-1121. https://doi.org/10.2307/256928
- Tuan, L. T. (2016). How servant leadership nurtures knowledge sharing the mediating role of public service motivation. *International Journal of Public Sector Management*, 29(1), 91-108. https://doi.org/10.1108/IJPSM-06-2015-0112
- Tuan, L. T. (2022). Leader crisis communication and salesperson resilience in face of the COVID-19: The roles of positive stress mindset, core beliefs challenge, and family strain. *Industrial Marketing Management*, 102, 488-502. https://doi.org/10.1016/j.indmarman.2021.12.006
- Tucker, L. R., & Lewis, C. (1973). A reliability coefficient for maximum likelihood factor analysis. *Psychometrika*, 38(1), 1-10. https://doi.org/10.1007/BF02291170
- van de Schoot, R., & Hox, J. (2013). Robust methods for multilevel analysis. In M. A. Scott, J. S. Simonoff & B. D. Marx (Eds.), *The SAGE handbook of multilevel modeling* (p. 402), SAGE Publications. https://dx.doi.org/10.4135/9781446247600
- van Maanen, J., & Kunda, G. (1989). Real feelings: Emotional expression and organizational culture. In L. L. Cummings & B. M. Staw (Eds.), *Research in organizational behavior* (Vol. 11, pp. 43-104). JAI Press.
- van Mol, C. (2017). Improving web survey efficiency: The impact of an extra reminder and reminder content on web survey response. *International Journal of Social Research Methodology*, 20(4), 317-327. https://doi.org/10.1080/13645579.2016.1185255
- van Prooijen, J. W., & van der Kloot, W. A. (2001). Confirmatory analysis of exploratively obtained factor structures. *Educational and Psychological Measurement*, 61(5), 777-792. https://doi.org/10.1177/00131640121971518

- van Woerkom, M., Bakker, A. B., & Nishii, L. H. (2016). Accumulative job demands and support for strength use: Fine-tuning the job demands-resources model using conservation of resources theory. *Journal of Applied Psychology*, 101(1), 141-150. https://doi.org/10.1037/apl0000033
- Vanhala, S., & Tuomi, K. (2006). HRM, company performance and employee well-being. *Management Revue*, 17(3), 241-255.
- Vashistha, A., Okeke, F., Anderson, R., & Dell, N. (2018). You can always do better!: The impact of social proof on participant response bias. *Proceedings of the 2018 CHI Conference on Human Factors in Computing Systems*, 1-13. https://doi.org/10.1145/3173574.3174126
- Vine, B. (2004). *Getting things done at work the discourse of power in workplace interaction*. John Benjamins Pub.
- Vinokur, A. D., & Schul, Y. (2002). The web of coping resources and pathways to reemployment following a job loss. *Journal of Occupational Health Psychology*, 7(1), 68-83. https://doi.org/10.1037/1076-8998.7.1.68
- Vitality Works. (2019). Wellbeing360. https://vitalityworks.health/vitality-services/wellbeing-360/https://vitalityworks.health/vitality-services/wellbeing-360/
- Vittersø, J. (2016). *Handbook of eudaimonic well-being. International handbooks of quality-of-life* (1st ed.). Springer International. https://doi.org/10.1007/978-3-319-42445-3
- Volini, E., Schwartz, J., Denny, B., Mallon, D., Van Durme, Y., Hauptmann, M., Yan, R., & Poynron, S. (2020). *The social enterprise at work: Paradox as a path forward. 2020 Deloitte global human capital trends*. https://www2.deloitte.com/us/en/insights/focus/human-capitaltrends/2020/designing-work-employee-well-being.html
- Wang, Y., Yuan, C., & Zhu, Y. (2017). Coaching leadership and employee voice behavior: A multilevel study. Social Behavior and Personality: An International Journal, 45(10), 1655-1664. https://doi.org/10.2224/sbp.6593
- Wang, Z., Panaccio, A., Raja, U., Donia, M., Landry, G., Pereira, M. M., & Ferreira, M. C. (2022). Servant leadership and employee wellbeing: A cross cultural investigation of the moderated path model in Canada, Pakistan, China, the US, and Brazil. *International Journal of Cross Cultural Management: CCM*, 22(2), 301-325. https://doi.org/10.1177/14705958221112859
- Ward, A. K., Beal, D. J., Zyphur, M. J., Zhang, H., & Bobko, P. (2021). Diversity climate, trust, and turnover intentions: A multilevel dynamic system. *Journal of Applied Psychology*, 107(4), 628-649. https://doi.org/10.1037/ap10000923
- Waterman, A. S. (1993). Two conceptions of happiness: Contrasts of personal expressiveness (eudaimonia) and hedonic enjoyment. *Journal of Personality and Social Psychology*, 64(4), 678-691. https://doi.org/10.1037/0022-3514.64.4.678
- Waterman, A. S., Schwartz, S. J., Zamboanga, B. L., Ravert, R. D., Williams, M. K., Bede Agocha, V., Yeong Kim, S., & Brent Donnellan, M. (2010). The questionnaire for eudaimonic well-being: Psychometric properties, demographic comparisons, and evidence of validity. *The Journal of Positive Psychology*, 5(1), 41-61. https://doi.org/10.1080/17439760903435208
- Watkins, M. W. (2018). Exploratory factor analysis: A guide to best practice. *Journal of Black Psychology*, 44(3), 219-246. https://doi.org/10.1177/0095798418771807

- Watson, D., Clark, L. A., & Tellegen, A. (1988). Development and validation of brief measures of positive and negative affect: The PANAS scales. *Journal of Personality and Social Psychology*, 54(6), 1063-1070. https://doi.org/10.1037/0022-3514.54.6.1063
- Wells, J. D., Hobfoll, S. E., & Lavin, J. (1997). Resource loss, resource gain, and communal coping during pregnancy among women with multiple roles. Communal coping during pregnancy among women with multiple roles. *Psychology of Women Quarterly*, 21(4), 645-662. https://doi.org/10.1111/j.1471-6402.1997.tb00136.x
- Westfall, P. H., & Henning, K. S. S. (2013). *Texts in statistical science: Understanding advanced statistical methods*. Taylor & Francis.
- Wharton, A. S. (2009). The sociology of emotional labor. *Annual Review of Sociology*, 35(1), 147-165. https://doi.org/10.1146/annurev-soc-070308-115944
- Wheeler, A. R., Halbesleben, J. R. B., & Whitman, M. V. (2013). The interactive effects of abusive supervision and entitlement on emotional exhaustion and co-worker abuse. *Journal of Occupational* and Organizational Psychology, 86(4), 477-496. https://doi.org/10.1111/joop.12034
- Whitman, M. V., Halbesleben, J. R. B., & Holmes, O. (2014). Abusive supervision and feedback avoidance: The mediating role of emotional exhaustion. *Journal of Organizational Behavior*, 35(1), 38-53. https://doi.org/10.1002/job.1852
- Wicke, F. S., Krakau, L., Löwe, B., Beutel, M. E., & Brähler, E. (2022). Update of the standardization of the Patient Health Questionnaire-4 (PHQ-4) in the general population. *Journal of Affective Disorders*, *312*, 310-314. https://doi.org/10.1016/j.jad.2022.06.054
- Wieland, A., Durach, C. F., Kembro, J., & Treiblmaier, H. (2017). Statistical and judgmental criteria for scale purification. *Supply Chain Management*, 22(4), 321-328. https://doi.org/10.1108/SCM-07-2016-0230
- Wilson, J. (2014). *Essentials of business research: A guide to doing your research project* (2nd ed.). SAGE Publications.
- Wilson, K. S., Sin, H. P., & Conlon, D. E. (2010). What about the leader in Leader-member exchange? The impact of resource exchanges and substitutability on the leader. *The Academy of Management Review*, 35(3), 358-372. https://doi.org/10.5465/AMR.2010.51141654
- Winter, S. D., & Depaoli, S. (2022). Performance of model fit and selection indices for Bayesian structural equation modeling with missing data. *Structural Equation Modeling*, 29(4), 531-549. https://doi.org/10.1080/10705511.2021.2018656
- Worku, Z. (2019). A study of employee perceptions about performance appraisal at Transnet engineering, South Africa. *Journal of Applied Business Research*, 35(5), 145-156. https://doi.org/10.19030/jabr.v35i5.10307
- World Health Organisation. (2020, March 11). WHO Director-General's opening remarks at the media briefing on COVID-19 - 11 March 2020. https://www.who.int/director-general/speeches/detail/whodirector-general-s-opening-remarks-at-the-media-briefing-on-covid-19---11-march-2020
- Worley, R. M., Lambert, E. G., & Worley, V. B. (2022). Can't shake the prison guard blues: Examining the effects of work stress, job satisfaction, boundary violations, and the mistreatment of inmates on

the depressive symptomatology of correctional officers. *Criminal Justice Review*. Advance online publication. https://doi.org/10.1177/07340168221123229

- Wright, T. A., & Cropanzano, R. (2000). Psychological well-being and job satisfaction as predictors of job performance *Journal of Occupational Health Psychology*, 5(1), 88-94.
- Wrzesniewski, A., McCauley, C., Rozin, P., & Schwartz, B. (1997). Jobs, careers, and callings: People's relations to their work. *Journal of Research in Personality*, 31(1), 21-33. https://doi.org/10.1006/jrpe.1997.2162
- Wu, J. Y., Lin, J. J. H., Nian, M. W., & Hsiao, Y. C. (2017). A solution to modeling multilevel confirmatory factor analysis with data obtained from complex survey sampling to avoid conflated parameter estimates. *Frontiers in Psychology*, 8, 1464. https://doi.org/10.3389/fpsyg.2017.01464
- Xanthopoulou, D., Bakker, A. B., Demerouti, E., & Schaufeli, W. B. (2009). Work engagement and financial returns: A diary study on the role of job and personal resources. *Journal of Occupational* and Organizational Psychology, 82(1), 183-200. https://doi.org/10.1348/096317908X285633
- Xerri, M. J., Brunetto, Y., & Farr-Wharton, B. (2019). Support for aged care workers and quality care in Australia: A case of contract failure? *Australian Journal of Public Administration*, 78(4), 546-561. https://doi.org/10.1111/1467-8500.12379
- Xerri, M. J., Cozens, R., & Brunetto, Y. (2022). Catching emotions: The moderating role of emotional contagion between Leader-member exchange, psychological capital and employee well-being. *Personnel Review*. https://doi.org/10.1108/PR-11-2021-0785
- Xiao, Q., Cooke, F. L., Mavondo, F., & Bamber, G. J. (2022). Antecedent and employee well-being outcomes of perceived benefits schemes: A two-wave study. *International Journal of Manpower*, 43(5), 1166-1181. https://doi.org/10.1108/IJM-08-2020-0374
- Xie, Chang, C.-N., & Singh, S. (2021). Emotional intelligence, voice and flow: A team-level study of work teams. *Team Performance Management*, 27(7/8), 524-539. https://doi.org/10.1108/TPM-12-2020-0110
- Yean, T. F., Johari, J., Yahya, K. K., & Chin, T. L. (2022). Determinants of job dissatisfaction and its impact on the counterproductive work behavior of university staff. SAGE Open, 12(3). https://doi.org/10.1177/21582440221123289
- Zhang, X., & Savalei, V. (2022). New computations for RMSEA and CFI following FIML and TS estimation with missing data. *Psychological Methods*. Advance online publication. https://doi.org/10.1037/met0000445
- Zhou, F., & Jiang, C. (2015). Leader-member exchange and employees' safety behavior: The moderating effect of safety climate. *Procedia Manufacturing*, 3, 5014–5021. https://doi.org/10.1016/j.promfg.2015.07.671
- Zyphur, M. J. (2022). *Multilevel SEM in Mplus. Live Streaming 2-Day Workshop* [Live streaming video]. Instats. https://instats.org/

Appendices

Appendix 1. Survey

Team Influence on Employee Wellbeing

UTS Ethics Approval Number: ETH21-5814

As an employee working at ORGANISATION NAME, we warmly invite you to complete this survey. I am a PhD student at UTS and this survey is part of the data collection of my research, where I am studying the team enablers of employee wellbeing. The survey has been designed to capture data regarding your wellbeing at work. There are no right or wrong answers, please seek to be completely honest in your answers. It takes about 7 to 14 minutes to complete and all of your responses will be anonymous and treated confidentially. For more information about how your data will be used, please click here.

Do you consent to participate in this survey online?

- Yes, I consent to participate in this survey
- No, I do not consent to participate in this survey

While care has been taken to ensure the survey is as brief as possible and non-confrontational in style if at any stage the content makes you feel uncomfortable you may cease giving data. If this survey causes you distress and/or if you would like to speak to someone about how you feel, you may like to refer to free services such as Head to Health (headtohealth.gov.au), Beyond Blue (1300 22 4636 / beyondblue.org.au) or Lifeline (13 11 14 / lifeline.org.au).

What is your Employee Code?

This survey is trying to capture the team enablers of employee wellbeing. To be able to link your responses with your work team, you have been asked to supply the following 'code'. This code will be stripped out of your response after your work team has been linked, and the code will not be disclosed at any time to your organisation.

- First 3 letters of your Manager's first name
- First 3 letters of your Manager's last name

Psychosocial Safety Climate

This scale is designed to capture the shared perceptions regarding policies, practices and procedures for the protection of worker psychological health and safety.

	Strongly disagree	Disagree	Neutral	Agree	Strongly agree
Managers show support for stress prevention through involvement and commitment (PSC_1)					
Management clearly considers the psychological health of employees to be of great importance (PSC_2)					
There is good communication here about psychological safety issues which affect me (PSC_3)					
In my organisation, the prevention of stress involves all levels of the organisation (PSC_4)					

Supervisor Relationship

This questionnaire contains items that ask you to describe your relationship with your direct supervisor.

For each one of the items, please indicate the degree to which you think the item is true for you by clicking one of the responses that appear below the item.

LMX_1 Do you know where you stand with your direct supervisor and do you usually know how satisfied your direct supervisor is with what you do?

Rarely

- Occasionally
- Sometimes
- Fairly often
- Very often

LMX_2 How well does your direct supervisor understand your job problems and needs?

- Not a bit
- A little
- A fair amount
- Quite a bit
- A great deal

LMX_3 How well does your direct supervisor recognise your potential?

- Not at all
- o A little
- A fair amount
- Fairly often
- Very often

LMX_4 Regardless of how much formal authority your direct supervisor has built into his or her position, what are the chances that your direct supervisor would use his or her power to help you solve problems in your work?

- o None
- o Small
- Moderate
- o High
- Very high

LMX_5 Again, regardless of the amount of formal authority your direct supervisor has, what are the chances that he or she would "bail you out" at his or her expense?

- o None
- o Small
- Moderate
- o High
- Very high

LMX_6 I have enough confidence in my direct supervisor that I would defend and justify his or her decision if he or she were not present to do so.

- Strongly disagree
- Disagree
- Neutral
- Agree
- Strongly agree

LMX_7 How would you characterize your working relationship with your direct supervisor?

- Extremely ineffective
- Worse than average
- o Average
- Better than average
- Extremly effective

Emotional Contagion

This is a scale that measures a variety of feelings and behaviours in the workplace.

There are no right or wrong answers, please try to be completely honest in your answers.

	Strongly disagree	Disagree	Slightly disagree	Slightly agree	Agree	Strongly agree
When there is excitement happening around me at work, I remain cool (EC_1)						
At work, if colleagues appear to be worried about something, I generally remain calm (EC_2)						
If colleagues around me feel depressed, I also tend to feel depressed (EC_3)						
If a colleague felt upset, I would also feel upset (EC_4)						
At work, I would become nervous if colleagues around me appear nervous (EC_5)						
The colleagues around me have a strong influence on my mood (EC_6)						

Employee Wellbeing

This construct is designed to capture the level of employee wellbeing.

There are no right or wrong answers, please try to be completely honest in your answers.

	Strongly disagree	Disagree	Slightly disagree	Slightly agree	Agree	Strongly agree
Overall, I am reasonably happy with my work life (EWB_1)						
Most days I feel a sense of accomplishment in what I do at work (EWB_2)						
I feel content with my work (EWB_3)						
I get a sense of joy from my work (EWB_4n)						

Work Stress

This construct is designed to capture the level of work stress.

There are no right or wrong answers, please try to be completely honest in your answers.

	Strongly disagree	Disagree	Neutral	Agree	Strongly agree
A lot of the time my job makes me very frustrated or angry (WS_1)					
I am usually under a lot of pressure when I am at work (WS_2)					
When I am at work I often feel tense or uptight (WS_3)					
I am usually calm and at ease when I'm at work (WS_4)					
There are a lot of aspects of my job that make me upset (WS_5)					

You're almost at the end. You've already completed 85% of the survey!

What is your gender?

- o Female
- Male 0
- 0 Intersex
- Transgender 0
- Prefer not to say
- o Other

What is your year of birth?

▼ 21 (2008) ... 1900 (1900)

Approximately, how many years have you been working in this organisation?

- Less than 1 year
- o 1-2 years
- 3-5 years
 6-10 years
 11-15 years
- More than 15 years

Appendix 2. Recruitment flyer and video link

Team Influence on EMPLOYEE WELLBEING

Aims:

- Identify how work teams influence their members' wellbeing
- Examine how individual wellbeing shape the wellbeing of their coworkers.
- Explore how organisations can support their employees' wellbeing

Project brief:

Opportunity to participate:

I am inviting organisations to 'sign up' to the study, and invite their employees working as part of a team of at least 3 members. The online survey takes between 7 to 10 minutes to complete. Participation in the study is free for all eligible Australian organisations.

I am a PhD student researching the team enablers of employee wellbeing. Employee wellbeing focuses on workers physical, social, and psychological state at work. It is a contagious variable as employees influence others' wellbeing.

In 2020, employee wellbeing had the largest gap between importance and readiness, with 80% of organisations finding it important for their success, but only 12% saying they were ready to address it (2020 Deloitte Global Human Capital Trends).

The objective of this project is to develop a deeper understanding on how organisations can support their employees' wellbeing and how employees influence their coworkers wellbeing when working together.

Benefits for participants:

In return, participating organisations will receive a benchmark report with the results summary of the research. This will assist the development of effective workplace plans to work on your teams' wellbeing.

REGISTER YOUR INTEREST

If you are interested in joining the study, please complete the online form available <u>here</u> and I will be in touch shortly.

REGISTER YOUR INTEREST HERE

CONTACT INFORMATION

Aglae Hernandez Grande PhD Student - University of Technology Sydney E: Aglae.HernandezGrande@student.uts.edu.au M·I

Professor Simon Darcy Principal Supervisor - University of Technology Sydney E: Simon.Darcy@uts.edu.au UTS Ethics Number: ETH21-5814

YouTube video link: https://www.youtube.com/watch?v=Xcm7gARRmF4

Appendix 3. Generic participant information sheet

This participant information sheet was adapted to each participant organisation.



INFORMATION SHEET AND CONSENT FORM FOR ONLINE SURVEYS

UTS HREC ETH21-5814 - TEAM ANTECEDENTS OF EMPLOYEE WELLBEING

What is the research study about?

The purpose of this research is to establish a better understanding of employee wellbeing as well as a deeper awareness of its team enablers. With this research, I wish to help developing healthier and better workplaces, where employee wellbeing is a crucial aim of every organisation.

You have been invited to participate to this study as an employee of NAME OF THE ORGANISATION; however, individual participation is voluntary, and it will not have any impact on employment.

Who is conducting this research?

My name is Aglae Hernandez Grande, and I am a PhD student at UTS. My principal supervisor is Professor Simon Darcy, and he can be contacted via email at <u>simon.darcy@uts.edu.au</u>.

Do I have to take part in this research study?

Participation in this study is voluntary. It is completely up to you whether or not you decide to take part. If you decide to participate, I will invite you to read this information carefully, and complete an online survey that will take approximately 7-14 minutes to complete. The survey uses a rating scale and will ask you questions about your relationship with your supervisor, how others' feelings and behaviours influence you, as well as questions on your own wellbeing, for example, 'Overall, I am reasonably happy with my work life'.

In order for the study to be valid, all questions should be answered, however, you can change your mind at any time and stop completing the survey without consequences and your already-given responses will be deleted and not used in any analysis.

Are there any risks/inconvenience?

While care has been taken to ensure the survey is as brief as possible and non-confrontational in style, if at any stage the content makes you feel uncomfortable you may stop the survey. If answering or considering the survey questions causes you distress and/or if you would like to speak to someone about how you feel, you may like to refer to free services such as Head to Health (headtohealth.gov.au), Beyond Blue (1300 22 4636 / beyondblue.org.au) or Lifeline (13 11 14 / lifeline.org.au). You can also contact your employee assistance program (DETAILS HERE).

What will happen to information about me?

The survey is anonymous and confidential. It is collected by an external research team and all data will be stored in a secure location at UTS. We don't collect any identifiable data that it is not provided by you in the survey. By clicking the Yes, I consent to participate in this survey, you consent to the research team using the information you give for the research project. You will then be able to progress to the survey questions. At the beginning of the survey, you will be asked to confirm you have read and understood this information sheet. You will then be asked to provide a six-letter code, with the three first letters of your manager's first and last name. This code will be used to identify the team in which you belong so we can study the team's influence on employee wellbeing. To ensure your identity remains anonymous, the six-letter code will be stripped out of your response as soon as the team has been linked and stored on a secure storage within UTS. The code will never be shared with anyone outside of the research team. All this information will be treated confidentially and will not have any impact on employment. Your information will only be used for the purpose of this research project. Participating organisations may receive a benchmark report with the aggregated results of the study, but your individual response will not be disclosed at any stage to anyone, except as required by law.

What if I have concerns or a complaint?

If you have concerns about the research that you think I or my supervisor can help you with, please feel free to contact us on:

- Aglae Hernandez Grande (PhD Student): aglae.hernandezgrande@student.uts.edu.au
- Professor Simon Darcy (Principal Supervisor): simon.darcy@uts.edu.au

If you would like to talk to someone who is not connected with the research, you may contact the Research Ethics Officer on 02 9514 9772 or <u>Research.ethics@uts.edu.au</u> and quote this number *ETH21-5814*.

Appendix 4. Multilevel exploratory factor analysis for psychosocial safety climate and emotional contagion

```
VARIABLE:
```

```
NAME = ...;
USEVARIABLES = PSC_1 PSC_2 PSC_3 PSC_4
EC_1 EC_2 EC_3 EC_4 EC_5 EC_6;
CLUSTER = Group;
ANALYSIS:
TYPE IS TWOLEVEL EFA 1 2 1 2;
```

```
Appendix 5. Multilevel McDonald's omega code for the scale of employee wellbeing
```

```
VARIABLE:
     NAME = ...;
     USEVARIABLES = EWB 1 EWB 2 EWB 3 EWB 4;
     CLUSTER = Group;
ANALYSIS:
     TYPE IS TWOLEVEL;
     PROCESSOR = 2;
MODEL:
     %WITHIN%
           EWB W by EWB 1* EWB 2* EWB 3* EWB 4* (lW1-lW4);
           EWB W@1;
           EWB 1 EWB 2 EWB 3 EWB 4 (eW1-eW4);
           %BETWEEN%
           EWB B by EWB 1* EWB 2* EWB 3* EWB 4* (1B1-1B4);
           EWB B@1;
           EWB 1 EWB 2 EWB 3 EWB 3 (eB1-eB4);
Model Constraint:
New(sum12B sumeB omegaB corB sum12W sumeW omegaW corW);
suml2B = (lB1+lB2+lB3+lB4) **2;
sumeB = eB1 + eB2 + eB3 + eB4;
omegaB = suml2B/(suml2B+sumeB);
corB = sqrt(omegaB);
suml2W = (lW1+lW2+lW3+lW4) **2;
sumeW = eW1 + eW2 + eW3 + eW4;
omegaW = suml2W/(suml2W+sumeW);
corW = sqrt(omegaW);
```

```
Appendix 6. Multilevel structural equation code to test the hypothesised model
VARIABLE:
NAME = ...;
```

```
USEVARIABLES = EWB WS PSC LMX EC;
CLUSTER = Group;
DEFINE:
CENTER EC (GRANDMEAN);
ANALYSIS:
```

```
TYPE IS TWOLEVEL RANDOM;
    ESTIMATOR = BAYES;
    FBITERATIONS = 10000;
    PROCESSORS = 2;
MODEL:
    %WITHIN%
    S | EWB ON WS;
    EWB ON PSC LMX;
    WS ON PSC LMX;
    LMX ON PSC;
    %BETWEEN%
    S ON EC;
    EWB ON WS PSC LMX;
    WS ON PSC LMX;
    LMX ON PSC;
OUTPUT:
    STDYX TECH8;
```